

# TABLE OF CONTENTS

ARTIC	LE 1 - PREAMBLE	
1.01	PURPOSE OF AGREEMENT	18
1.01	EXISTING LEGISLATION	
1.03	FUTURE LEGISLATION	
1.04	HUMAN RIGHTS	
(a)	Application	
(b)	Unfair Labour Practices	
1.05	NO OTHER AGREEMENT	
(a)	Sole Bargaining Agent	
(b)	No Other Agreement	
$\frac{1.06}{1.07}$	CORRESPONDENCE	
<u>1.07</u>		
ARTIC	LE 2 - UNION RECOGNITION AND NEGOTIATION	20
2.01	BARGAINING UNIT DEFINED	
2.02	MANAGEMENT RIGHTS	
(a)	Rights of University	
(b)	Affairs and Facilities	
(c)	Fair and Reasonable	
2.03	NEW POSITIONS	
(a)	Notification of Union	
(b) 2.04	Disagreement	
ARTIC	LE 3 - UNION SECURITY	22
3.01	EMPLOYEES AT DATE OF CERTIFICATION	
3.02	NEW EMPLOYEES	
3.03	CHANGES IN STATUS	
ARTIC	LE 4 - UNION RIGHTS AND ACTIVITY	23
4.01	NOTICE OF ABSENCE FROM WORKPLACE	
4.02	RECOGNITION OF UNION REPRESENTATIVES	
4.03	CONTACTING AT WORK	
4.04	TIME OFF FOR UNION BUSINESS AND ACTIVITIES	
(a) (b)	Without Pay - Conventions	
(0) (c)	Without Pay - Bargaining Committee Meetings	
(d)	With Pay - Negotiations	
(e)	With Pay - Steward Duties	
(f)	With Pay - Grievance	
(g)	With Pay - Union Grievance Committee	
(h)	Without Pay - Union Position	
4.05	UNIVERSITY AND UNION SHALL ACQUAINT NEW EMPLOYEES	
(a) (b)	A greement in Effect	25
4.06	Agreement in Effect Union Steward to Acquaint	
	Union Steward to Acquaint	
(a)		25
(a) (b)	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES	
(b) 4.07	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES Dispute Pay UNION GENERAL MEETING	
(b) 4.07 4.08	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES Dispute Pay UNION GENERAL MEETING USE OF UNIVERSITY ROOMS	25 25 25 25 25 25 25 25 25
(b) 4.07 4.08 4.09	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES Dispute Pay UNION GENERAL MEETING USE OF UNIVERSITY ROOMS UNION OFFICE SPACE	25 25 25 25 25 25 25 25 25 26
(b) 4.07 4.08 4.09 4.10	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES. Dispute Pay UNION GENERAL MEETING USE OF UNIVERSITY ROOMS UNION OFFICE SPACE APPLICATION TO OVERTIME PREMIUM	25 25 25 25 25 25 25 25 25 25 26 26
(b) 4.07 4.08 4.09 4.10 4.11	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES. Dispute. Pay. UNION GENERAL MEETING. USE OF UNIVERSITY ROOMS. UNION OFFICE SPACE APPLICATION TO OVERTIME PREMIUM. UNION INSIGNIA.	25 25 25 25 25 25 25 25 25 26 26 26 26
(b) 4.07 4.08 4.09 4.10 4.11 (a)	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES. Dispute. Pay. UNION GENERAL MEETING. USE OF UNIVERSITY ROOMS. UNION OFFICE SPACE APPLICATION TO OVERTIME PREMIUM. UNION INSIGNIA. Wearing or Displaying.	25 25 25 25 25 25 25 25 25 26 26 26 26 26 26
(b) 4.07 4.08 4.09 4.10 4.11 (a) (b)	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES Dispute Pay UNION GENERAL MEETING USE OF UNIVERSITY ROOMS UNION OFFICE SPACE APPLICATION TO OVERTIME PREMIUM UNION INSIGNIA Wearing or Displaying Inclusion in Documents	25 25 25 25 25 25 25 25 25 26 26 26 26 26 26 26
(b) 4.07 4.08 4.09 4.10 4.11 (a)	Union Steward to Acquaint RIGHT TO REFUSE TO CROSS PICKET LINES. Dispute. Pay. UNION GENERAL MEETING. USE OF UNIVERSITY ROOMS. UNION OFFICE SPACE APPLICATION TO OVERTIME PREMIUM. UNION INSIGNIA. Wearing or Displaying.	25 25 25 25 25 25 25 25 25 25 26 26 26 26 26 26 26 26

ARTIC	LE 5 - CHECK-OFF OF UNION FEES AND DUES	28
5.01	AUTHORIZATION	28
5.02	DEDUCTION OF DUES	28
(a)	Schedule	
(b)	University's Obligation	
5.03 5.04	COLLECTION AND FORWARDING OF DUES	
5.04 5.05	NOTIFICATION INDEMNITY	
5.06	REVOCATION	
5.07	YEAR-END SUMMARY	
5.08	RECORD OF UNION DEDUCTIONS (T4 SLIPS)	
ARTIC	LE 6 - UNIVERSITY - UNION RELATIONS	30
6.01	REPRESENTATIONS	30
6.02	SEARCH COMMITTEE - ADMINISTRATOR	
6.03	ACCESS TO UNIVERSITY PREMISES	30
(a)	Negotiation and Grievance	30
(b)	Space	
6.04 6.05	INFORMATION UNION - UNIVERSITY RELATIONS	
0.03 (a)	Joint Standing Committee	
(b)	Meetings	
(c)	Agenda Items	31
6.06	EMPLOYEE ATTENDANCE AT MEETINGS	31
(a)	Required to Attend	
(b)	Invited to Attend	
ARTIC	LE 7 - GRIEVANCE	
7.01	DEFINITION OF A GRIEVANCE	
<u>7.02</u>	RIGHT TO GRIEVE	
<u>7.03</u>	SELECTION GRIEVANCES	
7.0 <u>4</u>	GRIEVANCE PROCEDURE	
7.0 <u>5</u> 7.06	GRIEVANCE STAGE - STEP 2	
(a)	Complaint Submitted to Management.	
(b)	Management and Union Steward Meet	33
(c)	Dispute Not Resolved	
7.0 <u>7</u>	STEP 3	
7.0 <u>8</u>	NON-RESOLVED GRIEVANCES	
(a) (b)	Referral to Arbitration Time Limit	
7.09	TECHNICAL OBJECTIONS TO GRIEVANCES	
7.10	RECORD OF GRIEVANCE	
7. <u>11</u>	TIME LIMITS	34
(a)	Extension	
(b)	Substance	
7.1 <u>2</u>	GRIEVANCE OF GENERAL APPLICATION OR INTERPRETATION	
7.1 <u>3</u> 7.14	GRIEVANCE INVOLVING DISCIPLINE EMPLOYEE TIME OFF TO ATTEND TO GRIEVANCE	
7.1 <u>4</u> 7.1 <u>5</u>	MAINTENANCE OF EMPLOYEE RIGHTS AND BENEFITS	
(a)	Settlements	
(b)	Maintenance of Position	35
7.1 <u>6</u>	DISCLOSURE OF INFORMATION	
7.1 <u>7</u>	EFFECT OF SETTLEMENTS	
7.1 <u>8</u>	DEVIATION FROM GRIEVANCE PROCEDURE	36
ARTIC	LE 8 - ARBITRATION	37
8.01	ARBITRATORS	37
8.02	ARBITRATION	

8.03	DECISION OF ARBITRATOR	
8.04	EXPENSES OF ARBITRATION	37
8.05	DIFFERENCES	
(a)	Investigation	
(b)	Definition	
(c)	Recommendation	
8.06	EXPEDITED ARBITRATION	
(a)	Arbitrator	
(b)	Presentation of Facts	
(c)	Decision	
(d)	Multiple Grievances	
(e)	Expedited Arbitration	
8.07	PRE-HEARING CONFERENCE	39
	CLE 9 - CATEGORIES OF EMPLOYEES	
9.01	EMPLOYEE DEFINED	
9.02	REGULAR FULL TIME (RFT)	
9.03	REGULAR PART TIME (RPT)	40
9.0 <u>4</u>	TEMPORARY FULL TIME (TFT)	40
9.0 <u>5</u>	TEMPORARY PART TIME (TPT)	40
9.0 <u>6</u>	EXCEPTIONS	40
9.07	REGULARIZATION	40
(a)	Review	
(b)	Failure to Reach Agreement	
(c)	Recommendations	
9.08	FILLING REGULARIZED POSITIONS	
9.09	EMPLOYEE APPOINTMENTS	
(a)	Acknowledgement	
(b)	Weekend Work	
(c)	Offer Letter Copy	
9.10	WORK PRACTICUMS	
(a)	Definition	
(b)	Employee Responsibilities	
(c)	Limitations	
(d)	Notification to Union	
9.11	WORK EXPERIENCE PLACEMENTS/CAREER AND PERSONAL PLANNING PROG	
9.12	STUDENT EMPLOYMENT PROGRAMS	
	CLE 10 - DISCIPLINE	
10.01	PROCEDURE	45
10.02	24 HOURS ADVANCE NOTICE	45
(a)	Accompanied by Steward	45
(b)	Purpose of Meeting	45
(c)	Access to Personnel File	
(d)	Right to Refuse Participation	
10.03	SUSPENSIONS SHALL BE IN WRITING	45
10.04	DISCHARGES	45
10.05	BURDEN OF PROOF	45
10.06	MATERIALS ON FILE	
(a)	Evidence in a Hearing	
(b)	Disciplinary Documentation	
(c)	Documentation to be Destroyed	
10.07	NO ORAL WARNING OR REPRIMAND AS DISCIPLINE	
ARTIC	LE 11 - SENIORITY	47
11.01	DEFINITION AND PURPOSE	47
11.01	ENTITLEMENT	
11.02	ACCRUAL	
(a)	Seniority Accrual	
(b)	Special Circumstances	
(9)	- r - · · · · · · · · · · · · · · · · ·	

11.04	MAINTENANCE	48
(a)	Maintained for Duration	
(b)	Maintained Twelve Months	
(c)	Maintained Five Working Days	
(d)	Accrual Conditions	
11.05	LOSS	
(a)	Discharged	
(b)	Maximum Amount Allowed	
(c)	Loss of Employment Status	
(d)	Severance Pay	49
11.06	ON APPOINTMENT TO NEW POSITION	49
11.07	SENIORITY LIST	49
11.08	DETERMINING SENIORITY FOR EMPLOYEES HIRED ON THE SAME DAY	
ARTIC	CLE 12 - LAYOFF	50
12.01	DEFINITION	50
	Type 1 (Involuntary Cessation of Work)	
(a)	Type 2 (Bridge Period)	
(b)	Type 3 (Normal Expiry of Temporary Appointment)	
(c)		
12.02	TYPE 1 LAYOFF (INVOLUNTARY CESSATION OF WORK)	
12.03	NOTICE	
(a)	Temporary Layoff (Type 1)	
(b)	Permanent Layoff (Type 1)	
12.04	ELECTIONS	
(a)	Upon Receipt of Notice of Layoff	
(b)	Human Resources Department to Provide Pertinent Information	
12.05	NOTE: TIME LINES	
12.06	TEMPORARY LAYOFF OPTIONS	
(a)	Bumping	
(b)	Layoff Exceeding 15 Days	
(c)	Employee May Exercise a Bump	
(d)	15 Calendar Days Notice of Displacement	
(e)	Employee With 9100 Hours of Seniority	
(f)	Employee With Less Than 9100 Hours of Seniority	
(g)	Continuation at Pay Step	
(h)	Employee Who Bumps Shall be on Probation	
(i)	Bumped Employee May Also Exercise Right to Bump	
(j)	Regular Employee Bumps Into Temporary Position	
(k)	Recall	
(1)	Process	54
12.07	PERMANENT LAYOFF OPTIONS - BUMPING	54
(a)	Employee Must Advise University of Precise Position	54
(b)	University Has Right to Maintain Efficient Staff	
(c)	Upon Notice of Permanent Layoff	
(d)	Employee May Exercise a Bump	
(e)	45 Calendar Days' Notice of Displacement	
(f)	Effective Date of Bump Prior to Assuming Position	
(g)	Employee Who Fills Job Through Bumping	
(b)	Employee Who Bumps Shall be on Probation	
(i)	Regular Employee Failing Probation in Bumped Position	
(j)	Employee Who Bumps Into Temporary Position	
(k)	Successful Bump Into Position Held by Temporary Employee	
(1)	Employee May Reconsider Bumping Choice During Probation	
(n)	Displaced Employee Options.	
12.08	PERMANENT LAYOFF OPTIONS - JOB VACANCY/RETRAINING	
12.08 (a)	Job Vacancy	
(a) (b)	Upon Election of Job Vacancy	
	Possession of Knowledge, Skills and Abilities	
(c) (d)	No Need for Retraining	
	-	
(e) (f)	Need for Retraining Commencement of Retraining	
	Method of Retraining	
(g)	weinou of renatiling	

(h)	Reconsideration Upon Completion of Retraining	
(i)	Successful Completion of Retraining	
(j)	Retrained Employee on Probation	
(k)	Priority Rights for Permanent Vacancies	
(1)	Following Expiry of Notice Period	
12.09	PERMANENT LAYOFF OPTIONS - SEVERANCE PAY	
(a)	Severance Pay	
(b)	Loss of Employment Status	
(c)	Monies Advanced to be Severance Pay	
12.10	HEALTH BENEFITS MAINTENANCE DURING TEMPORARY LAYOFF	
(a)	Regular Employee on Temporary Layoff to Pay Premiums	
(b)	University to Reimburse Employee for University's Share Upon Return to Work	
12.11	TEMPORARY LAYOFF BECOMES PERMANENT	
(a)	Process	
(b)	Choices	
(c)	Exception	
(d) 12.12	Reactivation of Former Position	
	TYPE 2 LAYOFF (BRIDGE PERIOD)	
(a)	Rights	
12.13	TYPE 3 LAYOFF (NORMAL EXPIRY OF TEMPORARY APPOINTMENT)	
(a)	Rights	
12.14	PRIORITIES	
(a)	Priority I	
(b)	Priority II	
(c)	Priority III	
(d)	Priority IV Exception	
(e) (f)	Seniority	
12.15	INFORMATION TO BE GIVEN TO UNION	
	Layoff List	
(a) (b)	Recall List	
ARTIC	LE 13 - HOURS OF WORK	64
13.01	WORK DAY AND WORK WEEK	64
13.01 (a)	WORK DAY AND WORK WEEK Work Day	64 64
13.01 (a) (b)	WORK DAY AND WORK WEEK Work Day Work Week	64 64 64
13.01 (a) (b) 13.02	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK	64 64 64 64
13.01 (a) (b) 13.02 (a)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day	
13.01 (a) (b) 13.02 (a) (b)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week	
13.01 (a) (b) 13.02 (a) (b) 13.03	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY	
13.01 (a) (b) 13.02 (a) (b) 13.03 (a)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week	
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Day	
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Day Modified Work Day	$ \begin{array}{c}                                     $
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Day Modified Work Day Modified Work Week Administrative Head to Consider Employee's Request	$ \begin{array}{c}                                     $
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week. MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week. Modified Work Week. Modified Work Day Modified Work Week. Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week. MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week. Modified Work Week. Modified Work Week. Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS Exceptional Work Week	$\begin{array}{c} & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 65 \\ \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b)	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week. MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week. Modified Work Week. Modified Work Day Modified Work Week. Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS	$\begin{array}{c} & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 64 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 65 \\ & & 66 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a)	WORK DAY AND WORK WEEK. Work Day. Work Week STANDARD WORK DAY AND STANDARD WORK WEEK. Standard Work Day. Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS Exceptional Work Week Restrictions Approval	$\begin{array}{c}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05	WORK DAY AND WORK WEEK Work Day Work Week STANDARD WORK DAY AND STANDARD WORK WEEK Standard Work Day Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS Exceptional Work Week Restrictions Approval APPROVAL OF HOURS OF WORK	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c)	WORK DAY AND WORK WEEK. Work Day. Work Week STANDARD WORK DAY AND STANDARD WORK WEEK. Standard Work Day. Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS Exceptional Work Week Restrictions Approval	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a)	WORK DAY AND WORK WEEK	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b)	WORK DAY AND WORK WEEK	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 67 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07	WORK DAY AND WORK WEEK. Work Day Work Week. STANDARD WORK DAY AND STANDARD WORK WEEK. Standard Work Day Standard Work Week. MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS. Exceptional Work Week. Restrictions Approval APPROVAL OF HOURS OF WORK. Approval Changes. MEAL BREAK. REST PERIODS.	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 67 \\ & 67 \\ \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06	WORK DAY AND WORK WEEK. Work Day. Work Week STANDARD WORK DAY AND STANDARD WORK WEEK. Standard Work Day. Standard Work Week MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS Exceptional Work Week Restrictions Approval APPROVAL OF HOURS OF WORK Approval Changes MEAL BREAK REST PERIODS Two Fifteen Minute Rest Periods	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 67 \\ & 67 \\ & 67 \\ & 67 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07 (a)	WORK DAY AND WORK WEEK. Work Day Work Week. STANDARD WORK DAY AND STANDARD WORK WEEK. Standard Work Day Standard Work Week. MODIFIED WORK WEEK AND MODIFIED WORK DAY Modified Work Week Modified Work Week Administrative Head to Consider Employee's Request EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS. Exceptional Work Week. Restrictions Approval APPROVAL OF HOURS OF WORK. Approval Changes. MEAL BREAK. REST PERIODS.	$\begin{array}{c} & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 64 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 65 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 66 \\ & 67 \\ & 67 \\ & 67 \\ & 67 \\ & 67 \\ & 67 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07 (a) (b)	WORK DAY AND WORK WEEK	$\begin{array}{c} & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 64 \\ & & & 65 \\ & & & 65 \\ & & & 65 \\ & & & 65 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 66 \\ & & & 67 \\ & & & 67 \\ & & & 67 \\ & & & 67 \end{array}$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) (d) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07 (a) (b) (c)	WORK DAY AND WORK WEEK. Work Day	$\begin{array}{c} & & & & & & & & & & & & & & & & & & &$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) 13.04 (a) (b) 13.05 (a) (b) 13.06 13.07 (a) (b) (c) 13.08	WORK DAY AND WORK WEEK	$\begin{array}{c} & & & & & & & & & & & & & & & & & & &$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (b) (c) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07 (a) (b) (c) 13.08 (a) (b) (c)	WORK DAY AND WORK WEEK. Work Day	$\begin{array}{c} & & & & & & & & & & & & & & & & & & &$
13.01 (a) (b) 13.02 (a) (b) 13.03 (a) (c) 13.04 (a) (b) (c) 13.05 (a) (b) 13.06 13.07 (a) (b) (c) 13.08 (a) (b) (c)	WORK DAY AND WORK WEEK	$\begin{array}{c} & & & & & & & & & & & & & & & & & & &$

13.10	MINIMUM DAILY PAY	
13.11	STANDBY DUTY	
(a)	Scheduled Standby Duty	68
(b)	Advance Notice	
(c)	Consecutive Weekends	
13.12	EXCEPTIONAL WORK DAY	
13.13	EMPLOYEE'S REQUEST FOR REDUCTION OF HOURS	69
13.14	EMPLOYER'S REDUCTION OF HOURS	69
ARTIC	CLE 14 - HOURLY PREMIUMS	
14.01	SHIFT PREMIUM	70
(a)	6:30 pm - 6:30 am	
(b)	3:00 pm - 12:01 am	70
14.02	SHORT CHANGE-OVER	70
ARTIC	CLE 15 - OVERTIME	71
15.01	DEFINITIONS	
(a)	Standard or Non Standard Work Week Hours	
(b)	Modified Work Week Hours	
(c)	Straight Time	
(d)	Double Time	
(e)	Compensating Time Off.	
15.02	RIGHT TO REFUSE OVERTIME	
15.03	OVERTIME COMPENSATION	
(a)	Non-Standard Work Day	
(b)	Double Time	
(c)	Modified Work Week	
(d)	Work on Other Than a Regularly Scheduled Day	
(e)	Paid Holiday	
(f)	Compensating Time Off.	
(g) (h)	Overtime Bank Leaves of Absence	
15.04	CALL BACK	
(a)	Additional Overtime	
(b)	Call Back to Work	
(c)	Temporary Employees	
(d)	Transportation Difficulties	
15.05	OVERTIME BREAKS	
15.06	SHARING OF OVERTIME	
15.07	OVERTIME MINIMUM	
15.07	OVERTIME INFORMATION	
	TRAVEL	
15.09		
(a)	International Travel Domestic Travel	
(b) (c)	Travel Insurance	
ARTIC	CLE 16 - HOLIDAYS	75
16.01	PAID HOLIDAYS	75
(a)	Standard Paid Holidays	
(b)	Other Paid Holidays	75
(c)	Additional Paid Holidays	
16.02	HOLIDAYS FALLING ON A SATURDAY OR SUNDAY	75
(a)	Saturday or Sunday	75
(b)	Other Than Saturday or Sunday	75
(c)	Modified Work Week	
(d)	Facility Closed	
16.03	EMPLOYEES REQUIRED TO WORK ON A PAID HOLIDAY	76
16.04	HOLIDAY COINCIDING WITH A DAY OF VACATION	76
16.05	PAYMENT FOR HOLIDAYS	
(a)	Regular Employees	76
(b)	Other Employees	76

(c)	Modified Work Week	77
(d)	Lieu Days	77
(e)	Paid Holiday and Lieu Day	
16.06	OTHER DAYS	
(a) (b)	Not Paid Holidays Facility Closed	
(b) (c)	Payment	
(d)	Request to Work	
. ,	CLE 17 - ANNUAL VACATION	
17.01	CALENDAR YEAR	70
17.01	VACATION ENTITLEMENT	
(a)	Appointments Less Than Four Months	
(b)	Vacation Entitlement	
(c)	Incomplete Calendar Year	79
(d)	Temporary Employees	79
(e)	Regular Employees	
17.03	MODIFIED WORK WEEK	
17.04	GRANTING OF VACATION LEAVE	
17.05	SPLIT VACATIONS	
17.06	VACATION SUBSTITUTION OVERLAP OF VACATION WITH OTHER LEAVES OF ABSENCE	
17.07 17.08	VACATION CARRY-OVER	
17.08 (a)	Annual Vacation	
(b)	Vacation Bank	
(c)	Excess Vacation Bank	
(d)	Bridge Period Employees	
17.09	PRO-RATA ENTITLEMENT	
17.10	NO CALL BACK FROM VACATION OR BRIDGE PERIOD WITHOUT EMPLOY.	EE'S CONSENT
	81	
17.11	NO CHANGE IN SCHEDULED VACATION OR BRIDGE PERIOD WITHOUT EM	IPLOYEE'S
	ENT	81
17.12	ENT IMPACT OF TEMPORARY PROMOTION	
	ENT	
17.12 17.13	ENT IMPACT OF TEMPORARY PROMOTION	81 82 82
17.12 17.13	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION	
17.12 17.13 ARTIC	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay	
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness	
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee	
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02	ENTIMPACT OF TEMPORARY PROMOTION IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY	
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03	ENTIMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP)	
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD	81 82 82 83 83 83 83 83 83 83 83 83 83 83 83 83
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS	81 82 82 83 83 83 83 83 83 83 83 83 83 83 83 83
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement	81 82 82 83 83 83 83 83 83 83 83 83 83 83 83 83
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement REDUCED HOURS, PHASED-IN, RETURN TO WORK	81 82 82 83 83 83 83 83 83 83 83 83 83 83 83 83
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION <b>CLE 18 - SICK LEAVE</b> SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work	81 82 82 83 83 83 83 83 83 83 83 83 83 83 83 83
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement REDUCED HOURS, PHASED-IN, RETURN TO WORK	81         82         82         83         84         84         85         85         85
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Work Return to Work	81         82         82         83         84         84         85         85         85         85
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION <b>CLE 18 - SICK LEAVE</b> SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Work Return to Position Rights, Obligations and Entitlements	81         82         82         83         84         84         85         85         85         85         85         85         85         85         85         85         85         85         85         85         85         85         86
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b>	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Position Rights, Obligations and Entitlements Payments from Workers' Compensation	81         82         82         83         84         84         85         85         85         85         85         86         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION <b>CLE 18 - SICK LEAVE</b> SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness. Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Position Rights, Obligations and Entitlements Payments from Workers' Compensation <b>CLE 19 - CAREER DEVELOPMENT</b> PURPOSE	81         82         82         83         84         84         85         85         85         85         85         86         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b>	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Position Rights, Obligations and Entitlements Payments from Workers' Compensation	81         82         82         83         84         84         85         85         85         85         85         85         86         87         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01 (a)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION <b>CLE 18 - SICK LEAVE</b> SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Position Rights, Obligations and Entitlements Payments from Workers' Compensation <b>CLE 19 - CAREER DEVELOPMENT</b> PURPOSE Dispersal of Funds	81         82         82         83         84         84         84         85         85         85         85         86         87         87         87         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01 (a) (b)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP) RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS. Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Work Return to Position Rights, Obligations and Entitlements Payments from Workers' Compensation. CLE 19 - CAREER DEVELOPMENT PURPOSE Dispersal of Funds Carry Forward	81         82         82         83         84         84         84         85         85         85         85         85         85         85         86         87         87         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01 (a) (b) 19.02	ENT IMPACT OF TEMPORARY PROMOTION	81         82         82         83         84         84         85         85         85         85         85         85         85         85         85         86         87         87         87         87         87         87         87         87         87         87         87         87         87         87          87          87        <
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01 (a) (b) 19.02 19.03 (a) (b)	ENT IMPACT OF TEMPORARY PROMOTION NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION CLE 18 - SICK LEAVE SICK LEAVE ENTITLEMENT. Sick Leave With Pay Proof of Illness Pregnant Employee SICK LEAVE AND SHORT TERM DISABILITY JOINT EARLY INTERVENTION PROGRAM (JEIP). RETURN TO WORK FROM LTD LONG TERM REPLACEMENTS. Regular Status Replacement. REDUCED HOURS, PHASED-IN, RETURN TO WORK Return to Work Return to Work Return to Work Return to Work Return to Work Return to Work CLE 19 - CAREER DEVELOPMENT PURPOSE Dispersal of Funds Carry Forward. TRAINING LEAVE - EMPLOYER REQUIRED. TRAINING LEAVE - EMPLOYEE REQUESTED. Request for Career Development Eligibility.	81         82         82         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         84         84         85         85         85         85         85         85         85         85         85         86         87
17.12 17.13 <b>ARTIC</b> 18.01 (a) (b) (c) 18.02 18.03 18.04 18.05 (a) 18.06 (a) (b) (c) (d) <b>ARTIC</b> 19.01 (a) (b) 19.02 19.03 (a)	ENT IMPACT OF TEMPORARY PROMOTION	81         82         82         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         83         84         84         84         85         85         85         85         85         85         85         85         85         86         87         87         87         87         87         87         87         87         87         87         87         88         88          88          88          88

(e)	Discretion of University	
(f)	Prohibition	
19.04	LIMITATION	
19.05	DEFERRED SALARY LEAVE	
(a)	Applying for Deferred Salary Leave	89
(b)	Approval of Deferred Salary Leave	89
(c)	Funding for Deferred Salary Leave	
(d)	Period for Salary Deferral	
(e)	University Obligation	
(f)	Administrative Expenses	
(g)	Regulations	
(h)	Taking the Leave of Absence	
(i)	Health and Welfare Benefits	
19.06	APPRENTICESHIP	90
ARTIC	LE 20 - SPECIAL AND OTHER LEAVE	91
20.01	BEREAVEMENT LEAVE	91
(a)	Leave	
(b)	Timing	
(c)	Definition of Immediate Family	
(d)	Employee to Advise University	
(e)	Bereavement Leave During Vacation	
20.02	SPECIAL LEAVE WITH PAY	
(a)	Citizenship	
(b)	Mourning	
(c)	Family Responsibility	
20.03	PUBLIC DUTIES	
(a)	Seeking Election (Provincial or Federal)	
(b)	Seeking Election (Municipal)	
(c)	Upon Election	
(d)	Maintenance of Seniority	
20.04	LEAVE FOR COURT APPEARANCES	
(a)	Jury Selection/Jury Duty Court Appearance	
(b) (c)	Pay	
(c) (d)	Jailed Pending Appearance	
20.05	ELECTIONS	
20.05	LEAVE OF ABSENCE WITHOUT PAY	
20.00 (a)	Requesting	
(a) (b)	Emergency Absence	
20.07	PERSONAL LEAVE WITH PAY	
20.08	COMMUNITY EMERGENCY	
20.08	TERMINATION OF LEAVE	
20.09	DONATING BLOOD	
20.10	EXAMINATION LEAVE	
	NO CALL BACK FROM LEAVE OF ABSENCE	
20.12		
<u>20.13</u>	GENDER TRANSITION LEAVE	
20.14	DOMESTIC OR SEXUAL VIOLENCE LEAVE	
ARTIC	CLE 21 - PARENTAL LEAVE	97
21.01	APPLICATION	
21.02	MATERNITY AND <u>PARENTAL</u> LEAVE	
(a)	Prior to the Date of Birth	97
(b)	On the Date of the Birth or Adoption	
(c)	Extension to the Leave	97
<u>(d)</u>	Vacation Accrual While on Leave	
21.0 <u>3</u>	CESSATION OF EMPLOYMENT OR RETURN TO WORK	
(a)	Cessation of Maternity or Parental Leave	
(b)	Return to Employment	98
21.0 <u>4</u>	POST MATERNITY OR PARENTAL JOB POSTING RIGHTS	
(a)	Job Postings/Competitions	98

(b)	Continuation of Status	
(c)	Qualification	
21.0 <u>5</u>	BENEFIT COVERAGE	
21.0 <u>6</u>	VACANCY CREATED BY GRANTING OF PARENTAL OR MATERNITY LEAVE	
(a) (b)	Filling a Vacancy Extension	
21.07	APPLICATION OF SICK LEAVE	
21.0 <u>7</u> 21.0 <u>8</u>	TEMPORARY EMPLOYEES	
21.0 <u>8</u> 21.09	IMPACT OF LEGISLATION	
21.0 <u>7</u> 21.10	SUPPLEMENTAL EMPLOYMENT BENEFIT FOR MATERNITY AND PARENTAL LI	
_	CLE 22 - SAFETY AND HEALTH	
22.01	CONDITIONS	102
22.02	HEALTH AND SAFETY COMMITTEE	
(a)	Definition of Committee	
(b)	Compensation	
(c)	Unsafe Conditions	
22.03	UNSAFE AREAS	
22.04	RECOMMENDATIONS	
22.05	INDUSTRIAL FIRST AID REQUIREMENTS	
22.06	TRANSPORTATION FOR EMPLOYEES REQUIRING MEDICAL CARE	
22.07	SHUTDOWNS	
(a)	Health and Safety	
(b) (c)	Refusal to Work Order of Inspector	
22.08	HEALTH AND SAFETY TRAINING	
ARTI	CLE 23 - WORK CLOTHING	104
23.01	SUPPLY OF WORK CLOTHING	104
23.02	PROTECTIVE CLOTHING	
23.03	MAINTENANCE OF WORK CLOTHING	
ARTI	CLE 24 - TECHNOLOGICAL CHANGE	105
24.01	DEFINITION	105
(a)	Definition	
(b)	Introduction of Equipment	105
24.02	CONSEQUENCES	
(a)	Introduction of Change	
(b)	Training	
(c) (d)	Failure to Complete Training Election Not to be Retrained	106
24.03	DISCUSSION	
24.03	RESULTING AGREEMENTS	
24.05	DISPUTE RESOLUTION	
(a)	Findings	
(b)	Action	
24.06	NON DISPUTE	
(a)	Bumping	
(b)	Vacancy	
(c) (d)	Pay Vacancy per 12.08	
	CLE 25 - APPOINTMENTS	
25.01	DEFINITION	
25.02	JOB POSTINGS	
(a) (b)	Posting	
(b) (c)	Contents of Posting Advertisements	
(d)	Union to Get Copy of Posting	
25.03	APPOINTMENT PROCEDURES	

(a)	Durations of up to One (1) Calendar Month Maximum	
(b)	Durations from One (1) Calendar Month to Three (3) Calendar Months Maximum	
(c)	Duration of Greater than Three (3) Months	
(d)	Precedence of Layoff List	
(e)	Precedence of Permanently Laid Off Employees	
(f)	Rights of Unsuccessful Candidates	
25.04	NOTIFICATION TO UNION	
25.05	RIGHT TO GRIEVE	110
25.06	INTERIM APPOINTMENTS	110
(a)	Grievance	111
(b)	Competition	111
25.07	APPOINTMENTS	111
(a)	Required Knowledge, Skills and Abilities	
(b)	Seniority	111
(c)	Compensation	111
(d)	Job Duties	
25.08	PROBATION	111
(a)	Current Employee	111
(b)	Former Employee	112
(c)	Incomplete Initial Probation Period	112
25.09	PREVIOUS INCUMBENCY	112
(a)	Status	
(b)	Loss of Status	
25.10	EXTERNAL HIRES	112
ARTIC	LE 26 - CLASSIFICATION	113
26.01	CLASS SPECIFICATION	113
(a)	Copy of Class Specification to Employee	
(b)	Copy of Class Specification to Union	
(c)	Prior to Posting	
(d)	Overview of Duties	
(e)	Shall Not Include	
(f)	Integrity to be Maintained	
(g)	University Not Bound	
(b)	Preparation	
26.02	BENCHMARKS	
26.03	CLASSIFICATION REVIEW	
(a)	Request	
(b)	Duties Have Changed	
(c)	Process	
26.04	TEMPORARY RECLASSIFICATION	
26.05	NOTIFICATION	
(a)	Request by Employee	
(b)	Results to Union	
26.06	RIGHT TO APPEAL	
	Disagreement	
(a) (b)	Failure to Resolve Dispute	
(c)	Classification Adjudicator	
26.07	SALARY ON RECLASSIFICATION	
20.07 (a)	Increase in Salary	
(a) (b)	Step Placement	
(0) (c)	Increment, Seniority and Benefits	
(d)	Decrease in Salary	
26.08	ADJUDICATION	
20.08 (a)	Adjudicator	
(a) (b)	Resolution	
(0) (c)	Disagreements	
(d)	Costs	
26.09	JOB CLASSIFICATION REVIEW	
ARTIC	LE 27 - PAYMENT OF WAGES AND ALLOWANCES1	118
27.01	EQUAL PAY	118
27.01		110

PAY DAYS	118
Biweekly Pay	118
Method of Pay	
Higher Pay Group	118
Pay Higher Class Specification	110
Return to Former Position	119
Vehicle	
Driver's License	120
REIMBURSEMENT FOR PARKING EXPENSES WHILE ON UNIVERSITY BUSINESS	121
T F 28 - HEAT TH AND WEI FARE	122
Life Insurance Coverage	125
Health Funding Account	126
Employee Must Maintain Coverage	127
ENROLMENT	
DEDUCTIONS: PREMIUMS AND CONTRIBUTIONS	128
T F 29 - CENERAL CONDITIONS	129
Change in Equipment	
	Biweekk Pay           RATES OF PAY           RATES OF PAY           RATES OF PAY           RATES OF PAY           APPOINTMENT TO NEW POSITION           Higher Pay Group           Pay Group           SUBSTITUTION           Definition           Pay           Higher Class Specification           Calculation of Higher Rate of Pay           Duration           Return to Former Position           MILE AGE ALLOWANCE           EMPLOYEE'S VEHICLE           Vehicle           Driver's License           TRANSPORTATION           PAY MATRIX (INCREMENTS)           REIMBURSEMENT FOR PARKING EXPENSES WHILE ON UNIVERSITY BUSINESS .           LE 28 HEALTH AND WELFARE           BASIC MEDICAL PLAN           LONG TERM DISABILITY           Participation           Coverage and Eligibility           FLEXIBLE BENEFITS PLAN           Eligibility           F

(b)	Renovations	.129
(c)	Consultation	
(d)	Implication of Consultation	
29.04	COMMUNITY FACILITIES	
29.05	ABSENCE WITHOUT LEAVE	
(a)	Absence Without Notification	
(b)	Repeated Absence	
29.06	RETIREMENT	
(a)	Age of Retirement Rights and Privileges	
(b) (c)	Library Card and Access to Facilities	
(d)	Tuition Costs	
(e)	Employee Assistance Plan	
29.07	EARLY RETIREMENT PLAN	131
(a)	Eligibility	
(b)	Agreement	
(c)	Benefit Options	
(d)	Financial Counselling	
29.08	EMPLOYEE FILE	
(a) (b)	Inspection Incidental Information	
29.09	EVALUATION	
29.10	CAPILANO UNIVERSITY COURSES	
(a)	Credit Courses	
(b)	Continuing Education Courses	
(c)	Cost Recoverable/Non-base Funded Courses	
(d)	Time Spent Attending a Course	
(e)	Commencement and Completion	
29.11	MULTIPLE APPOINTMENTS	
29.12	PERSONAL AND SEXUAL HARASSMENT	
(a) (b)	Prohibition Against Personal and/or Sexual Harassment Definition of Personal Harassment	
(0) (c)	Definition of Sexual Harassment	
(d)	Complaints	
(e)	Personal Harassment	
29.13	ELECTRONIC MONITORING	136
29.14	PERSONAL DUTIES NOT REQUIRED	
29.15	PROTECTION AGAINST LEGAL ACTION	
(a)	Protection for Employees	
(b)	Protection for Retired Employees	
(c)	University to Assume Costs	
29.16	LEGISLATION LIE DETECTOR TESTS	13/
29.17		
ARTIC	LE 30 - EMPLOYEE ASSISTANCE PLAN	138
30.01	ADMINISTRATION OF EMPLOYEE ASSISTANCE PLAN	138
ARTIC	LE 31 - DURATION OF CONTRACT	139
31.01	DURATION	130
31.02	NEW AGREEMENT	
31.02	CONTINUATION OF PRESENT AGREEMENT	
31.03	LABOUR RELATIONS ACT 50(2) EXCLUSION	
31.05	EFFECTIVE DATES	
31.06	ERRORS & OMISSIONS	
31.07	PAYMENT OF COLLECTIVE AGREEMENT	
ARTIC	LE 32 - EXPLANATORY NOTES	140
32.01	INTENT AND MEANING	
32.02	OBJECTIVE INTERPRETATION	
32.03	NOTIFICATION REQUIREMENTS	

ARTICLE 33 - WORKLOAD	143
33.01 REASONABLE WORKLOAD	143
33.02 WORKLOAD DISPUTES	
<ul> <li>33.03 JOINT WORKLOAD REVIEW COMMITTEE</li> <li>(a) Complaints</li> </ul>	
(b) Recommendations	143
ARTICLE 34 - WORKERS' COMPENSATION SUPPLEMENT	144
34.01 LEAVE OF ABSENCE	
34.02 WORKERS' COMPENSATION SUPPLEMENT	
APPENDIX A	
MEMBERSHIP APPLICATION AND	
UNION DUES DEDUCTION AUTHORIZATION	
APPENDIX B	
CLASS SPECIFICATION INFORMATION AND CRITERIA	
APPENDIX C(A)	
APPLICATION AND WAIVER FORM RE: REGULAR EMPLOYEES BENEFITS	
APPENDIX C(B)	
APPLICATION AND WAIVER FORM	
RE: TEMPORARY EMPLOYEES BENEFITS	
APPENDIX D	151
CARRIER'S APPEAL PROCESS	151
TERMINATION OR DENIAL OF A DISABILITY CLAIM	151
APPENDIX F – SALARY SCALE EFFECTIVE JULY 1, 2020 – JUNE 30, 2021 <sup>1</sup>	153
APPENDIX G – SALARY SCALE EFFECTIVE JULY 1, 2021 – JUNE 30, 2022 <sup>1</sup>	154
APPENDIX N - EXCLUSIONS	155
LETTER OF UNDERSTANDING 1	158
RE: TESTING	158
LETTER OF UNDERSTANDING 2	160
RE: UNIVERSITY CLOSURE POLICY	160
LETTER OF UNDERSTANDING 3	161
RE: ARTICLE 19 CAREER DEVELOPMENT	161
LETTER OF UNDERSTANDING 4	162
RE: EXCEPTIONAL WORK DAY – <u>CENTRE FOR SPORT &amp; WELLNESS</u>	162
LETTER OF UNDERSTANDING 5	163
RE: LABOUR MARKET ADJUSTMENT FOR SELECTED IT SYSTEMS POSITIONS	163
LETTER OF UNDERSTANDING 6	164
RE: EARLY CARE AND LEARNING RECRUITMENT AND STRATEGY FUNDING	164
LETTER OF UNDERSTANDING 7	165
RE: TARGETED PRESCRIPTION SPENDING ACCOUNT (TPSA)	165
LETTER OF UNDERSTANDING 8	167

RE: TEMPORARY VACATION CARRY-OVER INCREASE	167
LETTER OF UNDERSTANDING 9	
RE: STRATEGIC FUNDING FOR FRONT LINE SUPPORT	
LETTER OF UNDERSTANDING 10	
<u>Re: Support Staff Innovation Fund</u>	

# **COLLECTIVE AGREEMENT**

THIS AGREEMENT is effective July 1, 2019 to June 30, 2022 and entered into in <u>February</u> 2020.

BETWEEN:

CAPILANO UNIVERSITY, in the Province of British Columbia, (hereinafter called the "University" or the "University Board")

## OF THE FIRST PART

AND:

<u>MoveUP (</u>CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378),

(hereinafter called the "Union")

#### OF THE SECOND PART

WHEREAS the University Board is an employer within the meaning of the Labour Relations Code of British Columbia;

AND WHEREAS the Union is a Trade Union within the meaning of the said Act and is the bargaining authority for that group of employees engaged as staff at Capilano University and its various centres:

NOW THIS AGREEMENT WITNESSETH that the parties hereto agree each with the other as follows:

## ARTICLE 1 - PREAMBLE

## 1.01 PURPOSE OF AGREEMENT

The purpose of this Agreement is to set forth and establish the terms and conditions of employment so that efficient operations and harmonious relationships may be maintained between the University and the Union to the benefit of both Parties and the community they serve.

## 1.02 EXISTING LEGISLATION

The Parties recognize and agree that they cannot be obligated or bound by any term, condition or provision which would be contrary to any existing Federal or Provincial legislation. In the event that any term, condition, or provision, or part thereof, which is incorporated into this Agreement, whether by inadvertence, error, or misunderstanding, is in fact or in law contrary to such Federal or Provincial legislation, then such term, condition or provision or part thereof, is void and of no effect.

## **1.03 FUTURE LEGISLATION**

In the event that future Federal or Provincial legislation makes invalid any provision of this Agreement, the remaining provisions shall remain in effect for the term of the Agreement. The University and the Union shall confer to settle upon a mutually agreeable provision to be substituted for the provision so altered or invalidated.

Should the parties not reach agreement within 45 working days, either party can make a referral to arbitration under article 7.08.

#### 1.04 HUMAN RIGHTS

#### (a) Application

The Parties agree that the provisions of Section 8 of the Human Rights <u>Code</u> of British Columbia, <u>RSBC 1996</u>, <u>Chapter 210</u>, apply as though included in and forming part of this Agreement, and for the purposes of this Agreement personal life style, sexual orientation, physical disability unrelated to job performance, previous and current psychiatric treatment unrelated to job performance and number of dependants shall not constitute reasonable cause for the purposes of Sub Section 1 of the said Section 8.

## (b) Unfair Labour Practices

The Parties acknowledge and agree to observe the prohibitions contained in the Unfair Labour Practices section of the Labour Relations Code of British Columbia.

## **1.05 NO OTHER AGREEMENT**

#### (a) Sole Bargaining Agent

The University recognizes the Union as the sole bargaining agent for all employees covered by the Union's certification.

## (b) No Other Agreement

The University agrees not to enter into any agreement with any employee or group of employees which conflicts with any of the terms or conditions of this Agreement or which provides for any terms or conditions of employment which are not expressly provided for by this Agreement, except where the Agreement expressly permits such individual agreement.

## **1.06 CORRESPONDENCE**

The Parties recognize and agree that unless specifically indicated to the contrary, all notifications and correspondence directed to the Union office shall be addressed to the Union, and those directed to the University shall be addressed to the Vice President, Human Resources.

## 1.07 HEADINGS

The headings and sub-headings used in this Agreement are inserted for convenience and reference purposes only and shall not be used as an aid to interpretation.

## **ARTICLE 2 - UNION RECOGNITION AND NEGOTIATION**

## 2.01 BARGAINING UNIT DEFINED

The bargaining unit is as in the Certification issued by the Labour Relations Board of British Columbia on December 17, 1974, as amended on March 31, 1983, November 18, 1986, and on February 20, 2009.

## 2.02 MANAGEMENT RIGHTS

## (a) **Rights of University**

The Union recognizes and agrees that except as specifically abridged, delegated, granted or modified by this Agreement, all of the rights, powers and authority the University had prior to the signing of this Agreement are retained solely and exclusively by the University, and remain without limitation within the rights of management, which rights are not subject to the grievance procedure or arbitration.

## (b) Affairs and Facilities

Without limiting the generality of the foregoing, the University reserves the sole and exclusive right to operate and manage its affairs and facilities in all respects as it sees fit, including the right to hire, discipline or discharge for just and sufficient cause, and to make and alter from time to time rules and regulations to be observed by the employees, except that this right shall not supersede any other express provisions of this Agreement.

#### (c) Fair and Reasonable

The University agrees that, in the exercise of its rights as set out above, it must act in a fair and reasonable manner.

#### 2.03 NEW POSITIONS

#### (a) Notification of Union

When the University creates any new position during the term of this Agreement, except those positions included in another Union's certification, or instructional administrators to whom faculty report, the Union shall be notified immediately and, at the request of either party, the parties will meet to discuss whether or not the position is covered by the Union's certification. In the event of disagreement, the Labour Relations Board will be asked to decide the issue, if referred by either party.

#### (b) Disagreement

In any event, if the Parties cannot agree within five (5) days of the notice of the new position, the University may fill the position in which case it shall establish an interim salary group, and shall extend to any employee appointed, the same terms and conditions applicable as if the Union Certification covered the position, pending the agreement of the Parties or a decision of the Labour Relations Board.

## 2.04 SALARY FOR NEW POSITIONS

Where the Parties agree or the Labour Relations Board decides that a new position is included within the unit, a salary grouping for the position will be discussed. If the Parties cannot agree within five (5) days of the decision that the new position is included within the unit, on a salary grouping applicable to the position, the University may establish a salary group for it temporarily and the grievance procedure shall apply and the issue may proceed to final and binding arbitration if necessary.

## ARTICLE 3 - UNION SECURITY

## 3.01 EMPLOYEES AT DATE OF CERTIFICATION

All persons within the certification of the Union as of December 17, 1974 who were members of the Union on or before that date shall remain members of the Union as a condition of continuing employment.

#### **3.02 NEW EMPLOYEES**

As a condition of employment, all new employees shall become Union members immediately upon their date of hire and shall remain Union members for the duration of their employment.

## 3.03 CHANGES IN STATUS

At the time an employee is hired or when <u>their</u> status is subsequently changed, the University agrees to notify the union with a copy of <u>their Offer Letter</u>.

## ARTICLE 4 - UNION RIGHTS AND ACTIVITY

#### 4.01 NOTICE OF ABSENCE FROM WORKPLACE

Whenever possible, an employee exercising any right arising under Article 4 shall advise <u>their</u> supervisor of any absence from <u>their</u> workplace. When undertaking union duties, union stewards will not interrupt the work of other union employees without the approval of the appropriate administrator.

## 4.02 **RECOGNITION OF UNION REPRESENTATIVES**

The University shall recognize the Union Stewards, members of the Union Grievance Committee, and all persons authorized to act on behalf of the Union, in accordance with the Union Constitution.

#### 4.03 CONTACTING AT WORK

The authorized representatives of the Union shall have the right to contact employees at work on matters respecting this Agreement and its administration. The supervisor of the department being entered for the purpose of contacting employees shall be advised by the Union representatives of the time of their entry and exit whenever possible.

#### 4.04 TIME OFF FOR UNION BUSINESS AND ACTIVITIES

The Union and the University will co-operate to see that suitable arrangements are made to have an employee's regular duties covered in a reasonable manner so as to avoid interfering with the operating requirements. When undertaking union duties, union stewards will not interrupt the work of other union employees without the approval of the appropriate administrator. Requests for leaves of absence without loss of seniority for the following activities shall not be unreasonably withheld:

#### (a) Without Pay - Conventions

without pay to a reasonable number of representatives of the Union to attend conventions of the Union and bodies to which the Union is affiliated when requested in writing by the Union naming the employees for whom the request applies;

#### (b) Without Pay - Union Business

without pay for a reasonable number of representatives of the Union to attend to Union business which requires them to leave their premises of employment when requested in writing by the Union naming the employees for whom the request applies;

## (c) Without Pay - Bargaining Committee Meetings

without pay for a reasonable number of employees who are representatives of the Union on a Bargaining Committee to attend meetings of the Bargaining Committee when requested in writing by the Union naming the employees for whom the request applies;

## (d) With Pay - Negotiations

with pay to three employees who are representatives of the Union on the Bargaining Committee to leave their employment to carry on negotiations with the University, when requested in writing by the Union naming the employees for whom the request applies;

#### (e) With Pay - Steward Duties

with pay to named Stewards to perform duties necessary for the administration of this Agreement;

#### (f) With Pay - Grievance

with pay to employees called to appear as a party or as a witness in a grievance hearing or at an arbitration;

#### (g) With Pay - Union Grievance Committee

with pay for up to three members of the Union Grievance Committee when such formally meets pursuant to Article 7;

#### (h) Without Pay - Union Position

without pay to an employee to hold a position with the Union or its affiliates. Time spent with the Union will be considered as service with the University and the employee will continue to accrue seniority with the University during such period. At the commencement of the leave, an employee will advise the University whether or not they wish to participate in the University Benefit Plans. It is understood that participation is subject to approval of the carrier of the Plan. At the Union's request, the University will act as paymaster during the leave period and the Union will reimburse the University for the cost of such salary and benefits premiums. If the duration of the leave exceeds 8 months, the University may fill the position and the person so chosen shall be considered as having temporary full-time or temporary part-time status notwithstanding the categorization otherwise applicable under Article 9.

#### 4.05 UNIVERSITY AND UNION SHALL ACQUAINT NEW EMPLOYEES

#### (a) Agreement in Effect

The University agrees to acquaint new employees with the fact that an Agreement is in effect, with the conditions of employment set out in the Articles dealing with Union Security and Check-off of Union Dues.

#### (b) Union Steward to Acquaint

A Union Steward shall interview each new employee within regular working hours, without loss of pay to Steward or employee, for the purposes of acquainting the new employee with the benefits and duties of Union membership and to provide the employee with a copy of the Collective Agreement. The length of the interview shall be kept to a reasonable time (e.g. one-half hour to an hour).

## 4.06 RIGHT TO REFUSE TO CROSS PICKET LINES

## (a) Dispute

Failure to cross a legally constituted picket line which arises out of a labour dispute under the Labour Relations Code of British Columbia shall not be considered a violation of this Agreement nor grounds for disciplinary action.

#### (b) Pay

Employees should not expect to receive pay for work not performed in exercising any right given by the clause.

#### 4.07 UNION GENERAL MEETING

The University shall allow employees a two hour break without pay once every month so that they may attend the Union general meeting. The Union general meeting shall be scheduled at the same time as one of the University meeting block periods.

#### 4.08 USE OF UNIVERSITY ROOMS

The University shall allow the Union to book University rooms through the regular booking procedures of the University, for meetings of Union Committees, the General Membership, and other such meetings.

## 4.09 UNION OFFICE SPACE

The University will provide a location for a Union office subject to the availability of space to first meet priority institutional requirements as determined by the University. The University will provide appropriate notice to and consultation with the Union regarding any change to the office location.

## 4.10 APPLICATION TO OVERTIME PREMIUM

Time expended by an employee for the purpose of Articles 4.03, 4.04, 4.05 and 4.07 shall not be used for the purpose of computing any overtime or premium entitlement contained in this Agreement.

## 4.11 UNION INSIGNIA

## (a) Wearing or Displaying

Union members shall have the right to wear or display the recognized insignia of the Union.

## (b) Inclusion in Documents

The recognized insignia of the Union shall include the designation "MoveUP". This designation shall, at the employee's option, be placed on any document which is prepared or processed in any form or manner, in whole or in part, by any member of the Union. This designation shall be placed at the bottom of the last page of any such document.

#### 4.12 INFORMATION TO BE ROUTINELY PROVIDED TO THE UNION

Once every month the University shall provide to the Union a list of employees within the bargaining unit setting out the employees':

- address
- telephone number
- payroll number

this provision is in addition to the University's obligations under Articles 9.09(c) and 25.04.

Once every four (4) months the University shall provide the Union with a list of employees within the bargaining unit setting out their seniority in terms of hours worked.

## 4.13 UNION COMMUNICATIONS AND VOTING

(a) The University agrees that the Union shall have the right to use the University's <u>email</u> to communicate with the employees in the bargaining

unit. Employees shall have the right to use the University's <u>email</u> to communicate with the Union. Both the Union and the employees shall at all times adhere to the University's policies regarding <u>email</u> usage. Failure to do so will cause this provision to come to an end. The University shall provide the Union and each employee in the bargaining unit with a copy of said policies and any updates thereto.

(b) It is agreed that the Union shall have the right to place ballot boxes in the workplaces of the University for the purposes of conducting Union elections, referenda, polling or collective agreement votes.

## ARTICLE 5 - CHECK-OFF OF UNION FEES AND DUES

## 5.01 AUTHORIZATION

As a condition of employment, all employees on hire shall be required to sign an authorization (Appendix "A") for deduction of Union dues from their wage or salary. A copy of the completed authorization shall be sent to the Treasurer of the Union.

## 5.02 **DEDUCTION OF DUES**

#### (a) Schedule

The University shall begin the deduction of dues from the first regular pay to which the employee is entitled.

#### (b) University's Obligation

The Union recognizes and agrees that the University's obligation to deduct dues is expressly restricted to making only such deductions as are permitted by law and by the valid authorization of each employee.

## 5.03 COLLECTION AND FORWARDING OF DUES

The University shall forward the collected dues by cheque together with a list of employees' names and identification numbers, monthly and year-to-date amounts deducted to the Treasurer of the Union within five (5) working days after the last day of each month in which the dues are collected.

#### 5.04 NOTIFICATION

The Union shall advise the University in writing of all dues required by the Union and of any changes which occur in connection with such dues.

#### 5.05 INDEMNITY

The Union shall indemnify the University and hold it harmless against any and all suits, claims, demands and liabilities that shall arise out of, or by reason of any action taken or not taken by the University for the purposes of complying with any provision of this Article.

## 5.06 **REVOCATION**

The Union agrees that, should any employee revoke <u>their</u> assignment, the University must forthwith cease to make deductions. Upon written instruction by the Union, the University agrees that, where an existing employee revokes <u>their</u> authorization, the employee's appointment shall be terminated forthwith. The University will provide the Union with a copy of any revocation of authorization received. Termination under this provision will not constitute a layoff under Article 12.

# 5.07 YEAR-END SUMMARY

Except by reason of circumstances beyond the control of the University, by January 31st each year the University shall supply to the Union a list of total deductions on behalf of the Union made for each employee during the previous year.

## 5.08 RECORD OF UNION DEDUCTIONS (T4 SLIPS)

The University shall supply each employee, without charge, with a record for income tax purposes indicating the amount of applicable deductions the University has paid to the Union on behalf of the employee in the previous calendar year. Absent exceptional circumstances, such record shall be provided to each employee prior to March 1 of the succeeding calendar year.

## **ARTICLE 6 - UNIVERSITY - UNION RELATIONS**

#### 6.01 **REPRESENTATIONS**

No employee or group of employees shall act as Union representatives on University committees, to which the Union may be entitled to send representatives, without proper authorization of the Union.

#### 6.02 SEARCH COMMITTEE - ADMINISTRATOR

Whenever the University strikes a search committee that has representatives provided by the Capilano University Faculty Association, representatives of the Union will also be invited to be part of that search committee.

The number of representatives from any other constituency shall not exceed the number of Union representatives on the committee.

#### 6.03 ACCESS TO UNIVERSITY PREMISES

#### (a) Negotiation and Grievance

The University agrees that access to its premises shall be given to members of the Union when dealing with or negotiating with the University, as well as for the purpose of investigating and assisting in the settlement of a grievance.

#### (b) Space

In order to facilitate the orderly, as well as the confidential, investigation of specific grievances, the University shall make available to Union Representatives or Stewards temporary use of an office or similar facility.

## 6.04 INFORMATION

The University agrees to provide the Union with all available non-confidential information relating to employees as may be required by the Union for collective bargaining purposes. The Union recognizes that certain information is confidential to the University and Government at specific periods of time, during which such information may not be released to the Union. The University shall provide the Union with such information if and when it becomes available.

#### 6.05 UNION - UNIVERSITY RELATIONS

The Union and the University recognize the mutual value of ongoing joint discussions in matters pertaining to working conditions, employment, employee classifications, services, and labour relations.

#### (a) Joint Standing Committee

There shall be established a Joint Standing Committee composed of the President of the University or designate(s) and the President of the Union or

designate(s) and one additional member appointed by both sides. Decisions of the Joint Standing Committee must be unanimous.

# (b) Meetings

The Committee shall meet at the call of either party at a mutually agreeable time and place. Unless otherwise agreed, the quorum for Joint Standing Committee meeting shall be two (2) University representatives and two (2) Union representatives. The University and Union will make every effort to ensure two (2) people will be available for any meeting.

## (c) Agenda Items

The Committee shall submit recommendations to the Parties for ratification on matters specifically referred to it by mutual agreement of the University and the Union, including reviewing matters, other than grievances, relating to the maintenance of good relations between the Parties.

## 6.06 EMPLOYEE ATTENDANCE AT MEETINGS

The Parties recognize that employees attend meetings pursuant to their job assignment or pursuant to their authorized representation on behalf of the Union.

## (a) Required to Attend

When an employee is required by the University to attend a meeting in either of the above capacities, regardless of when the meeting occurs, <u>they</u> shall receive pay for attending the meeting.

## (b) Invited to Attend

When an employee is invited by the University, but not required, to attend a meeting pursuant to <u>their</u> authorized representation on behalf of the Union, <u>they</u> shall receive pay for the meeting if it occurs at a time when <u>they</u> would otherwise have worked. If the meeting occurs at a time when the employee would not otherwise have worked, <u>they</u> shall not receive pay for attending the meeting.

## ARTICLE 7 - GRIEVANCE

## 7.01 DEFINITION OF A GRIEVANCE

A grievance shall be defined as any dispute or controversy between the University and the Union or between the University and one or more of its employees covered by this Agreement in respect of any matter involving the interpretation, application or administration of any provision of this Agreement; any matter involving the alleged violation of the Agreement; any matter arising out of the employment relationship affecting or involving employees covered by this Agreement; or any question as to whether any matter is grievable or arbitrable.

#### 7.02 RIGHT TO GRIEVE

- (a) Any employee who considers herself themselves aggrieved shall have the right to initiate and to process a grievance under this Agreement, subject to the consent of the Union, in which case the Union shall at all times control carriage of the grievance on behalf of the employee.
- (b) The Union shall have the right to initiate and to process a grievance under this Agreement on behalf of itself, or on behalf of any employee, or on behalf of any group of employees.
- (c) The University shall have the right to initiate and to process a grievance under this Agreement with respect to the Union's actions.
- (d) It is mutually agreed that any employee or party exercising her their or its rights under this Agreement does so without prejudice to her their or its relations with any employee or party or representative of either party.

#### 7.03 SELECTION GRIEVANCES

Within ten (10) working days of being notified by the Human Resources Department that <u>they were</u> an unsuccessful candidate for a posted job vacancy, under the procedure of 25.02, unless the job vacancy was filled by an employee pursuant to Articles 12 and 24, an employee may grieve the University's decision at Step 3 of the grievance procedure by forwarding such a grievance to the Human Resources Department.

#### 7.0<u>4</u> GRIEVANCE PROCEDURE

An employee shall have Union Representation present at all grievance meetings after Step 1. All grievances shall be dealt with in the following manner:

#### 7.0<u>5</u> COMPLAINT STAGE - STEP 1

An aggrieved employee or <u>their</u> Union Representative may discuss <u>their</u> complaint with <u>their</u> local supervisor. If the dispute is not resolved orally, the Union may formalize the grievance through submission to Step 2. When a complaint is being discussed at Step 1, the thirty (30) day time limit in Step 2 with respect to the initiation of the grievance will be frozen.

# 7.0<u>6</u> GRIEVANCE STAGE - STEP 2

## (a) Complaint Submitted to Management

Should a complaint be unresolved at Step 1, the complaint may be submitted by the Union Steward or Union Representative in writing to the appropriate management representative, no later than thirty (30) calendar days after the employee was notified verbally or in writing, or <u>they</u> first became aware of the action or circumstance giving rise to the grievance, with a copy to the Human Resources Department. Such submission may be undertaken without first proceeding through Step 1.

## (b) Management and Union Steward Meet

The Manager and the Union Steward or Representative shall meet within seven (7) working days and attempt a resolution of the grievance. The grievor may attend this meeting.

## (c) Dispute Not Resolved

If the Manager and the Union Steward or Representative are unable to resolve the dispute at Step 2, the Union or the University may refer the matter to Step 3 within fourteen (14) calendar days of the expiry of the time limit under 7.06(b).

# 7.0<u>7</u> STEP 3

Within ten (10) working days of referral to Step 3, the Parties will meet to discuss the grievance. If discussion at this meeting leads to a unanimous resolution of the grievance such resolution shall become the decision at Step 3. Should no resolution be reached at the meeting, the University will render its decision on the grievance within fifteen (15) working days of the meeting at Step 3.

## 7.08 NON-RESOLVED GRIEVANCES

## (a) Referral to Arbitration

Grievances that have not been resolved through the grievance procedure may be referred by either party to Arbitration, with written notice to the Parties involved. The referral to Arbitration must be made within thirty (30) calendar days of the expiry of the time limit provided in Step 3.

#### (b) Time Limit

The same time limit provided in  $7.0\underline{8}(a)$  shall apply to recourse to any procedures pursuant to the Labour Relations Code of British Columbia.

## 7.09 TECHNICAL OBJECTIONS TO GRIEVANCES

No grievance shall be defeated merely because of a technical error in processing the grievance. Reasonable amendments mutually agreed upon in writing may be allowed at any step, the intention being that matters in dispute be dealt with in a fair and equitable way.

#### 7.<u>1</u>0 RECORD OF GRIEVANCE

Copies of all formal written grievances and all formal replies shall be sent to the Union office and the Human Resources Department without delay. Resolutions shall be acted upon by all Parties.

## 7.<u>11</u> TIME LIMITS

## (a) Extension

The time limits prescribed for the performance of any step in the Grievance Procedure may be extended by mutual agreement of the Parties involved at each step.

## (b) Substance

The time limits specified in the Grievance Procedure shall not be deemed as technical errors but points of substance.

## 7.12 GRIEVANCE OF GENERAL APPLICATION OR INTERPRETATION

Where a dispute involving a question of general application or interpretation of the Agreement occurs; or where a group of employees has a common grievance; or where the Union as a whole or the University has a grievance, Step 1 and Step 2 shall be by-passed and the dispute shall be referred directly to Step 3 of the Grievance Procedure.

Such reference must be made no later than thirty (30) calendar days after the group of employees, the Union, or the University became aware of the action or circumstance giving rise to the grievance.

## 7.1<u>3</u> GRIEVANCE INVOLVING DISCIPLINE

In the case of a dispute arising from an employee's reprimand, suspension or discharge, Step 1 and Step 2 shall be by-passed and the dispute shall be taken directly to Step 3 of the Grievance Procedure.

# 7.1<u>4</u> EMPLOYEE TIME OFF TO ATTEND TO GRIEVANCE

Except in the case of discharge or suspension, an employee shall be permitted the necessary time off with pay to attend to the adjustment of a grievance and shall have the right to be present at all stages of the procedure if so requested by either party. An employee whose period of suspension is over and who has returned to work shall have the full entitlement of this clause for time off from work with pay.

# 7.1<u>5</u> MAINTENANCE OF EMPLOYEE RIGHTS AND BENEFITS

## (a) Settlements

Settlements reached at any step of the Grievance Procedure shall be applied in accordance with the settlement or to the date set by the Arbitrator.

## (b) Maintenance of Position

Except in the case of a dismissal or suspension grievance, an employee shall maintain <u>their</u> position with no loss of pay, rank, seniority, benefits or privileges during the Grievance Procedure or Arbitration.

## 7.16 DISCLOSURE OF INFORMATION

The parties agree to facilitate the processing of grievances by exchanging the relevant information and materials in a timely manner.

# 7.1<u>7</u> EFFECT OF SETTLEMENTS

Where the Employer and the Union agree to the settlement of a grievance, such settlement shall be in writing and shall be final and binding on both Parties and each Employee in the bargaining unit affected by the settlement.

# 7.18 DEVIATION FROM GRIEVANCE PROCEDURE

- (a) The University will not enter into discussion, communication or negotiation of any kind with respect to a grievance with the grievor(s) once a grievance has been initiated by the Union without the prior, express written consent of the Union.
- (b) In the event that, after having initiated a grievance through the grievance procedure, an employee endeavours to pursue the same matter by any other legal means, the Union agrees that, pursuant to this Article, the grievance shall be considered to have been abandoned on a "without prejudice" basis to any new grievance that should arise in the future.
- (c) The grieving party may at its discretion by written notice withdraw any grievance at any time without prejudice to its position in future with respect to any new grievance that should arise in the future.

# **ARTICLE 8 - ARBITRATION**

# 8.01 ARBITRATORS

All grievances submitted to arbitration shall be adjudicated by a single arbitrator selected on a rotational basis, depending on availability, from the following list:

- 1. John McConchie
- 2. John Hall
- 3. Joan Gordon
- 4. Colin Taylor
- 5. Julie Nichols
- 6. David McPhillips

## 8.02 ARBITRATION

The Arbitrator may determine <u>their</u> own procedure in accordance with the Labour Relations Code of British Columbia and shall give full opportunity to all Parties to present evidence and make representations. They shall hear and determine the dispute or allegation and shall make every effort to render a decision within reasonable time.

## 8.03 DECISION OF ARBITRATOR

The decision of the Arbitrator shall be final and binding on both Parties. The Arbitrator shall not make any award contrary to the conditions or articles of this Agreement, or in amendment to this Agreement.

# 8.04 EXPENSES OF ARBITRATION

Both Parties to the Arbitration shall pay for all their own expenses and one-half of the fees and expenses of the Arbitrator.

#### 8.05 DIFFERENCES

Where a difference arises between the Parties relating to the dismissal, discipline, or suspension of an employee, or to the interpretation, application, operation or alleged violation as to whether a matter is arbitral, during the term of this Agreement an Arbitrator shall, at the request of either party:

#### (a) Investigation

investigate the differences;

# (b) Definition

define the issue in the difference, and

## (c) Recommendation

make written recommendations to resolve the difference within five (5) days of the date of the receipt of the request; and for those five (5) days from that date, time does not run in respect of the grievance procedure.

# 8.06 EXPEDITED ARBITRATION

For the purpose of accelerating the resolution of applicable grievances, the Parties may mutually agree to refer to Expedited Arbitration any matter properly processed, as a grievance, in accordance with the provisions of the grievance procedure contained in this Agreement.

# (a) Arbitrator

An Arbitrator shall be selected from amongst those Arbitrators named in this Agreement to hear the matter in dispute in accordance with the provisions of this Article and must be able to hear the matter within sixty (60) calendar days. If the Arbitrator is unable to meet within the prescribed time the parties will select the next available Arbitrator.

# (b) **Presentation of Facts**

The facts of the matter in dispute shall be presented during Expedited Arbitration by a designated representative of the Union and a designated representative of the Employer, who shall not be lawyers or individuals whose primary employment purpose is the processing of arbitrations.

# (c) Decision

The decision of the Arbitrator shall be of no precedential value and shall not be referred to by either Party in any other proceeding.

# (d) Multiple Grievances

The Parties may, by mutual agreement, refer a group of grievances, related or unrelated, to be heard pursuant to this Clause 8.06 by a single arbitrator.

# (e) Expedited Arbitration

All other provisions of this Article with respect to Arbitration and the Arbitration process shall apply to Expedited Arbitration.

# 8.07 PRE-HEARING CONFERENCE

- (a) A party may request the holding of a pre-hearing conference at a time and place to be fixed by the arbitrator, which conference may be held by telephone, if convenient.
- (b) The arbitrator may on <u>their</u> own initiative direct that a pre-hearing conference be held.
- (c) A pre-hearing conference shall consider the simplification of the issues, the possibility of obtaining admissions which might facilitate the hearing and any other matters that may aid in the speedy and efficient disposition of the matters in dispute between the parties.
- (d) Following a pre-hearing conference, the arbitrator may make an order reciting the results of the conference and giving such directions as <u>they</u> <u>think</u> just. The order shall control the subsequent course of the proceedings unless modified at the arbitration hearings to prevent injustice.
- (e) The arbitrator who presides at the pre-hearing conference is seized of the matter in dispute.
- (f) The provisions of the Clause 8.07 shall also apply with respect to any Expedited Arbitration under this Agreement.

# **ARTICLE 9 - CATEGORIES OF EMPLOYEES**

## 9.01 EMPLOYEE DEFINED

Employee shall mean all clerical, technical and service employees who are covered by the Certification granted to the Union by the British Columbia Labour Relations Board on December 17, 1974 as amended on March 31, 1983, November 18, 1986, and February 20, 2009 except those excluded pursuant to Appendix N who shall be considered exempt for the purposes of this Agreement.

It is recognized that other positions which are in dispute may be referred by either party to the Labour Relations Board of British Columbia for a decision pursuant to the Labour Relations Code.

## 9.02 **REGULAR FULL TIME (RFT)**

Regular Full Time (RFT) shall mean any employee who works for thirty-five (35) hours per week for eight (8) months of the year or more, for an indefinite or recurring term, or an employee who fills a position regularized under Article 9.07.

## 9.03 **REGULAR PART TIME (RPT)**

Regular Part Time (RPT) shall mean any employee who works for less than thirtyfive (35) hours per week for eight (8) months of the year or more, for an indefinite or recurring term, or an employee who fills a position regularized under Article 9.07.

## 9.0<u>4</u> TEMPORARY FULL TIME (TFT)

Temporary Full Time (TFT) shall mean any employee who works for thirty-five (35) hours per week for a definite term of up to twelve (12) months.

#### 9.0<u>5</u> TEMPORARY PART TIME (TPT)

Temporary Part Time (TPT) shall mean any employee who works for less than thirty-five (35) hours per week for a definite term of up to twelve (12) months.

#### 9.0<u>6</u> EXCEPTIONS

The Parties agree that circumstances may be such that the operational requirements of the University are best served by an appointment with a definite term in excess of twelve (12) months.

Extensions or renewals of appointments beyond the twelve (12) month term will be by mutual written agreement of the Parties.

## 9.07 **REGULARIZATION**

In preparation for the review set out below, the University will provide the Union with a copy of all temporary staffing activities summarized by classification and work area. This information will be provided to the Union by January 15 of each

year covering the previous calendar year. The list will contain the following information: position, position number, incumbent, employee category, start date and end date. The University will provide a complete listing annually by January 15<sup>th</sup> of each year.

# (a) Review

The University and the Union shall meet to jointly review all temporary staffing activities of the previous calendar year by February 15th each year. Where agreement can be reached between the parties on whether or not a position should be regularized, a joint recommendation shall be made to the University President by February 28th each year.

# (b) Failure to Reach Agreement

If the Parties cannot reach agreement on whether or not a position should be regularized, the University President shall meet with the Union Representatives to hear the reasons for the Union's recommendation.

# (c) Recommendations

Recommendations that are concurred with by the University President shall be implemented and will be effective not later than the first working day of September in the current year.

# 9.08 FILLING REGULARIZED POSITIONS

When a Regular Full Time or Regular Part Time position is created as a result of the exercise of Article 9.07 by the University, it shall be filled pursuant to the provisions set out in Article 25.

# 9.09 EMPLOYEE APPOINTMENTS

## (a) Acknowledgement

At the time of employment and appointment, the employee and the University shall acknowledge, through the completion by the University and acceptance by the employee of a<u>n Offer Letter</u>, the following:

relevant date of commencement duration of appointment approved work day and work week pay group position (where applicable) employee category. position number

# (b) Weekend Work

At the time of hiring, the University shall indicate if weekend work may become a requirement of the hire.

# (c) <u>Offer Letter</u> Copy

A copy of the completed <u>Offer Letter</u> shall be provided to the Union within seven (7) calendar days of employment, appointment or any change to the conditions listed at Article 9.09(a).

## 9.10 WORK PRACTICUMS

The Union recognizes and agrees that from time to time during the term of this Agreement there will be work performed by students as a part of their work experience practicums.

# (a) Definition

When such students undertake practicum work experiences which may occur at the University, and when the duties undertaken while on such practicums might otherwise have been performed, had they been undertaken, by employees of the University covered by the Union certification; the parties agree that

Such students:

- (i) are not employees of the University;
- (ii) are not within the Union certification;
- (iii) will not be paid.

# (b) Employee Responsibilities

Such employees:

- (i) may assist such students to assimilate to the work location through orientation, supervision and training, as necessary;
- (ii) who undertake such orientation, supervision and training during student practicums shall not be eligible for reclassification based on these functions alone.

# (c) Limitations

Student practicums shall be limited to one student per semester per work area and up to ten students in the Children's Centre per semester for a period of normally not longer than 250 hours in total, per student, per semester. Additional students may be requested with the consent of the Union.

# (d) Notification to Union

The University shall notify the Union in advance of the number of students being placed for practicum work experience within the University.

# 9.11 WORK EXPERIENCE PLACEMENTS/CAREER AND PERSONAL PLANNING PROGRAMS

Given the initiatives of the Provincial Government regarding Work Experience Programs for students in Grades 11 and 12, the University and the Union agree to the following in respect to the offering of "Work Experience Placements" and "Job Shadowing Opportunities" at the University associated with this particular program.

- (a) Students shall be enrolled in a secondary school level program.
- (b) Participating students shall not replace, displace, delay the filling of a vacant budgeted/approved position in the bargaining unit, or replace leaves of absence.
- (c) Duration of participation of any one student in the Career Preparation option shall not exceed 100 hours in any 12 month period.
- (d) Duration of participation of any one student in the Career and Personal Planning option shall not exceed 30 hours in any 12 month period.
- (e) Students shall not receive any form of remuneration for tasks performed, nor be entitled to become members of the Union.
- (f) The employee assigned to oversee, or provide a "shadowing" opportunity to a student shall be provided with adequate time to do so.
- (g) The University will complete a Work Experience Placement Partnership Agreement form for each student with each school district participating in the program, which will also be signed by the local Union Representative (Job Steward or Executive Councillor) and a copy forwarded to the Union office.
- (h) This Agreement may be cancelled by either the Union or the University providing 30 days notice, and is without prejudice to either's position regarding the offering of such placements.

# 9.12 STUDENT EMPLOYMENT PROGRAMS

- (1) <u>The parties acknowledge the value to Capilano students of being able to</u> participate in Student Employment Programs on campus and the positive impact such programming has on overall student recruitment and retention.
- (2) The parties recognize that Student Employment Programs provide practical learning opportunities for students' within their academic programs and are of value in assisting students in obtaining work experience and financial support as part of the educational experience.
- (3) The parties recognize and agree that students will not displace MoveUP members. The University further recognizes that in areas where services are provided by MoveUP members, students can be employed to only complement the services provided and not for the purpose of eroding the scope of the bargaining unit or replacing any employee(s) in the bargaining unit, including but not limited to, any such employee who is on layoff status under the Collective Agreement.
- (4) <u>Prior to the posting of any new student position the University shall notify</u> <u>the Union in writing of the nature of work to be performed, work</u> <u>location(s), duration of employment and hours of work per week.</u>
- (5) <u>The Union shall have the right to refer any violation of Article 9.12</u> <u>directly to expedited arbitration pursuant to the applicable provisions of the</u> <u>Collective Agreement.</u>

# ARTICLE 10 - DISCIPLINE

## 10.01 **PROCEDURE**

No employee shall be disciplined except for just cause.

### 10.02 24 HOURS ADVANCE NOTICE

The University agrees to notify an employee twenty-four (24) hours in advance of any interview of a disciplinary nature and shall indicate:

#### (a) Accompanied by Steward

their right to be accompanied by a Union Steward or Union Representative;

## (b) **Purpose of Meeting**

the purpose of the meeting, including whether it involves the employee's personnel file;

# (c) Access to Personnel File

that if the employee personnel file is to be considered during the interview and the employee grants <u>their</u> permission, the employee and <u>their</u> Union Steward or Union Representative shall, before the meeting, have access to this file;

## (d) **Right to Refuse Participation**

the employee has the right to refuse to participate or to continue to participate in such interview unless <u>they have</u> received the notice hereinabove provided for.

## 10.03 SUSPENSIONS SHALL BE IN WRITING

All suspensions shall be in writing from Administrative Heads (as defined in Appendix N) with all reasons for the action contained therein. The Union shall be notified at the time the notice is given.

## 10.04 DISCHARGES

All discharges shall be on the written authority of the President with all reasons for the action contained therein.

#### **10.05 BURDEN OF PROOF**

In cases of discipline, the burden of proof of just cause shall rest with the University. The charge submitted by the University shall be limited to information in the written notice given.

## 10.06 MATERIALS ON FILE

# (a) Evidence in a Hearing

The University shall not introduce as evidence in a hearing any document from an employee's file the existence of which <u>they were</u> not aware.

# (b) Disciplinary Documentation

An employee shall be given a copy of any disciplinary documentation placed on <u>their</u> file and may have <u>their</u> comments relating to the documentation placed on <u>their</u> file. Should <u>they</u> wish to dispute any such entry <u>they</u> shall be entitled to recourse through the grievance procedure. When disputes are resolved in favour of the employee, at the request of the employee, the University shall remove all references to the disciplinary action. No inference regarding discipline shall be made from an action or inaction by either the employee or the University.

# (c) Documentation to be Destroyed

Upon the employee's request, any documentation placed on file pursuant to 10.06(b) shall be removed from the file and destroyed in the presence of the employee after expiration of eighteen (18) months from the date it was issued, provided there has been no further disciplinary documentation relating to the same issue (SEE ALSO 29.08).

# 10.07 NO ORAL WARNING OR REPRIMAND AS DISCIPLINE

An oral warning or reprimand shall not be deemed to be a disciplinary measure and shall not be reported in any employee's personnel file as described in Clause 10.06(b) or otherwise referred to by the University in any arbitration or other legal proceeding except for the purpose of showing the matter was brought to the employee's attention prior to formal discipline.

# ARTICLE 11 - SENIORITY

## 11.01 **DEFINITION AND PURPOSE**

Seniority, which is the length of an employee's service within the bargaining unit as a whole, shall be used to distinguish between employees who would otherwise be considered equal.

### 11.02 ENTITLEMENT

All employees shall accrue seniority from their date of hire.

The basis of accrual shall be 152 hours for each month of active employment; prorated for part time employees.

## 11.03 ACCRUAL

#### (a) Seniority Accrual

Seniority shall continue to accrue for the duration of the following circumstances and as set out in other provisions of this agreement:

- (i) all leaves with pay;
- (ii) L.T.D., W.C.B., or Parental Leave (Maternity, <u>Parental</u> or Adoption Leave);
- (iii) exercising a right to refuse to cross a picket line as per Article 4.06(a);
- (iv) conscription;
- (v) University excludes position and employee successfully competes back into unit within one (1) year of exclusion;
- (vi) University excludes position and Union succeeds in getting jurisdiction back;
- (vii) government excludes position and Union succeeds in getting jurisdiction back;
- (viii) leave without pay to act as full time Officer or representative with the Union.
- (ix) while on the Layoff List (TYPE 1 LAYOFF);
- (x) on suspension without pay;
- (xi) while not working due to strike or lockout.

# (b) Special Circumstances

Seniority shall continue to accrue for the first ninety (90) calendar days only for the following circumstance:

all leaves without pay (including temporary leave to accept an excluded position), except as expressly provided otherwise by Clause 11.03(a) above.

# 11.04 MAINTENANCE

# (a) Maintained for Duration

Seniority shall be maintained for the duration of the following circumstance:

- (i) bridge periods (TYPE 2 LAYOFF);
- (ii) excess of ninety (90) days per Article 11.03 (b);

# (b) Maintained Twelve Months

Seniority shall be maintained for a maximum period of twelve (12) calendar months for the following circumstance:

(i) following the expiry of an appointment having a definite duration (TYPE 3 LAYOFF);

# (c) Maintained Five Working Days

Seniority shall be maintained for a maximum period of five (5) working days for the following circumstance:

(i) after resignation;

## (d) Accrual Conditions

If in a maintenance period employment results in the accrual of more than thirty-five (35) hours of seniority, a new maintenance period shall commence at the termination of that employment. If, however, such employment accrues thirty-five (35) or fewer hours of seniority, the seniority total at the end of the original maintenance period reverts to an accumulation of only those hours worked within that period.

## 11.05 LOSS

An employee or former employee shall lose seniority and be removed from the seniority list in the following circumstances:

## (a) Discharged

if discharged for just cause;

## (b) Maximum Amount Allowed

seniority has been maintained to the maximum amount allowable under Article 11.04;

# (c) Loss of Employment Status

on loss or termination of employment status (subject to express exceptions set out in this agreement);

# (d) Severance Pay

on acceptance of severance pay.

# 11.06 ON APPOINTMENT TO NEW POSITION

An employee who is appointed to a new position shall carry <u>their</u> previous accumulation of seniority to such position.

# 11.07 SENIORITY LIST

The University shall maintain a seniority list.

The seniority list shall be revised every four (4) months. The list shall show the date upon which each employee's service commenced and the total seniority each employee has accrued. The Union shall be supplied with an electronic copy.

# 11.08 DETERMINING SENIORITY FOR EMPLOYEES HIRED ON THE SAME DAY

When two (2) or more employees commence work with the University on the same day their relative seniority shall be determined by a method of random selection mutually agreed between the University and the Union.

# ARTICLE 12 - LAYOFF

# 12.01 DEFINITION

A layoff is a cessation of employment which may occur in the following ways:

## (a) Type 1 (Involuntary Cessation of Work)

An involuntary cessation occurring during the term of an employee's appointment which may be of a temporary nature to a maximum of one hundred and twenty-three (123) calendar days, or may be of a permanent nature.

# (b) Type 2 (Bridge Period)

Predetermined time span of non-work occurring between active segments of an employee's appointment. Bridge periods may occur in a continuous regular appointment, or in a temporary duration appointment.

# (c) Type 3 (Normal Expiry of Temporary Appointment)

A predetermined cessation occurring at the end of a temporary hire having an expiry date designated at the commencement of an employee's appointment.

Other cessation of employment including suspension, dismissal, resignation and leaves of absence are not included in the layoff definition.

## 12.02 TYPE 1 LAYOFF (Involuntary Cessation of Work)

An involuntary cessation occurring during the term of an employee's appointment which may be of a temporary nature to a maximum of one hundred and twenty-three (123) calendar days, or may be of a permanent nature.

# **12.03 NOTICE**

# (a) Temporary Layoff (Type 1)

- (i) Subject to operational requirements and after consultation with the Union, the University shall determine in which positions layoff shall occur after, when possible, reducing its temporary workforce through the expiry of their appointments.
- (ii) Subject to 12.03(a)(i), layoff shall be in the reverse order of seniority, first for Temporaries, and then for Regular status employees.
- (iii) The University shall attempt to minimize the option of bumping by selecting for layoff positions those incumbents who have either the least seniority or have shorter appointments, or have both.

- (iv) If a temporary layoff occurs the University shall inform the Union and the employees to be laid off, in writing, not less than thirty (30) calendar days before the layoff is to commence. The notice shall state the date the layoff shall commence.
- (v) An employee shall be considered laid off at the date established under 12.03(a)(iv) above.

# (b) Permanent Layoff (Type 1)

- (i) The University shall provide the Union and affected employee(s) with not less than ninety (90) calendar days notice, in writing, of its intention to introduce a permanent layoff.
- (ii) At the time notice is provided, the University shall provide the Union with the reasons for the introduction of the permanent layoff and with as much information related to it as possible.

# 12.04 ELECTIONS

# (a) Upon Receipt of Notice of Layoff

An employee who has received notice of <u>their</u> layoff must, prior to the expiry of the notice period, elect one of the following options:

- (i) if the layoff is of a temporary duration:
  - (1) bumping, or
  - (2) layoff list;
- (ii) if the layoff is of a permanent duration:
  - (1) bumping,
  - (2) a job vacancy or retraining for a job vacancy, or
  - (3) severance pay.

# (b) Human Resources Department to Provide Pertinent Information

It is the responsibility of the Human Resources Department to provide to the employee all information pertinent for making the election.

The election made under this Article must be forwarded, in writing, to the Human Resources Department and to the Union. Any employee not exercising this election within the given time limit shall be deemed to have elected the Layoff List if the layoff is of a temporary nature, or Severance Pay if the layoff is of a permanent nature.

# 12.05 NOTE: TIME LINES

Employees are cautioned to carefully monitor time lines set out in 12.03(a)(iv) and 12.03(b)(i). For example, if the bumping option is chosen, a decision should be made early after notice is received to avoid the possibility of being on leave without pay awaiting the opportunity to access position bumped. It must be remembered that bumped employees are also entitled to notice periods (see 12.06(d), 12.07(e) and 12.07(f)).

# 12.06 TEMPORARY LAYOFF OPTIONS

## (a) Bumping

It is understood and agreed that the application of this procedure shall not affect the University's right to maintain an efficient staff.

# (b) Layoff Exceeding 15 Days

If it is anticipated the layoff will exceed fifteen (15) calendar days, or if an employee has been laid off for fifteen (15) calendar days, <u>they</u> may bump as follows:

- (i) A Temporary employee laid off prior to the expiry of <u>their</u> appointment shall have the right to bump into a position held by a less senior Temporary employee.
- (ii) An employee on a probationary period as defined in Article 25.08 shall have no bumping rights.
- (iii) A Regular employee shall have bumping rights into any position held by a less senior Regular employee or any position held by a Temporary employee.

# (c) Employee May Exercise a Bump

An employee may exercise a bump provided they meet:

the criteria set out in 12.06(b), and

the normal staffing criteria set out in Article 25.07. Where an employee is denied an appointment under this process, the University shall advise <u>them</u> and the Union in writing of the reasons for such denial.

# (d) 15 Calendar Days Notice of Displacement

A bump may only take place after the employee to be bumped has received fifteen (15) calendar days notice of <u>their</u> displacement.

# (e) Employee With 9100 Hours of Seniority

An employee with 9100 or more hours of seniority who fills a job through bumping shall be paid at the greater of <u>their</u> old pay group or the pay group of the position <u>they have</u> bumped into.

# (f) Employee With Less Than 9100 Hours of Seniority

An employee with less than 9100 hours of seniority who fills a job through bumping shall be paid at the pay group of the position <u>they have</u> bumped into.

# (g) Continuation at Pay Step

In both 12.06(e) and 12.06(f) above, the employee shall continue to be placed at the pay step <u>they</u> held prior to the bump, and shall be subject to the benefits and other conditions of employment attendant to the position <u>they</u> bumped into.

# (h) Employee Who Bumps Shall be on Probation

An employee who bumps shall be on a probationary period of one month. At the discretion of the University, the probationary period may be extended in one (1) month periods for no longer than a total probationary period of three (3) months. If during the probationary period the employee finds the job unsatisfactory or is unable to meet the basic job requirements to the satisfaction of the University, <u>they</u> shall be placed on the Layoff List of the classification/work area from which <u>they were</u> laid off.

# (i) Bumped Employee May Also Exercise Right to Bump

An employee who has been bumped by another may in turn exercise <u>their</u> right to bump subject to application of the same conditions set out in this Article.

# (j) Regular Employee Bumps Into Temporary Position

A Regular employee who has bumped into a Temporary position shall have a recurring bumping right with the expiry of the Temporary position. A Temporary who has bumped into a Temporary position shall have <u>their</u> status expire with the expiration of the position should <u>they</u> hold it at that time. A residual right to go on <u>their</u> Layoff List shall exist to the employee at the expiry of the Temporary position <u>they</u> bumped into.

# (k) Recall

(i) A Layoff List shall be established for each classification/work area from which an employee is laid off.

(ii) An employee shall remain on a Layoff List until recalled or until their layoff becomes permanent, whichever is sooner.

# (l) Process

- (i) When a position is reactivated, the most senior employee on the Layoff List for that position shall be offered active status. If <u>they</u> refuse the recall, the next most senior employee will be offered the active status, and so on, until the Layoff List is exhausted. For the purposes of recall only, employees who have bumped are deemed to be simultaneously on the Layoff List.
- (ii) As applicable, it shall be the joint decision of the Administrators responsible for the areas of the position being recalled and the position the recalled employee holds to determine the time lines of the recall.
- (iii) An employee must respond to a recall notice within five (5) calendar days of being contacted by the University. In circumstances where the University fails to contact the employee by telephone, contact shall be deemed to have been made on delivery or attempted delivery by registered mail or courier to the address last given to the University by an employee. Subject only to extenuating circumstances, failure to respond to a recall shall result in the negation of all rights to the position. However, an employee shall continue to hold seniority rights for one (1) year from their last active day of employment.
- (iv) If a recalled position is not successfully staffed through the recall procedure, it shall be staffed through the Article 25 appointment procedure.

# 12.07 PERMANENT LAYOFF OPTIONS - Bumping

## (a) Employee Must Advise University of Precise Position

An employee who has elected to bump must advise the University at the time of the election of the precise position <u>they wish</u> to bump into.

# (b) University Has Right to Maintain Efficient Staff

It is understood and agreed that the application of this procedure shall not affect the University's right to maintain an efficient staff and that an employee must meet the requirements for the position into which they wish to bump.

# (c) Upon Notice of Permanent Layoff

An employee who has received notice of a permanent layoff selecting this option may bump as follows:

- (i) A Temporary employee permanently laid off prior to the expiry of <u>their</u> appointment shall have the right to bump into a position held by a less senior Temporary employee.
- (ii) An employee on a probationary period as defined in Article 25.08 shall have no bumping rights.
- (iii) A Regular employee shall have bumping rights into any position held by a less senior Regular employee, or any position held by a Temporary employee.

# (d) Employee May Exercise a Bump

An employee may exercise a bump provided they meet:

- (i) the criteria set out in 12.07(c), and
- (ii) the normal staffing criteria set out in Article 25.07. Where an employee is denied an appointment under this process, the University shall advise <u>them</u> and the Union in writing of the reasons for such denial.

# (e) 45 Calendar Days' Notice of Displacement

A bump may only take place after the employee to be bumped has received forty-five (45) calendar days' notice of <u>their</u> displacement.

# (f) Effective Date of Bump Prior to Assuming Position

If the effective date of the bump is prior to the date when the employee who has successfully bumped may assume <u>their</u> new position, the intervening time shall be considered as leave without pay.

## (g) Employee Who Fills Job Through Bumping

An employee who fills a job through bumping shall be subject to the benefits and other conditions of employment attendant to the new position.

However, they shall be paid based on the higher of their previous pay group or the pay group of the new position (in each case at their previous step) for the first ninety (90) calendar days in the new position. After this period, they shall be paid based on the pay group of the new position at the same step they held in their former position.

## (h) Employee Who Bumps Shall be on Probation

An employee who bumps shall be on a probation period of one (1) month which may, at the discretion of the University, be extended in one (1) month periods for no longer than two (2) additional months.

# (i) Regular Employee Failing Probation in Bumped Position

A Regular employee who bumps into a position held by a less senior Regular employee, and is subsequently rejected on probation, may choose a second bump. Any time lapse between <u>their</u> rejection on probation and when <u>they</u> may assume <u>their</u> new position shall be considered as leave without pay. If <u>they are</u> found unable to meet the needs of this second position during its probationary period, <u>they</u> shall have no further priority rights under this provision. However, such an employee shall retain <u>their</u> seniority for a nine (9) month period from the expiry of <u>their</u> original notice period.

# (j) Employee Who Bumps Into Temporary Position

An employee who bumps into a position held by a Temporary employee and is subsequently rejected on probation shall have no further priority rights under this provision. However, such an employee shall retain <u>their</u> seniority for a nine (9) month period from the expiry of <u>their</u> original notice period.

# (k) Successful Bump Into Position Held by Temporary Employee

An employee who successfully bumps into a position held by a Temporary employee shall on the expiry of the temporary appointment hold previous incumbency status for that classification/work area. (i.e. <u>their</u> seniority is valid for one year from <u>their</u> last day of employment).

# (I) Employee May Reconsider Bumping Choice During Probation

An employee may reconsider <u>their</u> bumping choice during the probation period and either receive severance pay as provided in Article 12.09 or retain seniority for nine (9) months from the expiry of the original notice period and receive severance pay at the expiry of this period if other employment is not found. The University may deduct all salary paid to the employee during the probation period from the amount of severance pay.

# (m) Displaced Employee Options

An employee who has been displaced by the bumping process (either directly by an employee who received Notice of Permanent Type 1 Layoff or indirectly by another displaced employee) may at any time after receiving notice of <u>their</u> own displacement elect one of the options set out in 12.04(a)(ii) within forty-five (45) calendar days of that notice and follow the same procedure and criteria set out in 12.07, 12.08, 12.09.

## 12.08 PERMANENT LAYOFF OPTIONS - Job Vacancy/Retraining

## (a) Job Vacancy

A Temporary employee may elect a temporary job vacancy, and a Regular employee may elect any job vacancy which the University is staffing.

# (b) Upon Election of Job Vacancy

An employee who has elected a job vacancy must advise the University at the time of election of the precise job vacancy for which <u>they wish</u> to apply. It is understood that the employee may make such election at any time during the notice period, "or within the nine (9) month period following the expiry of the notice period," and that the employee may or may not require training for such job vacancy.

# (c) Possession of Knowledge, Skills and Abilities

When an employee has elected a job vacancy, <u>they</u> shall have first right to it provided <u>they</u> already possess the required skills and abilities, or provided that <u>they</u> will possess the required skills and abilities for the job vacancy upon completion of retraining.

# (d) No Need for Retraining

When an employee elects a vacancy for which <u>they do</u> not need retraining, <u>they</u> shall be appointed to the position upon the expiry of the notice period (Article 12.03(b)(i)), or sooner if possible, or as applicable during the nine (9) month period following the expiry of the notice period, and shall be paid based on the higher of <u>their</u> previous pay group or the pay group of the new position for the first three (3) months in the position. After three (3) months in the position, <u>they</u> shall be paid based on the pay group of the position (in all cases at <u>their</u> previous step).

# (e) Need for Retraining

Retraining, if necessary, must commence after the expiry of the notice period and be accomplished within four (4) months.

# (f) Commencement of Retraining

By mutual agreement, the retraining may begin prior to this date. If retraining is not immediately available upon expiry of the notice period, or if an employee requests that retraining be delayed, any elapsed time between the end of the notice period and the start of retraining shall be leave of absence without pay. However, in circumstances where retraining is not immediately available, the University may consider it reasonable to extend the notice period until such training is available.

# (g) Method of Retraining

Determination of the method of retraining and its cost shall be the responsibility of the University. An employee shall receive <u>their</u> former salary during retraining. Wherever possible, an employee shall be advised in writing of these costs prior to the start of <u>their</u> retraining.

# (h) Reconsideration Upon Completion of Retraining

If during or immediately upon completion of retraining the employee wishes to reconsider <u>their</u> earlier election, or if <u>they do</u> not successfully complete retraining, <u>they</u> shall at that time have the option of receiving severance pay as provided in Article 12.09. The University may deduct all costs of retraining, including the employee's salary while retraining, from the amount of severance pay.

# (i) Successful Completion of Retraining

Upon successful completion of retraining, the employee shall be appointed to the position and shall be subject to all terms and conditions of employment applicable to the position. They shall be paid at the pay group applicable to the position, at the same step as they held in their former position, as of the date of appointment unless their period of retraining was of less than four (4) months duration, in which case they shall be paid the higher of their previous pay group or the pay group of the new position until expiry of four (4) months of combined retraining and being in the position.

# (j) Retrained Employee on Probation

An employee appointed to a vacancy either directly or after retraining shall be on a probationary period of one (1) month. At the discretion of the University, the probationary period may be extended in one (1) month periods for no longer than a total probationary period of three (3) months. If during the probationary period the employee finds the job unsatisfactory or is unable to meet the basic job requirements to the satisfaction of the University, <u>they</u> shall be terminated and shall receive severance pay as set out in Article 12.09. The University may deduct all costs of retraining, including the employee's salary while retraining, from the amount of severance pay.

# (k) Priority Rights for Permanent Vacancies

An employee who accepts a temporary vacancy under this option shall continue to have priority rights for any permanent vacancy within the nine (9) month period following the expiry of the notice period.

# (I) Following Expiry of Notice Period

During the nine (9) month period following expiry of the notice period, a Regular employee who has not found a permanent job vacancy may request an advance on severance up to the amount <u>they</u> would be entitled to receive as severance pay. Upon successful placement into a regular position under this option, an employee as a condition of being offered such employment shall repay the aforementioned advance.

#### 12.09 PERMANENT LAYOFF OPTIONS - Severance Pay

#### (a) Severance Pay

Employees who have not made an election pursuant to 12.04(a)(ii) or who have elected severance pay, on termination of their services, shall receive severance pay based on their salary at the time of displacement, as follows:

Years of Service Amount of Severance Pay

Less than 2	4 weeks of salary
2 or more	6
3 or more	<u>8</u>
4 or more	2 weeks per year of service, to a maximum of
	24 weeks

## (b) Loss of Employment Status

An employee who elects the severance pay option shall lose their employment status.

#### (c) Monies Advanced to be Severance Pay

Any monies advanced under Article 12.08(l) to an employee who is not permanently placed under the Job Vacancy priority, shall be deemed to have elected payment of severance pay hereunder.

## 12.10 HEALTH BENEFITS MAINTENANCE DURING TEMPORARY LAYOFF

## (a) Regular Employee on Temporary Layoff to Pay Premiums

Regular employees on a temporary layoff who were receiving benefits specified under Article 28 may retain their previous coverage to the greatest extent possible by regularly paying to the University an amount equal to the total premiums involved. This provision only has application to layoffs exceeding thirty (30) calendar days duration.

# (b) University to Reimburse Employee for University's Share Upon Return to Work

On return to work in a position which attracts the same benefits, the University shall reimburse the Regular employee for the University's share of the premiums which were paid by the employee during their layoff.

## 12.11 TEMPORARY LAYOFF BECOMES PERMANENT

## (a) **Process**

In those circumstances where an employee on a Temporary Layoff (which by definition can be to a maximum of one hundred and twenty-three (123)

calendar days) receives notice of <u>their</u> Permanent Layoff, the following shall apply:

The University shall, on serving notice of <u>their</u> Permanent Layoff, advise the employee that one of the following options shall apply for the ninety (90) calendar day notice period:

- (i) <u>they</u> shall return to the position <u>they were</u> laid off from;
- (ii) <u>they</u> shall be assigned to any vacancy within <u>their</u> competence and paid not less than the salary equal to that <u>they were</u> paid in the position <u>they were</u> laid off from;
- (iii) <u>they</u> shall be placed on or remain on the Layoff List and shall be paid at the rate of the position <u>they were</u> laid off from.
- (iv) If at receipt of notice of Permanent Layoff an employee is on the Layoff List (per 12.06(k)), the advisement of and choice of option pursuant to 12.11(a) selected by the University shall become effective on the employee's receipt of notice.
- (v) If at <u>their</u> receipt of notice of Permanent Layoff an employee is in a bumped position (per 12.06), the employee may opt to have the choice of option pursuant to 12.11(a) selected by the University apply to <u>them</u> on receipt of it, or to delay its application by remaining in the bumped position for a maximum duration of one hundred and twenty-three (123) calendar days from the commencement of <u>their</u> Temporary Layoff or ninety (90) calendar days from <u>their</u> receipt of notice of Permanent Layoff, whichever occurs first.

# (b) Choices

- (i) An employee permanently laid off in this manner must select one of the options set out in 12.07, 12.08, 12.09 (i.e. bumping, job vacancy/retraining or severance pay) within ninety (90) calendar days of <u>their</u> receipt of notice of <u>their</u> Permanent Layoff.
- (ii) An employee selecting the bumping option may reaffirm the bump <u>they</u> selected on the Temporary Layoff, or <u>they</u> may choose to bump into another position.
- (iii) All rights arising out of this provision shall terminate on the layoff becoming permanent or at an earlier date if an employee has completed the process arising out of <u>their</u> selection.

# (c) Exception

The choices set out in 12.11(b) shall not have application in those circumstances where a Temporary Layoff continues beyond one hundred and

twenty-three (123) calendar days if such arises out of and continues to be a direct result of a strike, lockout or similar labour dispute. In such circumstances, all those rights arising out of a Temporary Layoff shall continue.

# (d) Reactivation of Former Position

If a position from which a Regular employee has been permanently laid off is reactivated within one (1) year of the commencement of the layoff, a right of first refusal shall be extended to the laid off employee provided <u>they</u> had selected option 12.04(a)(ii)(1) Bumping or option 12.04(a)(ii)(2) Job Vacancy/Retraining.

# 12.12 TYPE 2 LAYOFF (Bridge Period)

Predetermined time span of non work occurring between active segments of an employee's appointment. Bridge periods may occur in a continuous regular appointment, or in a temporary duration appointment. The Employer will advise the union of the bridge period start and end dates for each bridge employee, on an annual basis.

# (a) Rights

An employee on a Type 2 Layoff:

- (i) is entitled to a separation certification issued pursuant to E.I. regulations at the commencement of the layoff;
- (ii) shall be on the Previous Incumbency List of <u>their</u> classification/work area for the duration of <u>their</u> layoff, and shall have a priority right to any employment being offered within <u>their</u> classification/work area during this period. In circumstances where two or more employees on a bridge period seek to exercise the priority, seniority shall govern.

# 12.13 TYPE 3 LAYOFF (Normal Expiry of Temporary Appointment)

A predetermined cessation occurring at the end of a temporary hire having an expiry date designated at the commencement of an employee's appointment.

# (a) **Rights**

An employee on a Type 3 Layoff:

- (i) is entitled to a separation certificate issued pursuant to E.I. regulations at the end of <u>their</u> Temporary hire.
- (ii) who is a previous incumbent (i.e. held a greater than one (1) month hire), shall go on the Previous Incumbency List of <u>their</u> classification/work area.

(iii) In circumstances where two or more employees with previous incumbency in a position seek to exercise the priority access, seniority shall govern.

# 12.14 **PRIORITIES**

Notwithstanding any other listed priorities in this Collective Agreement, the following list of priorities shall govern:

# (a) **Priority I**

Regular employees on permanent type 1 layoff, or regular employees on layoff due to technological change, or employees returning from LTD (18.03) or regular employees on long term replacements (Article 18.04)

# (b) **Priority II**

Regular employees on temporary 1 layoff, or regular employees on type 2 layoff.

# (c) Priority III

Temporary employees on permanent type 1 layoff, or temporary employees on type 1 layoff, or temporary employees on layoff due to technological change, or temporary employees on long term replacements (Article 18.04).

# (d) Priority IV

Temporary employee on type 3 layoff.

# (e) Exception

A temporary employee on a Previous Incumbency List with more seniority shall have a priority over both Priority III and Priority II Regular employee on type 2 layoff (SEE ALSO 21.08 and 25.09).

# (f) Seniority

Within each category of priority, seniority shall govern.

# 12.15 INFORMATION TO BE GIVEN TO UNION

# (a) Layoff List

The University will provide a designated Councillor of the Union, or <u>their</u> designate, with a current copy of the Layoff List referred to in this Article. The Councillor, or designate, may use University facilities and equipment to copy this list and to mail it to the Union office. The Union will rotate the designation of Councillors, or designates, for this function such that no

one Councillor, or designate, performs this function in two consecutive terms.

# (b) Recall List

A copy of any recall notice issued by the University pursuant to any provision of this Agreement must be given promptly by the University to the Councillor, or designate, cited in (a) above. That Councillor, or designate, may use University facilities and equipment to copy this notice and to mail it to the Union office. Each such recall notice must include, amongst other information, the job title and work location for the position being filled by recall and the name and seniority date of the person being recalled.

## ARTICLE 13 - HOURS OF WORK

# 13.01 WORK DAY AND WORK WEEK

#### (a) Work Day

The work day of an employee is the hours which <u>they are</u> scheduled to work in a given day.

### (b) Work Week

The work week of an employee is the hours which <u>they are</u> scheduled to work in a given work week, commencing at 0001 hours (12:01 am) Monday.

# 13.02 STANDARD WORK DAY AND STANDARD WORK WEEK

#### (a) Standard Work Day

The standard work day shall be seven (7) hours exclusive of the meal period.

## (b) Standard Work Week

The standard work week shall be thirty-five (35) hours and is composed of five (5) consecutive standard work days.

## 13.03 MODIFIED WORK WEEK AND MODIFIED WORK DAY

#### (a) Modified Work Week

A modified work week is an approved modification of

- (i) the standard work week which must average thirty-five (35) hours per week over a maximum two (2) week period and include a minimum of four (4) work days per week, or
- (ii) the non standard work week which must average over a maximum two (2) week period those hours which would have been worked over the same period on a non standard work week, and include a minimum of three (3) work days per week.

## (b) Modified Work Day

A modified work day is a modification of a standard work day or a non standard work day resulting from an employee working an approved modified work week.

# (c) Modified Work Week

A modified work week may arise at the request of either the Administrative Head or the employee. Each modified work week shall meet the following criteria:

- (i) the daily work of the Department shall be carried out;
- (ii) where the Department relates to other components of the University, the role of the Department within the University shall not be diminished or diluted;
- (iii) where applicable the Labour Relations Board approves the scheme before it is implemented.

# (d) Administrative Head to Consider Employee's Request

The Administrative Head shall consider an employee's request for a modified work week, if it meets the following additional criteria:

- (i) No additional costs to the University shall result by reason of the proposed modified work week in the following ways:
  - by reason of the requirement for additional staff;
  - by reason of the attraction of costs additional to those already incurred under an existing approved work week (13.01(b));
  - by reason of more than seven (7) hours of lieu time when a paid holiday falls on a day of rest;
  - by reason of paying overtime rates within the approved hours.
- (ii) Where long weekends or a work week of less than five (5) days results, there will be an equal distribution of days off amongst employees who work a modified week in a Department.
- (iii) No request for a modified work week shall be refused without written justification.

## 13.04 EXCEPTIONAL WORK WEEK BASED ON ANNUAL HOURS

## (a) Exceptional Work Week

Notwithstanding any other provision herein, by mutual agreement between the University and an employee, and only after Agreement with the Union, may an exceptional work week schedule be established based on annual hours in a calendar year.

# (b) Restrictions

In each circumstance the annual hours must equal those hours which would otherwise have been worked within the year subject to the following restrictions:

- (i) each scheduled work week must fall between 85% and 115% of those hours from which the exception is based (e.g. if the normal weekly hours are 35, a permitted exceptional weekly range must fall between 30 and 40 hours);
- (ii) all other entitlements are derived from the exceptional new base (e.g. overtime from 30 or 40 as approved).

# (c) Approval

Each schedule shall be initially agreed upon and implemented for a trial period with a one year maximum duration. Subsequent approvals may have an indefinite duration.

# 13.05 APPROVAL OF HOURS OF WORK

## (a) Approval

- (i) The work day and the work week of an employee shall be approved by the appropriate Administrative Head, and shall become effective through written transmittal of the approval to the employee from the Human Resources Department.
- (ii) No regular employee who was employed as at October 1, 1989 shall have <u>their</u> hours of work changed by the University without <u>their</u> agreement. The University will not change the hours of work of any other regular employee unless the operational needs of the University require such change. The University undertakes to review alternatives to a change in a regular employee's hours of work prior to its implementation. An employee who feels <u>their</u> hours of work are being changed unnecessarily may grieve that change. In the event of a grievance, the hours of work will not be changed until the grievance is concluded.

# (b) Changes

- (i) When the Administrative Head changes an employee's work day or work week, the employee shall be notified and the change posted fourteen (14) calendar days prior to the effective date.
- (ii) When an employee wishes to change <u>their</u> work day or work week, the approval of the appropriate Administrative Head must be obtained and posted fourteen (14) calendar days prior to the effective date.

- (iii) In either <u>13.05(b)(i) or 13.05(b)(ii)</u> the fourteen (14) day time limit may be shortened by mutual agreement between the employee and the appropriate Administrative Head.
- (iv) The time limits herein set out do not apply to emergency type situations.

# 13.06 MEAL BREAK

An employee shall be entitled to take an uninterrupted meal break, of a minimum duration of thirty (30) minutes, away from <u>their</u> work area. When an employee is officially requested by <u>their</u> supervisor to work during <u>their</u> meal break, whether in <u>their</u> own work area or in any other area, the employee shall be compensated at the applicable overtime rates.

## **13.07 REST PERIODS**

# (a) Two Fifteen Minute Rest Periods

An employee whose daily hours of work equal or are greater than seven (7) may take two fifteen (15) minute rest periods within such.

## (b) One Fifteen Minute Rest Period

An employee whose daily hours of work equal or are greater than four (4) but less than seven (7) hours may take one fifteen (15) minute rest period within such.

## (c) When to Take Rest Periods

Normally, a rest period cannot be taken within one hour of the commencement or end of <u>their</u> daily hours, lunch break or another rest period.

## **13.08 TIME REPORT**

## (a) Semi-Monthly Paid Employees

Employees must submit their time sheets monthly.

## (b) Hourly Paid Employees

Employees paid on an hourly basis must submit an <u>a time report</u> immediately following the period covered by the report.

# (c) Approval

Time reports shall be submitted for approval to the appropriate administrator identified pursuant to Appendix N.

# 13.09 ALLOCATION OF ADDITIONAL HOURS OF WORK TO REGULAR PART TIME EMPLOYEES

The University will make reasonable effort to ensure that additional hours of work are shared equitably amongst those regular part-time employees in an area who have the required knowledge, skills and ability to perform the work.

# **13.10 MINIMUM DAILY PAY**

- (a) Employees reporting for work at the call of the University shall be entitled to their regular wage for the entire period spent at the place of work, with a minimum of four (4) hours pay for each call out subject to (b) below.
- (b) Temporary Part-Time employees in the Children's Centre, <u>Centre for</u> <u>Sport & Wellness</u>, Theatre, or Events and Ceremonies reporting for work at the call of the University shall be entitled to their regular wage for the entire period spent at the place of work, with a minimum of two (2) hours pay for each call out.

# 13.11 STANDBY DUTY

# Note: This article applies only to employees working in the Facilities Department, the Information Technology Services Department and the <u>Centre for Sport & Wellness</u>

## (a) Scheduled Standby Duty

An employee scheduled on standby will be paid two (2) hours at straight time for the 24-hour period commencing daily at 8:00 am, Monday to Friday, inclusive; and four (4) hours at straight time for the 24-hour period commencing at 8:00 am on a Saturday, Sunday or paid holiday.

## (b) Advance Notice

Where possible, standby will be signed up on a voluntary basis with schedules posted at least 96 hours in advance. Should an employee be given less than 96 hours notice of standby duty, <u>they</u> will not be required to accept such duty.

# (c) Consecutive Weekends

No employee will be compelled to accept standby on two (2) consecutive weekends or on two (2) consecutive holiday weekends.

# 13.12 EXCEPTIONAL WORK DAY

An employee hired to work in the Theatre during a production or performance may agree to work an exceptional work day of longer than seven (7) hours per day. The employee shall be paid at the overtime rate for all hours in excess of ten (10) hours per day or thirty-five (35) hours per week.

# **13.13** EMPLOYEE'S REQUEST FOR REDUCTION OF HOURS

The University will not grant an employee's request for reduced hours of work per week without the consent of the Union. Such consent will not be unreasonably withheld.

# 13.14 EMPLOYER'S REDUCTION OF HOURS

In the event of lack of work and the University reduces the hours of work per week of a regular position, the provisions of Article 12 (Layoff) shall apply.

# **ARTICLE 14 - HOURLY PREMIUMS**

## 14.01 SHIFT PREMIUM

An hourly premium of \$.75 shall be paid:

## (a) 6:30 pm - 6:30 am

for each hour worked between 6:30 pm and 6:30 am, or

## (b) **3:00 pm - 12:01 am**

for all hours of a standard or modified work day (i.e. full time) where the shift commences between 3:00 p.m. and 12:01 a.m.

# 14.02 SHORT CHANGE-OVER

If shifts are scheduled so that there are not twenty (20) hours between the start of an employee's shift and the start of an employee's next shift, overtime rates shall apply to hours worked on the succeeding shift within the twenty (20) hour period. If the shift is changed at the employee's request, overtime rates shall not apply.

# ARTICLE 15 - OVERTIME

# 15.01 **DEFINITIONS**

### (a) Standard or Non Standard Work Week Hours

For employees working a standard or non standard work week, overtime means any working hours on duty in excess of seven (7) hours in a work day or thirty-five (35) hours in a work week.

## (b) Modified Work Week Hours

For employees working a modified work week, overtime refers to those hours on duty in excess of the approved hours in <u>their</u> work day or work week.

## (c) Straight Time

Straight time rate means the regular rate of remuneration.

## (d) Double Time

Double time means two times the straight time rate.

## (e) Compensating Time Off

Compensating time off means the product of overtime hours worked times the applicable overtime factor.

#### **15.02 RIGHT TO REFUSE OVERTIME**

All overtime shall be voluntary. Employees may refuse overtime individually without being subject to disciplinary action, but there shall be no concerted refusals of overtime.

# 15.03 OVERTIME COMPENSATION

## (a) Non-Standard Work Day

An employee working a non-standard work day (of less than seven (7) hours) or work week (of less than thirty-five (35) hours) shall be paid at the straight time rate for all hours in excess of <u>their</u> approved work day or work week which do not qualify for overtime payment.

#### (b) **Double Time**

The rate of double time shall be paid for all hours of overtime.

# (c) Modified Work Week

An employee working a modified work week shall be compensated at double time for all hours worked on any day other than <u>their</u> regularly scheduled day of work.

# (d) Work on Other Than a Regularly Scheduled Day

An employee who works on other than a regularly scheduled day of work after a standard five day work week shall be compensated at the rate of double time for all hours worked on that day.

# (e) Paid Holiday

An employee who works on a paid holiday as defined in Article 16.01 shall receive either <u>their</u> regular days pay or another day off, and shall receive additional compensation at the rate of double time for all hours worked on that day.

# (f) Compensating Time Off

An employee shall have the option of receiving equivalent compensating time off or of being paid for all overtime earned at any time prior to 15.03(g).

## (g) Overtime Bank

All overtime remaining in an employee's overtime bank at the end of the calendar year following that in which it was earned shall be paid out by the University.

## (h) Leaves of Absence

The provisions of Article 15.03 shall not apply to any leaves of absence with pay.

# 15.04 CALL BACK

## (a) Additional Overtime

If a request that an employee work additional overtime is made when the employee is at work, the additional time worked outside of the employee's regular scheduled hours of work shall be considered as overtime. If the request is made at a time when the employee is not at work, the time worked shall be considered as a Call Back.

## (b) Call Back to Work

An employee called back to work shall receive a minimum of four hours at overtime rates.

# (c) Temporary Employees

A temporary status employee hired on an "as required" basis shall not be considered called back to work unless <u>they have</u> already worked on the same day.

# (d) Transportation Difficulties

In circumstances where an employee called back to work encounters transportation difficulties, the University shall, at <u>their</u> request, reimburse <u>them</u> for taxi transportation covering the return trip to <u>their</u> place of residence.

# **15.05 OVERTIME BREAKS**

All employees who are officially requested to remain at work for a period of three (3) hours or more beyond their approved work day, shall receive a one-half hour paid break at the applicable overtime rate. The break may be taken before, during, or after the overtime period. An employee may also choose not to take a break, in which case any break due to the employee which is not taken shall be compensated for at the applicable overtime rate.

## 15.06 SHARING OF OVERTIME

There shall be no discrimination within individual departments in the allocation or sharing of overtime.

## **15.07 OVERTIME MINIMUM**

Officially requested overtime shall be paid for in one-quarter hour segments with a daily minimum payable of one hour for the first hour or portion of an hour worked.

## **15.08 OVERTIME INFORMATION**

The University shall forward to the Union a list of employees' names and identification numbers, with the monthly and year-to-date overtime hours worked, by February 15th of each calendar year.

# **15.09 TRAVEL**

## (a) International Travel

International travel is travel outside of Canada. Except for those employees where international travel is a requirement of the job, international travel is voluntary.

Employees on international travel will continue to receive their regular pay for the duration of the trip. Where travel is required on either Saturday or Sunday, a regular day in lieu will be given to the employee. Lieu days will be scheduled at a mutually agreeable time between the employee and <u>their</u> manager and will be taken within thirty (30) days of the return from the trip.

# (b) Domestic Travel

Domestic travel is travel within Canada. Except for those employees where domestic travel is a requirement of the job, domestic travel is voluntary.

Employees on domestic travel will continue to receive their regular pay for the duration of the trip. Where travel is required on either Saturday or Sunday, a regular day in lieu will be given to the employee. Lieu days will be scheduled at a mutually agreeable time between the employee and <u>their</u> manager and will be taken within thirty (30) days of the return from the trip.

# (c) Travel Insurance

Employees booking travel using a University credit card are covered by travel insurance benefits provided by the credit card carrier. Employees are encouraged to book travel using a University credit card in order to receive this coverage.

# **ARTICLE 16 - HOLIDAYS**

# 16.01 PAID HOLIDAYS

#### (a) Standard Paid Holidays

The University shall grant as paid holidays:

New Year's DayVictoria DayGood FridayCanada DayEaster MondayBC DayLabour DayBoxing DayFamily Day (effective February 1, 2013)

Thanksgiving Day Remembrance Day Christmas Day

#### (b) Other Paid Holidays

Any other day declared a holiday by the Federal, Provincial or Municipal Government in which municipality the employee regularly works and wherein the subject municipality has granted the same paid holiday to their own employees, except for days declared which have already been granted on another day pursuant to this Article.

#### (c) Additional Paid Holidays

In addition to the paid holidays specified herein, there shall be one (1) additional day of holiday granted by the University in each calendar year. The University reserves the right to schedule this holiday and written notice of the specific date will be given to the Union two (2) calendar weeks in advance. All employees must take the holiday on the date so specified.

## 16.02 HOLIDAYS FALLING ON A SATURDAY OR SUNDAY

#### (a) Saturday or Sunday

When any paid holiday (as per section 16.01) falls on a Saturday, the following Monday shall be observed. When any paid holiday falls on a Sunday, the following Monday (or Tuesday, where the preceding section already applies to the Monday) shall be observed.

## (b) Other Than Saturday or Sunday

Subject to subsection 16.02(c), when a paid holiday falls on an employee's day of rest other than on a Saturday or Sunday, the employee shall be given a day off with pay in lieu at a mutually agreeable time.

# (c) Modified Work Week

If the day of rest on which the paid holiday falls is such by reason of an employee working a modified work week, the employee shall be given seven hours off with pay in lieu at a mutually agreeable time.

# (d) Facility Closed

If the paid holiday falls on a Saturday or Sunday and the following Monday (or Tuesday) is observed as per 16.02(a) and the University facility is closed, an employee who is regularly scheduled to work on the Saturday or Sunday may instead take a day off without pay or as a charge to earned time off or vacation time.

# 16.03 EMPLOYEES REQUIRED TO WORK ON A PAID HOLIDAY

An employee who works on a paid holiday shall be compensated as per Article 15.03(e).

# 16.04 HOLIDAY COINCIDING WITH A DAY OF VACATION

When a day of paid holiday falls during an employees' vacation time, the paid holiday shall not count as a day of vacation, nor as a day worked.

# 16.05 PAYMENT FOR HOLIDAYS

## (a) **Regular Employees**

Regular employees shall not have their pay reduced by virtue of holidays specified in Article 16.01.

## (b) Other Employees

Temporary employees must have been employed for at least 30 calendar days in order to qualify for statutory holiday payment.

- (i) An eligible employee who has worked on at least 15 of the 30 days prior to a statutory holiday shall be entitled to an average day's pay for the holiday. To calculate an average day's pay, divide the total wages earned in the 30 day period (excluding overtime) by the number of days worked.
- (ii) An eligible employee who has worked fewer than 15 of the 30 days prior to a statutory holiday shall be entitled to pro-rated statutory holiday pay. Pro-rated statutory holiday pay is calculated by dividing the total wages earned in the 30 day period (excluding overtime) by 15.

# (c) Modified Work Week

A paid holiday is equivalent to a maximum of seven (7) hours. Therefore, employees working modified work weeks shall be required to make up any time difference between the holiday and their modified work day when the latter is in excess of seven hours. Such time must be made up in one of the following time report periods:

- that preceding the holiday;
- that the holiday falls in;
- that following the holiday.

# (d) Lieu Days

Payment for unused lieu days shall be made prior to an RPT employee's bridge period.

# (e) Paid Holiday and Lieu Day

If a paid holiday falls on a day which a regular part time employee would not otherwise have worked, <u>they</u> shall be entitled to a lieu duration equal to the average of <u>their</u> scheduled weekly hours (e.g., weekly hours divided by the number of scheduled working days).

# 16.06 OTHER DAYS

# (a) Not Paid Holidays

The following days or times therein are not paid holidays:

- (i) afternoon on the last working day prior to Christmas;
- (ii) afternoon on the last working day prior to New Year's Day;
- (iii) Easter Sunday.

# (b) Facility Closed

Subject to operational requirements, the University will be closed at times listed in 16.06(a).

# (c) Payment

An employee shall be paid for the times listed in 16.06(a) that which they would otherwise have received had it not been for this provision.

# (d) Request to Work

If an employee is officially requested to work during the times listed in 16.06(a), <u>they</u> shall be paid as if it were work on a paid holiday, in accordance with Article 15.03(e).

# ARTICLE 17 - ANNUAL VACATION

# 17.01 CALENDAR YEAR

For the purpose of computing vacation entitlement "calendar year" means the twelve (12) month period from January 1st to December 31st, inclusive.

## **17.02** VACATION ENTITLEMENT

#### (a) Appointments Less Than Four Months

Employees with an appointment duration of less than four (4) months shall earn and be paid vacation pay on the basis of 6% of gross earnings.

## (b) Vacation Entitlement

The basic vacation entitlement (expressed for Full Time status) is:

	Days	Days
	per month:	per annum:
Initial incomplete & first		
four complete calendar years	1.25	15
5th through 8th	1.667	20
9th	2.083	25
10th	2.167	26
11th	2.25	27
12th	2.333	28
13th	2.417	29
14th & subsequent	2.5	30

## (c) Incomplete Calendar Year

For the purposes of 17.02(b), an incomplete calendar year which commenced on July 1st or earlier shall be considered a complete calendar year.

## (d) Temporary Employees

Temporary employees shall be paid their vacation entitlement at an equivalent percentage based on gross earnings.

## (e) Regular Employees

Regular employees shall accumulate time and be granted such pursuant to Article 17.04.

# 17.03 MODIFIED WORK WEEK

Employees working on modified work weeks shall have their vacation entitlement converted to hours - one day's entitlement being equal to seven (7) hours.

## 17.04 GRANTING OF VACATION LEAVE

The employees in a department should settle a vacation schedule each year subject to the approval of the Administrative Head. <u>Approval will be based on legitimate operational requirements.</u>

## 17.05 SPLIT VACATIONS

Where an employee wishes to split <u>their</u> vacation, <u>their</u> second choice of vacation time shall be made only after all other employees concerned have made their initial selection.

## 17.06 VACATION SUBSTITUTION

Where vacation substitution is required, the University shall attempt to give existing employees the opportunity to substitute in higher paying positions providing the employees have the necessary skills. The University will make reasonable effort to ensure the equitable distribution of substitution amongst employees.

## 17.07 OVERLAP OF VACATION WITH OTHER LEAVES OF ABSENCE

When an employee is entitled to sick leave, family leave (bereavement, etc.), or any other approved leave of absence with pay during <u>their</u> vacation period, the applicable leave of absence shall prevail and there shall be no deduction from the employee's vacation entitlements. Instead, the period of vacation which is thus displaced by the applicable leave of absence shall be taken later at a time mutually acceptable to the employee concerned and the University. An employee claiming an entitlement to sick leave during <u>their</u> vacation must substantiate this claim by providing the University with a note from a doctor of the employee's choice stating that the employee was ill or injured such that <u>they</u> would have been unable to attend work during the period of claimed sick leave entitlement and this note must have been obtained by the employee either during or in reasonable proximity to the conclusion of the illness or injury.

## 17.08 VACATION CARRY-OVER

#### (a) Annual Vacation

Employees, other than Bridge employees, may carry over up to ten (10) days vacation leave per year, however are required to take a minimum of two weeks annual vacation each year.

## (b) Vacation Bank

Employees may bank any vacation up to a maximum of twenty (20) days.

#### (c) Excess Vacation Bank

Employees who have banked the maximum allowable vacation are required to take the excess prior to the end of each calendar year. The excess may be

paid out by mutual agreement between the employee and the Employer. Should an employee accumulate a vacation bank in excess of the maximum allowable as a result of being unable to utilize their vacation due to operational requirements or other circumstances beyond the employee's control, the employee and their manager shall discuss options to address the excess with Human Resources. The proposed resolution will be reviewed and mutually agreed upon by the Union

# (d) Bridge Period Employees

Employees with a bridge period shall have the option of receiving their full earned vacation entitlement as a cash payout at their current rate of pay. Employees who exercise this option shall receive vacation payout two weeks prior to the commencement of their bridge period.

# 17.09 PRO-RATA ENTITLEMENT

Vacation entitlement, granting, accumulation, and carry over shall for all employees working less than thirty-five (35) hours per week or less than twelve (12) months per year, be based on a pro-rata amount relative to the full time entitlements herein listed.

# 17.10 NO CALL BACK FROM VACATION OR BRIDGE PERIOD WITHOUT EMPLOYEE'S CONSENT

- (a) Once an employee has commenced a scheduled vacation, such employee shall not be called back to work by the University, without the consent of the employee.
- (b) If an employee agrees to a call back to work by the University after the employee has commenced a scheduled vacation, the University shall reimburse the employee for any direct costs incurred by the employee as a result of any such call back, and the employee's remaining vacation shall be rescheduled to a time mutually acceptable to the University and the employee.
- (c) The provisions of this Clause 17.10 shall apply equally with respect to an employee's bridge period.

# 17.11 NO CHANGE IN SCHEDULED VACATION OR BRIDGE PERIOD WITHOUT EMPLOYEE'S CONSENT

- (a) An employees' vacation schedule once approved, in accordance with this Agreement (Article 17.04), shall not be changed by the University without the consent of the employee. In the event that an employee changes position, job or work location, the employee's scheduled vacation will be subject to approval by the appropriate Administrative Head.
- (b) If an employee agrees to any change in <u>their</u> previously scheduled vacation, the University shall reimburse the employee for any direct costs

incurred by the employee as a result of any such change, and the employee's changed vacation period(s) shall be rescheduled to a time mutually acceptable to the University and the employee.

(c) Once the employee's bridge period has been determined for the year, the provisions of this Article 17.11 shall apply equally with respect to an employee's bridge period.

# 17.12 IMPACT OF TEMPORARY PROMOTION

If an employee's rate of pay at the time of <u>their</u> scheduled vacation is based upon the employee temporarily performing work in a job with a higher rate of pay than the employee's regular job and the University postpones the employee's vacation, the employee shall be entitled to that higher rate of pay for the purposes of establishing vacation pay, if at the time of the employee's rescheduled vacation <u>their</u> then current rate of pay is lower.

# 17.13 NO CESSATION OF OPERATIONS TO USE OR EXHAUST VACATION

The University shall not implement any cessation of operations, in whole or in part, such that the vacation entitlement of any employee is used or exhausted except with the prior express written consent of the employee(s) concerned and the Union.

## **ARTICLE 18 - SICK LEAVE**

#### **18.01** SICK LEAVE ENTITLEMENT

#### (a) Sick Leave With Pay

<u>Regular</u> employees shall be granted sick leave with pay for a maximum of thirty (30) continuous calendar days. <u>Temporary employees whose length of appointment is less than 4 months shall have their sick leave entitlement prorated.</u>

#### (b) **Proof of Illness**

The University may require proof of illness. In such case, where the University, acting in good faith, challenges the validity of the proof of illness submitted by an employee, the University shall have the right to request that the employee secure another medical opinion from a medical doctor selected by mutual agreement between the University and the Union. The University shall bear all costs to the obtaining of any medical proof of illness pursuant to this Clause 1(b).

## (c) **Pregnant Employee**

A pregnant employee, who is not already on parental leave, shall have access to sick leave for periods of incapacity caused by <u>their</u> pregnancy. The University may require proof of such incapacity. As in above, all costs pertaining to the proof of illness shall be borne by the University.

#### **18.02** SICK LEAVE AND SHORT TERM DISABILITY

Should a single illness exceed thirty (30) calendar days in duration, the employee will suffer no loss of pay for the first thirty (30) day period as recognized in Article 18.01. After the thirty (30) days of a single continuous illness, the short term disability shall apply. Employees on short term disability will receive 66.7% of the employee's regular monthly salary up to \$2,500 per month plus 50% of monthly salary above \$2,500 for a period up to eleven (11) months. The University will pay the premium costs of the short term disability plan. After twelve (12) months of a single continuous illness, benefits of the Long Term Disability Plan in Article 28 of this Agreement shall apply, in accordance with the terms of that Plan.

## **18.03** JOINT EARLY INTERVENTION PROGRAM (JEIP)

The parties have agreed to participate in the Post-Secondary Joint Early Intervention Program (JEIP). The parties also agreed that the JEIP will incorporate the following principles:

**Jointly Managed** – The program will be jointly managed by the Employer and the Union.

**Mandatory** - An employee may be referred for participation in the JEIP when absent from work for five or more consecutive days or where it appears that there is a pattern of consistent or frequent absence from work. If an employee is referred, the employee must participate in the JEIP.

**Rehabilitative** – The JEIP is rehabilitative in nature.

**Confidential** – The parties involved in the program will maintain confidentiality of all information.

## **18.04 RETURN TO WORK FROM LTD**

- (a) Employees who are on sick leave, short term disability or long term disability will have the right to return to their position for a period of up to three (3) years from onset of disability, unless there is a definite, physician-supported prognosis of a return to work within 90 calendar days following the date of expiry of the three (3) year period, in which case the three (3) year period will be extended to accommodate the return to work of the employee.
- (b) If the employee is determined to be medically fit to return to work at a later date, the employee has the rights set out in Article 12.04 (ii)(2) and (3). The University and the Union will make every reasonable effort to accommodate the employee in seeking a new position within the University.
- (c) In the event that a placement is not available within 30 days from the return to work date, or the employee does not wish to accept the placement(s) offered, the employee will be placed on the Layoff List for a period of twelve (12) months, during which the employee will have Priority 1 rights to any vacancies that may arise. In the event that an employee is not placed in a regular position within the 12 month period, they would be entitled to severance pay as outlined in 12.09.
- (d) If the employee is able to return to work within the time frame set out in (a) above, where the position the employee occupied is eliminated, the employee shall exercise their rights outlined in Article 12.04.
- (e) This Article does not apply to employees who are on LTD as of September 1, 1998. Those employees retain all the rights set out in the Collective Agreement in force on March 31, 1998.

# **18.05 LONG TERM REPLACEMENTS**

## (a) Regular Status Replacement

Where it is necessary to hire a Regular status replacement in a position normally held by another employee who is either on Long Term Disability or who has been sick in excess of thirty (30) continuous calendar days, and where the sick employee returns to <u>their</u> position within the time frame specified in Article 18.03(a), the following shall apply to the replacement employee:

- (i) <u>they</u> shall be given three (3) months notice of the termination of <u>their</u> employment;
- (ii) on termination, <u>they</u> shall have a priority right to the first vacant position or hire for which <u>they have</u> the necessary qualifications and abilities. If the placement herein arising is not permanent, this right shall continue to exist until permanency is achieved. It is expressly understood this provision overrides Article 25, and that an employee shall be paid at the rate of the position.

## 18.06 REDUCED HOURS, PHASED-IN, RETURN TO WORK

#### (a) Return to Work

Employees returning to work following leaves of absence due to a Workers' Compensation claim or a short- or long-term disability, and whose personal medical doctor(s) recommend(s) that for a period of time they work less daily and/or weekly hours of work than they did prior to such absence, may return to work on a reduced hours, phased-in, basis, subject to the mutual agreement in each case between the University and the Union.

#### (b) Return to Position

An employee returning to work on a reduced hours, phased-in, basis pursuant to Clause 18.06(a) above shall normally return to the job and work location <u>they</u> held immediately prior to the applicable absence, however; the employee may be placed in another position within the bargaining unit and by mutual agreement between the University and the Union.

#### (c) **Rights, Obligations and Entitlements**

An employee returning to work on a reduced hours, phased-in, basis pursuant to Clause(s) 18.0<u>6</u>(a) and/or (b) above shall continue to be kept "whole" under the Collective Agreement with respect to all the rights, obligations, entitlements and benefits <u>they</u> enjoyed immediately prior to the applicable leave of absence (including any changes thereto which come into force and effect during the employee's absence), save and except that the employee shall have reduced hours of work and a salary which is reduced accordingly. Benefits which are derivative of salary (e.g. sick leave and vacation), will be prorated in accordance with the reduced salary rate. Without limiting the generality of the foregoing, it is understood and agreed that such employee shall continue to have the same entitlement to all benefit plans coverage and benefits under the Collective Agreement which applied to the employee immediately prior to the applicable leave of absence, including any changes thereto which come into force and effect during the employee shall continue to have the same entitlement to all benefit plans coverage and benefits under the Collective Agreement which applied to the employee immediately prior to the applicable leave of absence, including any changes thereto which come into force and effect during the employee's absence.

# (d) Payments from Workers' Compensation

An employee returning to work on a reduced hours, phased-in, basis pursuant to Clause 18.06 shall continue to be entitled to applicable payments from Workers' Compensation or short or long-term disability. Where the external funding body continues to pay the full disability benefit during the phased-in return, the University will have no obligation to supplement this payment.

## ARTICLE 19 - CAREER DEVELOPMENT

## 19.01 PURPOSE

The welfare of the University, the job related competence of employees, and the opportunity to prepare for more responsible jobs that now exist or will exist in the foreseeable future are among the prime considerations in exercising the provisions of this Article.

The provisions of this Article are intended to encourage and assist employees in maintaining and improving skills or to assist in preparing them for foreseeable jobs within the University.

## (a) Dispersal of Funds

The Staff Career Development Committee will make recommendations to the University on the dispersal of staff career development funds, made available by the University in the amount of not less than ten thousand (\$10,000) dollars annually. The Committee will be comprised of five employees appointed by the Union. The Chairperson of this Committee will receive 175 hours of release time in each calendar year, funding for such shall be over and above the funds allocated to staff career development.

## (b) Carry Forward

In the event that the sum allocated for the aforementioned career development in any year is not fully utilized in the intended year, the unutilized amount shall carry forward as funding available for Staff Career Development in subsequent years, in addition to that amount referenced in 19.01(a).

## **19.02 TRAINING LEAVE - EMPLOYER REQUIRED**

When the University requires an employee to take refresher courses or to further <u>their</u> job related training, the University shall grant a leave of absence with pay to the employee. Further, the University shall bear the full cost of the courses or training, including tuition, entrance or registration fees, laboratory fees and required books. The University shall also reimburse the employee for such travelling, subsistence and other related expenses as are previously approved by the University. The employee shall not be required to make up any time missed from work because of <u>their</u> participation in the required courses or training.

## **19.03 TRAINING LEAVE - EMPLOYEE REQUESTED**

#### (a) Request for Career Development

When an employee requests a leave of absence with pay for career development, the following shall apply:

(i) Requests for training leave must be made to the Manager of Human Resources and must be accompanied by an outline of the purpose of the leave, a statement of the training activity to be engaged in by the employee while on leave, and an assessment of the request against the purpose of Article 19.01.

- (ii) Requests for training leave may be submitted at any time. However, requests for leave of greater than four (4) months duration will be considered once a year and the deadline for this is April 1 of each year. By the following May 30th, the University shall advise employees as to the disposition of their leave request.
- (iii) Requests for leaves of four (4) months or less will be considered as they are received and every effort will be made to inform the employee as to the disposition of <u>their</u> request within thirty (30) days of its receipt.

# (b) Eligibility

Any employee may request training leave of four (4) months or less; only regular employees with a minimum of 9,100 hours of seniority may apply for training leave of greater than four (4) months. Normally, an employee shall not be eligible for a further leave in excess of four (4) months until 7,280 hours of seniority have occurred from the completion of the previous such leave.

# (c) Pay During Leave

Employees whose request for training leave is approved, will be paid at the rate of 60% of salary and benefits, pro-rata where applicable.

# (d) Accrual of Seniority During Leave

Seniority shall continue to accrue during all leaves granted pursuant to this Article.

## (e) Discretion of University

At the University's discretion, an employee's request for training leave may be granted pursuant to 19.02.

# (f) Prohibition

The presence of this article is not to prohibit requests for leaves of absence without pay for educational purposes being made under Article 20.06.

## **19.04 LIMITATION**

The University may impose reasonable limitations upon the number and nature of such training sessions that individual employees may participate in.

## **19.05 DEFERRED SALARY LEAVE**

The objective of a Deferred Salary Leave Program (DLSP) is to provide employees with an opportunity to self-finance a planned period of absence for any reason, such as continuation of educational studies, travel or other interests.

# (a) Applying for Deferred Salary Leave

An employee desiring such leave shall make a written application to the University at least two (2) months prior to entry to the Plan. The minimum Leave Period will be six (6) consecutive full calendar months and the maximum will not exceed twelve (12) consecutive full calendar months. The leave should commence immediately following the end of the Deferral Period.

# (b) Approval of Deferred Salary Leave

Approval of Deferred Salary Leave shall not be unreasonably withheld. In the event that such leave is not granted, the reasons shall be given to the employee in writing.

# (c) Funding for Deferred Salary Leave

Funding for the Leave of Absence shall be in accordance with Revenue Canada Regulations.

## (d) Period for Salary Deferral

The maximum period for salary deferral is six (6) years. During this period, the maximum percentage of salary deferred shall not exceed the following:

- one year 33 1/3%
- two years 33 1/3%
- three years 33 1/3%
- four years 25%
- five years 20%
- six years 16 2/3%

Once per year, the participants may upon one (1) month's written notice to the University, alter their percentage amounts for the next or any subsequent year.

# (e) University Obligation

The University shall remit to the financial institution those sums specified by the employee and governed by the Plan. All investments shall be deposited only in an institution covered by the Canada Deposit Insurance Corporation (CDIC). The University and the Union shall not be liable to any participant for investment made under this clause.

# (f) Administrative Expenses

Administrative expenses of the Plan will be paid out of the Plan itself.

# (g) Regulations

Revenue Canada Regulations regarding payment of deferred funds must be followed. The program will comply with Federal Income Tax Regulations.

# (h) Taking the Leave of Absence

The taking of the Leave of Absence shall be governed by the following provisions:

- (i) The participants in the Plan shall give the University a minimum of six (6) months' notice prior to taking of such leave.
- Participants must take the deferred leave after a maximum of six (6) years within the Plan or after the salary deductions have totalled 100%, whichever occurs first. Employees may, no later than six (6) months prior to this date, request a postponement of the commencement of their leave. This postponement shall not exceed more than twelve (12) months.
- (iii) On return from deferred leave, the employee will be returned to the same department and position as at the point of taking leave and with the applicable salary and benefits. Federal Income Tax Regulations require that employees return to work for a period equal to the duration of the leave; therefore, Deferred Salary Leave cannot serve as an early retirement benefit.

# (i) Health and Welfare Benefits

The Health and Welfare benefits will be as follows:

- (i) The participant shall bear the full cost of the Health and Welfare benefits.
- (ii) Vacation credits do not accumulate during the period of such leave.
- (iii) The University will make superannuation deductions required by the Pension (University) Act.

# **19.06 APPRENTICESHIP**

The University agrees that when it fills a position, in accordance with Article 25, which might become part of an apprenticeship program, the posting for the position will indicate that possibility. All terms and conditions of the Collective Agreement shall apply to the position and incumbent unless otherwise expressly agreed by the Parties.

## ARTICLE 20 - SPECIAL AND OTHER LEAVE

## **20.01 BEREAVEMENT LEAVE**

#### (a) Leave

In the event of critical illness or death in an employee's immediate family, the employee not on leave of absence without pay shall be entitled to paid leave of absence as follows:

- i) Up to and including seven (7) consecutive calendar days for critical illness;
- ii) Up to and including seven (7) consecutive calendar days for death; or
- iii) Up to and including fourteen (14) consecutive calendar days for combined critical illness and death.

#### (b) Timing

Leave granted for critical illness will terminate in the event of death of the immediate family member and the employee will then be entitled to bereavement leave. Leaves under this Article must be taken during or in relation to the time of the actual occurrence of the critical illness or death. Any additional leave shall be without pay, or as a charge to earned time off or vacation time.

#### (c) Definition of Immediate Family

Immediate family is defined as an employee's spouse, a parent, child, brother, sister, grandparent or grandchild of either the employee or the employee's spouse; or any other person permanently residing in the employee's household. Should the deceased have been an intimate, in the sense of a close personal relationship, such leave shall be requested in writing to the Manager of Human Resources. The word "spouse" shall be deemed to include a common law spouse and partners of the same sex.

## (d) Employee to Advise University

In the event the sudden bereavement prevents the application for leave to be made in writing, the employee shall make every effort to personally advise the University, or shall have the University advised, of the reasons for such emergency absence.

#### (e) Bereavement Leave During Vacation

If an employee is on vacation at the time of bereavement, the employee shall be granted bereavement leave and shall have the number of days of bereavement leave added to <u>their</u> vacation entitlement.

# 20.02 SPECIAL LEAVE WITH PAY

Any employee shall be entitled to special leave at <u>their</u> regular rate of pay for the following:

# (a) Citizenship

Attend their formal hearing to become a Canadian citizen - one day.

# (b) Mourning

Upon request to the Manager of Human Resources, reasonable leave to attend a funeral as a mourner.

# (c) Family Responsibility

Upon request to the Manager, Human Resources, an employee may take up to four (4) days per year to meet responsibilities related to the care, health or education of <u>their</u> family. Such time off work may be taken at one (1) time or in increments of one (1) or more hours to a total of twenty-eight (28) hours.

An employee on leave of absence without pay shall not be entitled to these leaves and an employee on vacation shall not be entitled to (b) Mourning or (c) Family Responsibility leaves.

# 20.03 PUBLIC DUTIES

The University shall grant, on written request, leave of absence without pay for:

## (a) Seeking Election (Provincial or Federal)

The period of the Writ, for employees to seek election in a Provincial or Federal election.

# (b) Seeking Election (Municipal)

A maximum of fifteen (15) days for employees to seek election in a Municipal election.

# (c) Upon Election

If elected, the employee shall be granted leave of absence without pay for <u>their</u> term in office up to a maximum of six (6) years for election under 20.03(a) above, and up to a maximum of three (3) years for election under 20.03(b) above. Beyond the maximum term of leave herein, the employee shall be deemed to have terminated <u>their</u> employment with the University.

# (d) Maintenance of Seniority

An employee granted leave under this Article 20.03 shall maintain seniority in the bargaining unit, but shall have no right to displace employees upon return. However, such an employee shall within the limits of the period of time allowed by this article, have the right to apply for vacant positions within <u>their</u> qualifications. In the event of a vacancy in a position which appears to be within the qualifications, skills and abilities of an employee returning from public duty leave, such vacant position need not be posted until a decision with respect to the suitability of the employee is reached.

# 20.04 LEAVE FOR COURT APPEARANCES

# (a) Jury Selection/Jury Duty

The University shall grant leave with pay to employees who must attend jury selection or serve as jurors or Crown witnesses in a court action.

# (b) Court Appearance

In cases where an employee's private affairs (other than those specified in Section 20.02) have occasioned a court appearance, a leave of absence without pay shall be granted for a period of time reasonably required by such court appearance.

## (c) Pay

An employee in receipt of <u>their</u> regular earnings while serving at court shall remit to the University all monies paid to <u>them</u> by the Court, except travelling and meal allowances not reimbursed by the University.

## (d) Jailed Pending Appearance

In the event an employee is jailed pending a court appearance, such leave of absence shall be without pay.

## 20.05 ELECTIONS

Any employee eligible to vote in a Federal, Provincial or Municipal election or a referendum shall have four (4) consecutive hours during the hours in which the polls are open, in which to cast <u>their</u> ballot. The University shall designate the time of day in which the four (4) hours may be taken.

## 20.06 LEAVE OF ABSENCE WITHOUT PAY

#### (a) Requesting

Employees may request leave of absence without pay. Such request for leave shall be in writing to the Manager of Human Resources, who shall review the request and make every reasonable effort to obtain University approval. All approved leaves will be subject to the provisions outlined in Article 28.09 (b).

## (b) Emergency Absence

In the event of a personal emergency, such as sudden notification of a serious family illness, which prevents the application for leave to be made in writing, the employee shall make every effort to personally advise the University, or shall have the University advised, of the reasons for an emergency absence, which the employee wishes to have treated as such leave.

## 20.07 PERSONAL LEAVE WITH PAY

Employees may request Personal Leave with Pay in circumstances where existing entitlements as set out in this Agreement prove insufficient. Such requests for leave shall be in writing stating the reason for the request to the Manager of Human Resources.

## 20.08 COMMUNITY EMERGENCY

If an employee is called upon by the Municipality or the Provincial Government to aid the community served by the University in some disaster, the University agrees to retain the employee on the University payroll and agrees to reimburse the employee for the difference between any amount received for such service and <u>their</u> normal working day's pay. If an employee is on vacation leave or earned time off at the time of summons for such effort, the number of days so displaced shall be taken at a mutually agreeable time.

## 20.09 TERMINATION OF LEAVE

At the termination of a leave, an employee shall recommence performing the functions and duties <u>they</u> performed prior to the leave as evidenced by <u>their Offer</u> <u>Letter</u>.

## 20.10 DONATING BLOOD

An employee shall be granted leave of absence with pay for one (1) hour for the purpose of donating blood at a blood donor's clinic or health laboratory, subject to provision of documentary proof of such action.

# 20.11 EXAMINATION LEAVE

Leave of absence with pay shall be granted by the University to allow employees time off work to undertake examinations in respect of training or educational courses or programs required by the University or by mutual agreement between the Union and the University.

## 20.12 NO CALL BACK FROM LEAVE OF ABSENCE

- (a) Once an employee has commenced an approved leave of absence, such employee shall not be called back to work by the University without the consent of the employee.
- (b) Once an employee has commenced an approved leave of absence, such employee shall not return to work prior to the conclusion of the approved leave of absence without the consent of the University.
- (c) If an employee agrees to a call back to work by the University after the employee has commenced an approved leave of absence, the University shall reimburse the employee for any direct costs incurred by the employee as a result of such call back. The employee's remaining leave of absence shall, at the option of the employee, be rescheduled to a time mutually acceptable to the employee and the University.

## 20.13 GENDER TRANSITION LEAVE

An employee who provides a certificate from a medical practitioner confirming that the employee requires a leave of absence in order to undergo gender transition will be granted a leave for the procedure required during the transition period. The provisions of that leave will follow either an unpaid leave of absence or Article 18 – Sick Leave depending on the employee's request and approval by the provider.

The Union, the Employer and the employee will work together to tailor the general transition plan to the employee's particular needs.

# 20.14 DOMESTIC OR SEXUAL VIOLENCE LEAVE

An employee may require an absence from work to seek medical attention, counselling or other social or psychological services, or legal advice, or to seek new housing due to an employee and/or an employee's dependent child or a dependent person under their care experiencing domestic/sexual violence. In such a case, the employee shall be granted leave consistent with the applicable legislation and the Employer will provide up to three (3) days of leave with pay per calendar year.

Such leave may be taken intermittently or in one continuous period. The Employee, or the Union on the Employee's behalf, may request additional leave as provided elsewhere in this collective agreement including Article 20.07. Such request shall not be unreasonably denied.

The parties agree that if additional paid domestic or sexual violence leave is legislated into the Employment Standards Act in excess of what is provided under this article, the University will honor the amount of paid leave provided in the Employment Standards Act.

# ARTICLE 21 - PARENTAL LEAVE

# 21.01 APPLICATION

All provisions of this Article apply equally to <u>all genders</u>.

# 21.02 MATERNITY AND <u>PARENTAL</u> LEAVE

#### (a) **Prior to the Date of Birth**

- (i) An employee will be granted unpaid leave of absence for maternity or <u>parental</u> reasons relating to the birth or adoption of <u>their</u> child. Any absence granted under this policy will not exceed <u>seventy-eight</u> (78) continuous weeks, except as provided under Article 21.02(c). Thirty (30) days prior to commencement of the leave of absence, the employee will notify <u>their</u> Administrative Head of the number of weeks of leave <u>they</u> will be taking.
- Leave will begin at any time chosen by the employee during the six
   (6) week period immediately preceding the expected delivery date or date of adoption. Unless medical reasons make it impractical, employees qualifying for unpaid leave of absence shall provide notice as specified in Article 21.02(a)(i).

## (b) On the Date of the Birth or Adoption

An employee not exercising any other entitlement under this Article shall be granted a two (2) day leave with pay on, or immediately prior to, or after the date of birth or adoption of <u>their</u> child.

# (c) Extension to the Leave

On written request, an employee shall be granted an extension of up to one (1) year to the fifty-two (52) weeks of <u>parental</u> leave, provided the request is for medical reasons related to the infant and is supported by a medical certificate provided by the employee's physician. Any further extensions will be at the sole discretion of the University. There will be no annual vacation accrual during the extension period unless provided for elsewhere under the Collective Agreement.

# (d) Vacation Accrual While on Leave

Vacation entitlement shall continue to accrue while an employee is on leave for up to seventy-eight (78) weeks. Vacation earned pursuant to this clause may be carried over to the following year.

# 21.03 CESSATION OF EMPLOYMENT OR RETURN TO WORK

## (a) Cessation of <u>Maternity or</u> Parental Leave

Maternity or parental leave shall cease when the employee:

- (i) resigns during the period of leave;
- (ii) elects not to return to <u>their</u> position at expiry of the leave, or fails to do so within the prescribed time limit.

## (b) Return to Employment

Employees wishing to return to regular employment following maternity or <u>parental</u> leave shall notify their Administrative Head in writing at least thirty (30) calendar days prior to the desired date of return.

# 21.0<u>4</u> POST MATERNITY OR <u>PARENTAL</u> JOB POSTING RIGHTS

# (a) Job Postings/Competitions

The employee may apply while <u>they are</u> on maternity <u>or parental</u> leave. On temporary postings, the employee must be available to return to work within fourteen (14) days of notification of being the successful applicant in a job competition; otherwise the Administrative Head will consider <u>them</u> to have withdrawn from the competition.

If the employee is the successful applicant on a permanent posting, then <u>they</u> may return to work earlier or at the expiry of <u>their</u> maternity <u>or parental</u> leave.

## (b) Continuation of Status

An employee on maternity leave or <u>parental</u> leave who terminates by not returning to work, shall have the right to apply for job postings providing <u>they</u> maintain <u>their</u> status as a member in good standing of the Union.

## (c) Qualification

In order to qualify for the right to apply for job postings, the employee must advise the University of <u>their</u> resignation not later than twelve (12) weeks from the commencement of the leave of absence. The University may then proceed to fill the resultant job vacancy on a permanent basis.

## 21.0<u>5</u> BENEFIT COVERAGE

If an employee maintains coverage for medical, extended health, dental and group life insurance while on parental leave, the University agrees to continue to pay its share of premiums.

# 21.0<u>6</u> VACANCY CREATED BY GRANTING OF PARENTAL <u>OR</u> MATERNITY LEAVE

# (a) Filling a Vacancy

The University may fill the vacancy created by the granting of parental <u>or</u> maternity leave in accordance with Article 25. The employee selected to fill the vacancy shall continue to do so until the employee on parental <u>or</u> maternity leave has:

- (i) at the expiration of <u>their</u> leave, returned to <u>their</u> position;
- (ii) resigned, or advised the University of <u>their</u> election not to return to the position at the expiration of <u>their</u> parental leave, and the University has refilled the position.

# (b) Extension

Where an extension is granted under Article 21.02(b)(ii), the corresponding extension of an appointment pursuant to Article 21.06 need not be posted.

# 21.0<u>7</u> APPLICATION OF SICK LEAVE

Sick leave does not apply during parental leave.

# 21.08 TEMPORARY EMPLOYEES

A temporary employee who refuses an offer of employment because of maternity or <u>parental leave</u> reasons shall not lose <u>their</u> status as a previous incumbent for the position. If the temporary position is to be extended when the employee wishes to return to employment, <u>they</u> shall have first right to the appointment. (Also 12.14(e) and 25.09).

# 21.0<u>9</u> IMPACT OF LEGISLATION

The provisions for family related leave contained in this Article including, but not limited to, those concerning maternity <u>and parental</u> leave are intended to establish minimum standards. If any applicable legislation provides leave provisions which are more favourable to the employee, such legislation shall apply and prevail.

# 21.10 SUPPLEMENTAL EMPLOYMENT BENEFIT FOR MATERNITY AND PARENTAL LEAVE

- a) Effective the date of ratification, when on maternity or parental leave, a regular employee will receive a supplemental payment added to Employment Insurance benefits as follows:
  - i) The supplemental payment, during the waiting period not covered by the Employment Insurance Benefits, shall be seventy five percent (75%) of the employee's average base salary.
  - The allowance for the remainder of the twelve (12) month benefit period shall be an amount such that the combination of Employment Insurance benefits available for the standard maternity and parental leave and all other earnings shall not exceed seventy five percent (75%) of the employee's average base salary.

If the employee has opted for the extended parental EI Benefit, the aggregate amount of the entitlement may be divided into equal payments over the entire leave period, but shall not exceed the aggregate amount of the standard supplemental employment benefit.

- iii) The average base salary for the purpose of Clauses 21.10 (a)(i) and 21.10 (a)(ii) is the employee's average base salary for the twentysix (26) weeks preceding the maternity or parental leave. If the employee has been on unpaid leave for part of the preceding twenty-six (26) weeks, then up to four (4) weeks of that unpaid leave will be subtracted from the twenty-six (26) weeks for the purpose of calculating the average base salary.
- b) An employee is not entitled to receive Supplemental Employment Benefits and disability benefits concurrently.

- c) To receive Supplemental Employment Benefits, the employee shall provide the Employer with proof of application for and receipt of Employment Insurance benefits. An employee disentitled or disqualified from receiving Employment Insurance maternity or parental benefits is not eligible for the Supplemental Employment Benefits.
- d) To be entitled to the above noted benefits, an employee must sign an agreement that they will return to work and remain in the University's employ for a period of at least six months or equivalent to the leave taken, whichever is longer, after their return to work.
- e) Should the employee fail to return to work and remain in the employ of the University for the return to work period, the employee shall reimburse the University for the benefits above on a pro-rata basis.

# ARTICLE 22 - SAFETY AND HEALTH

# 22.01 CONDITIONS

The Union and the University agree that regulations made pursuant to the Workers' Compensation Act, or any other statute of the Province of British Columbia pertaining to the working environment, shall be fully complied with.

# 22.02 HEALTH AND SAFETY COMMITTEE

## (a) **Definition of Committee**

The Union and the University agree to maintain a Health and Safety Committee comprised of two (2) representatives from each party. The Faculty Association may be invited to appoint two (2) members. The Committee shall make recommendations to the President on unsafe, hazardous and dangerous conditions with the aim of preventing and reducing risk of occupational injury and illness. A copy of all minutes of the Health and Safety Committee shall be sent to the Union and the University. The University shall post the minutes online and on the Health and Safety bulletin boards.

# (b) Compensation

Any employee who serves on the Health and Safety Committee shall receive <u>their</u> regular straight time rate of pay for:

- (i) attending meetings of the Committee;
- (ii) investigating safety matters at the direction of the Committee

## (c) Unsafe Conditions

Any employee may make a written or verbal representation concerning unsafe working conditions to the Health and Safety Committee.

## 22.03 UNSAFE AREAS

No employee shall be required to work in an unsafe area.

## 22.04 RECOMMENDATIONS

The Union and the University agree that recommendations of the Health and Safety Committee where they are in accord with statutory requirements, shall be acted upon as soon as possible.

# 22.05 INDUSTRIAL FIRST AID REQUIREMENTS

The Union and the University agree that First Aid Regulations made pursuant to the Workers' Compensation Act shall be fully complied with. Where the University

requires that an employee obtain, renew or upgrade <u>their</u> Industrial First Aid Certificate, any fees, tuition or costs of course materials and books shall be borne by the University.

## 22.06 TRANSPORTATION FOR EMPLOYEES REQUIRING MEDICAL CARE

Transportation from their place of work to the nearest physician or hospital for employees requiring urgent medical attention shall be at the expense of the University, unless covered by the insurance plans.

# 22.07 SHUTDOWNS

If any work is temporarily shut down or modified as a result of:

# (a) Health and Safety

a complaint by an employee about health and safety;

# (b) Refusal to Work

a refusal to work in accordance with this Article or any applicable legislation;

# (c) Order of Inspector

an order of a government inspector;

every affected employee shall be kept "whole" by the University with respect to all pay, seniority, benefits and any other rights or entitlements which have accrued to the employee under this Agreement had there not been any shutdown or modification. Affected employees may be assigned temporarily to other jobs within the bargaining unit, as long as they possess the knowledge, skills and ability for the other job.

## 22.08 HEALTH AND SAFETY TRAINING

The University shall provide occupational health and safety training and education to every employee sufficient to enable each employee to work with maximum capability with respect to health and safety in <u>their</u> workplace at the University.

# ARTICLE 23 - WORK CLOTHING

# 23.01 SUPPLY OF WORK CLOTHING

When an employee is required by the University to wear specific clothing or safety footwear other than that found in <u>their</u> personal wardrobe, the University shall provide the designated clothing and safety footwear at no cost to the employee.

#### **23.02 PROTECTIVE CLOTHING**

The University shall provide protective clothing for any job which requires it at no cost to the employee.

## 23.03 MAINTENANCE OF WORK CLOTHING

It shall be the University's responsibility that clothing issued under sections 23.01 and 23.02 of this document, is maintained, cleaned and repaired and replaced as necessary, at no cost to the employees.

# ARTICLE 24 - TECHNOLOGICAL CHANGE

## 24.01 **DEFINITION**

"Technological Change" shall mean:

#### (a) Definition

the introduction by the University of a change in its work, undertaking, or business, or a change in its equipment or material from the equipment previously used by the University in its work, undertaking or business, or

## (b) Introduction of Equipment

a change in the manner, method or processes in which the University carries on its work, undertaking, or business related to the introduction of that equipment or material.

## 24.02 CONSEQUENCES

## (a) Introduction of Change

Where the University introduces or intends to introduce a technological change as defined in 24.01, which:

- (i) affects the terms and conditions or security of employment of one or more employees to whom the Collective Agreement applies, or
- (ii) alters significantly the basis upon which the Collective Agreement was negotiated,

the University shall provide the Union and affected employee(s), in writing, a minimum of thirty (30) days notice of such and at the time notice is provided, or as soon as possible thereafter, with the reasons in writing for the introduction of technological change and with as much information related to the technological change as possible.

# (b) Training

The University agrees that, prior to implementing technological change, it will make reasonable training in the new technology available to affected employees. Employees participating in such training outside of their normal working hours will be compensated for that time at their regular rate of remuneration, or at the discretion of their Administrator, will receive compensatory time off.

# (c) Failure to Complete Training

Where such training has been made available, failure to participate in, or successfully complete, the training will not prevent or delay the introduction of the technological change.

# (d) Election Not to be Retrained

Where an employee elects not to be retrained or does not successfully complete the retraining, the provisions of Article 24.06(a), 24.06(b), 24.06(c), 24.06(d) will apply.

# 24.03 DISCUSSION

Where the University has notified the Union of the intended technological change, the parties undertake to meet within the next thirty (30) days to discuss potential problems and attempt to reach agreement which will minimize the impact of such technological change on employees affected thereby.

# 24.04 RESULTING AGREEMENTS

Agreements reached during discussions between the parties shall be set out in a memorandum of agreement which shall have the same force and effect as the provisions of this Collective Agreement.

## 24.05 DISPUTE RESOLUTION

In the event that the introduction of technological change as defined in 24.01 creates a dispute between the University and the Union, either party may refer the matter to an arbitrator pursuant to Article 8 of this Agreement, bypassing all other steps in the grievance procedure.

The arbitrator shall decide whether or not the University has introduced or intends to introduce technological change and, upon deciding that the University has or intends to introduce a technological change, the arbitrator shall:

## (a) Findings

inform the Minister of Labour of their findings; and

## (b) Action

may then or later make any one or more of the following orders:

(i) that the change be made in accordance with the terms of the Collective Agreement unless the change alters significantly the basis upon which the Collective Agreement was negotiated;

- (ii) that the University will not proceed with the technological change for such period, not exceeding ninety (90) days, as the arbitrator considers appropriate;
- (iii) that the University reinstate any employee displaced by reason of technological change;
- (iv) that the University pay to the employee such compensation in respect to the employee's displacement as the arbitrator considers reasonable;
- (v) that the matter be referred to the Labour Relations Board.

# 24.06 NON DISPUTE

Where the Union does not dispute the introduction of technological change pursuant to Article 24.05, an employee who has received notice (per 24.02) must, prior to the expiry of ninety (90) days from receipt of that notice, elect one of the options available to them as listed below and according to Article 12.04:

# (a) Bumping

bumping,

## (b) Vacancy

a job vacancy or retraining for a job vacancy, or

(c) Pay

severance pay,

(d) Vacancy per 12.08

a job vacancy as per 12.08.

# **ARTICLE 25 - APPOINTMENTS**

## 25.01 **DEFINITION**

A job opening occurs when there is a vacant position or anticipated vacant position to which the University currently intends to appoint an individual or any other opening for appointment.

# 25.02 JOB POSTINGS

## (a) **Posting**

All job openings of one month or more not filled through the previous incumbency process shall be posted electronically for a period of not less than five (5) working days.

# (b) Contents of Posting

All job postings shall contain applicable class specification information and criteria including:

- job title
- nature of position
- required qualifications
- required knowledge and skills
- salary range
- work day and work week
- position number

# (c) Advertisements

Advertisements shall indicate that Capilano University is an equal opportunity employer and that Union membership in the Union shall be a condition of employment. If the bargaining unit status of the position has not been determined, the advertisement shall indicate that Capilano University is an equal opportunity employer, and that Union membership in the Union may be a condition of employment.

# (d) Union to Get Copy of Posting

The Union will be provided with a copy of new or reclassified job postings at least twenty-four (24) hours prior to the new or reclassified job being posted.

#### 25.03 APPOINTMENT PROCEDURES

#### (a) Durations of up to One (1) Calendar Month Maximum

While the method utilized by the University for staffing these needs shall be at its discretion, the University undertakes to refer, in a seniority sequence, former or current employees' names on the Recall List. It is clearly understood that insufficient lead time for an appointment of less than one (1) month will inhibit the University's ability to meet this undertaking. The University will advise the Union in writing of any renewals to one month hires. One month hires may be renewed twice, for a maximum duration of three (3) months, at which point, unless waived by agreement of the parties, the provisions of Article 25.03(c) will apply to any further hire.

# (b) Durations from One (1) Calendar Month to Three (3) Calendar Months Maximum

- (i) Where the class specifications and major work area of a job opening is one for which there has been a previous appointment, a previous incumbent shall have the first right to the appointment, except where there is an employee on a Layoff List pursuant to Article 12.06(k) in which case Article 25.03(d) shall govern.
- (ii) If no previous incumbent is available for the appointment, or if there was no previous incumbent, the job opening shall be posted. Employees on the Recall List with over one hundred (100) hours of seniority may indicate their preference for receiving notice of postings by electronic communication, telephone access or regular mail. An onus shall rest with those on the Recall List to advise Human Resources, in writing prior to a given closing date, of their desire to be considered for the job opening.
- (iii) After the closing date, Human Resources shall refer the names of those on the Recall List who have declared their interest and the names of employee candidates to the recruiting area in seniority fashion. The names shall be referred in two listings:

first, the names by seniority of those who have work experience consistent with the type of employment available;

second, the names by seniority of all other candidates; these candidates will have work experience which is not consistent with the employment available.

#### (c) Duration of Greater than Three (3) Months

(i) Where the class specification and major work area of a job opening is one for which there has been a previous appointment, a previous incumbent shall have the first right to the appointment unless the job opening is for a position created under Article 9.07, or where the job opening is of a temporary nature and where there is an employee on a Layoff List pursuant to Article 12.06(k) in which case Article 25.03(d) shall govern.

(ii) Where no previous incumbent is available for the appointment, or if there was no previous incumbent, the job opening shall be posted and the University may simultaneously advertise the job opening. A copy of the job posting shall be forwarded by mail to all names on the Recall list with over one hundred (100) hours of seniority. An onus shall rest with those on the Recall List to apply in writing if they wish to be considered for the appointment.

# (d) Precedence of Layoff List

- (i) In circumstances where any Layoff has been activated pursuant to article 12.06(k), the following procedure shall supersede those staffing processes set out in 25.03(a), 25.03(b), and 25.03(c) as such relate to Temporary hires.
- (ii) Pursuant to 25.03(d)(i), all employees on all Layoff Lists (as defined in Article 32) shall have priority rights to Temporary hires subject to Article 25.07.

# (e) Precedence of Permanently Laid Off Employees

Where an employee has been permanently laid off pursuant to Article 12.03(b), any right arising under that employee's election of the job vacancy/retraining option as set out in Article 12.08 shall be a priority right over all temporary and regular staffing procedures set out in Article 25.03(a), 25.03(b), 25.03(c) and 25.03(d).

# (f) Rights of Unsuccessful Candidates

On request, the University shall give an unsuccessful candidate for any job the reasons in writing why that employee's candidacy was not successful and the Union shall have the right to grieve the matter in accordance with clause 25.05 of this Agreement.

# 25.04 NOTIFICATION TO UNION

The Union shall receive an electronic copy of all <u>Offer Letters</u> for successful candidates and appointees.

# 25.05 RIGHT TO GRIEVE

There may be instances where the employee wishes to grieve the outcome of a selection. The process is outlined in 7.03.

# 25.06 INTERIM APPOINTMENTS

Any vacant position may be filled on an interim basis:

## (a) Grievance

where a grievance has been filed under Article 25.05;

## (b) Competition

while a competition to fill a job vacancy is in process.

## 25.07 APPOINTMENTS

#### (a) Required Knowledge, Skills and Abilities

Among those candidates who have the required knowledge, skills and abilities, the candidate with the most seniority shall be offered the appointment.

# (b) Seniority

Notwithstanding 25.07(a), candidates with less than 375 hours of seniority shall be treated as if they had no seniority for the purpose of appointments to Regular positions.

#### (c) Compensation

If a successful internal applicant to a job posting is not placed in the new job within four weeks of the appointment, the employee will then be paid the rate of the new position, if greater.

# (d) Job Duties

It is understood and agreed that any knowledge, skills and/or abilities which are established for any job must be related by the University, reasonably, fairly and consistently to the major job duties to be performed.

# 25.08 PROBATION

# (a) Current Employee

A current employee appointed pursuant to Article 25.07 shall be on a probationary period of three (3) month. For an employee working part time, the probationary period shall be the equivalent of three full months, i.e. 456 hours. The employee will be provided with an interim evaluation and a final review. If the employee finds the job unsatisfactory or is unable to meet the basic job requirements to the satisfaction of the University, the University shall either return them to their former position, or to one of an equal pay group.

## (b) Former Employee

A former employee with seniority appointed pursuant to Article 25.07 shall be on the same probationary period set out in 25.08(a). However, if <u>they are</u> unable to meet the basic job requirements to the satisfaction of the University, <u>their</u> employment status shall be terminated.

## (c) Incomplete Initial Probation Period

An individual appointed pursuant to Article 25.07, who has not completed an initial probation period of six months, shall serve a probationary period of six (6) months. If the employee is unable to meet the basic requirements to the satisfaction of the University, <u>their</u> employment status shall be terminated.

# 25.09 PREVIOUS INCUMBENCY

#### (a) Status

Arising out of the appointment procedures set out in 25.03, a previous incumbent shall not lose <u>their</u> status (of being a previous incumbent) by reason of <u>their</u> refusal to accept an employment offer of one month or less duration.

## (b) Loss of Status

However any refusal by a previous incumbent of an employment offer of greater than one month shall result in <u>their</u> loss of previous incumbency status for the particular classification/area, unless reasonable grounds are given by the employee for such refusal (SEE ALSO 12.14(e) and 21.0<u>8</u>).

# 25.10 EXTERNAL HIRES

If a vacancy cannot be filled under the preceding provisions of this Article, then the University shall have the right to hire from external sources, providing that the same knowledge, skills and abilities requirements are applied to external and internal candidates.

# **ARTICLE 26 - CLASSIFICATION**

It is the intent of the Parties that all positions be equitably and fairly evaluated.

## 26.01 CLASS SPECIFICATION

Every job in the bargaining unit shall be covered by a Class Specification. Such class specifications will conform with Appendix "B".

## (a) Copy of Class Specification to Employee

Every employee shall receive a copy of their class specification.

# (b) Copy of Class Specification to Union

The Union shall receive a copy of all class specifications with point and factor detail.

## (c) **Prior to Posting**

Prior to posting of any position the University shall provide the Union with a copy of the class specification with point and factor detail.

## (d) **Overview of Duties**

The class specifications shall contain an overview of the position and a list of the duties, responsibilities and qualifications an individual must possess in order to perform the work within the class specification.

# (e) Shall Not Include

The class specification shall not include such phrases as "desired", "preferred qualifications", nor "all other duties as required", "assists with more advanced duties" or "to act as a deputy for more senior staff members". The phrase "performs duties related to the qualifications and requirements of the job" is permissible.

# (f) Integrity to be Maintained

The integrity of the class specification plan shall be maintained. Towards this end, changes to the class specification shall not violate this Agreement, nor result in an unreasonable combination of functions or work requirements.

#### (g) University Not Bound

The creation of class specifications shall not bind the University to create and/or to fill the positions covered by such class specifications.

## (h) Preparation

Class specifications shall be prepared by the University in consultation with the incumbent(s), wherever possible. Class specifications shall be issued for all positions with a duration of hire of three (3) months or more.

## 26.02 BENCHMARKS

Benchmarks will be established by mutual agreement of the parties during the life of the Collective Agreement.

## 26.03 CLASSIFICATION REVIEW

## (a) Request

Request for a classification review may originate with the employee, the Union or the University. These requests must be in writing to the Human Resources Department with one copy to the Union.

## (b) **Duties Have Changed**

A reclassification review of a job is appropriate only when the job duties or responsibilities have substantively changed because of changes in the work or when the University or Union feels the job has been incorrectly classified.

#### (c) Process

A Human Resources Department representative and Union representative shall review the request with the employee. The classification review procedure must include the following:

- (i) a review of the changes to the class specification submitted in the request;
- (ii) the establishment, if possible, of the date on which the change of job duties occurred;

# 26.04 TEMPORARY RECLASSIFICATION

Prior to the implementation of a temporary reclassification, the University shall consult with the Union. Unless agreed to by the Union, no temporary reclassification shall exceed a duration of four (4) calendar months.

# 26.05 NOTIFICATION

#### (a) **Request by Employee**

Where a classification review is requested by an employee or the Union, the review shall be completed within sixty (60) calendar days. If the employee fails to provide information requested, which is necessary for completion of

the review, within twenty (20) calendar days of receipt of the request, the review shall be deemed to be abandoned. The employee may refile <u>their</u> request.

## (b) Results to Union

The results of the classification review of a new position will be communicated to the Union, in writing, no less than twenty-four (24) hours prior to its posting. The results of the classification review of an existing position will be communicated to the Union, in writing, not less than three (3) days before the public release of the information. Such results shall include the point and factor detail and any revisions to the class specification. If reclassification is not granted, the notification shall contain all reasons for the refusal.

## 26.06 RIGHT TO APPEAL

#### (a) Disagreement

If, within twenty (20) calendar days of receipt of the results of the classification review, the employee or the Union advises the University, in writing, of its disagreement with the result, one University representative and one Union representative, with expertise in job evaluation, shall meet and attempt to resolve the disagreement.

#### (b) Failure to Resolve Dispute

If the parties are unable to resolve the disagreement, the Union may refer the disagreement to the Classification Adjudicator within thirty (30) calendar days of the failure to reach agreement, or the expiry of thirty (30) calendar days of the date of the meeting, whichever comes first.

#### (c) Classification Adjudicator

In the event the University does not render a decision within sixty (60) calendar days per Article 26.05 a), the Union may refer the classification review to the Classification Adjudicator.

#### 26.07 SALARY ON RECLASSIFICATION

#### (a) Increase in Salary

An increase in salary resulting from a classification review shall be retroactive to the beginning of the pay period in which the date of change of job duties occurred, or, when no date can be established, retroactive to the beginning of the pay period in which the request for classification review was received by the Human Resources Department.

Retroactive payment shall be based on all hours worked, including overtime.

## (b) Step Placement

On reclassification of an employee's position, the employee shall move to the same step in the new salary as <u>they were</u> in the former pay group.

## (c) Increment, Seniority and Benefits

Reclassification will not affect an employee's increment date, seniority, or entitlement to fringe benefits.

# (d) Decrease in Salary

A decrease in salary resulting from a classification review shall mean that an employee's salary will be maintained above the maximum of the salary range for <u>their</u> job until such maximum is raised to a level above <u>their</u> salary. Red circled salary treatment. Any employee whose salary is blue circled at the date of this agreement will remain blue circled under this provision.

# 26.08 ADJUDICATION

## (a) Adjudicator

The Parties agree to select a Classification Adjudicator on a rotational basis, depending on availability, from the following list:

Heather Laing John Kinsey

The adjudicator's responsibility is to resolve classification disputes arising from this Article through the application of the agreed upon classification plan. The adjudicator shall proceed as soon as practical to resolve classification appeals by investigating the dispute, consulting with the parties, and may include a hearing and on-the-job review by the adjudicator. The adjudicator's decision will be final and binding on the parties.

#### (b) Resolution

Where the Union has referred a matter to adjudication (the named Adjudicator) will resolve the disagreement as expeditiously as possible. Such resolution shall become the classification level for the disputed position and shall become a benchmark under this Article.

## (c) Disagreements

In resolving disagreements under this Article, the classification Adjudicator shall be governed by the terms of the agreed upon classification plan between the Parties dated April, 1989.

# (d) Costs

All costs of the Classification adjudicator shall be shared equally between the Union and the University.

# 26.09 JOB CLASSIFICATION REVIEW

- (a) The University and the Union agree that they will each provide their own Job Evaluation Specialist on an on-going basis when needed to meet the deadlines set out in the Agreement for resolving classification review requests by employees.
- (b) The University and the Union agree that a "Job Evaluation Specialist" is an individual who has job classification as the primary responsibility of <u>their</u> employment with the University or the Union.

# ARTICLE 27 - PAYMENT OF WAGES AND ALLOWANCES

## 27.01 EQUAL PAY

The University shall not discriminate between male and female employees by employing a person of one sex for any work at a rate of pay that is less than the rate of pay at which a person of the other sex is employed for similar or substantially similar work.

# 27.02 PAY DAYS

#### (a) <u>Biweekly Pay</u>

Employees shall be paid <u>biweekly</u>. Where the pay dates fall on other than normal business days of the University, the applicable pay date shall be on the preceding business day prior to the pay date.

#### (b) Method of Pay

The University shall deposit an employee's pay cheque in an account at a financial institution designated by the employee. This provision applies to all employees paid on a semi-monthly (as opposed to hourly) basis.

#### 27.03 RATES OF PAY

An employee shall be paid for services rendered at the rates of pay set forth in Appendix "E".

#### 27.04 APPOINTMENT TO NEW POSITION

#### (a) Higher Pay Group

An employee who is appointed to a position with a higher pay group shall be paid at a rate in that pay group determined in the following way:

- (i) If <u>they have</u> less than 5,460 hours of seniority at the appointment, <u>they</u> shall be paid at the rate in the pay group which is immediately higher (i.e., the next step on the pay plan) than the rate <u>they</u> received in <u>their</u> former pay group.
- (ii) If <u>they have</u> 5,460 or more hours of seniority at the appointment, <u>they</u> shall be paid at the rate in the pay group which is two (2) steps higher (i.e., two (2) steps on the pay plan) than the rate <u>they</u> received in <u>their</u> former pay group.

# (b) Pay Group

An employee who is appointed to a position with the same or lesser pay group, shall be paid at the rate in that pay group which is equal to or the next lesser, if there is none equal, than the rate <u>they</u> received in <u>their</u> former pay group.

# 27.05 SUBSTITUTION

## (a) Definition

Substitution arises out of the University requesting and an employee agreeing to temporarily perform the duties of another class specification. The request shall be in writing indicating the position title. The University will make every reasonable effort to ensure the equitable distribution of substitution amongst employees.

# (b) Pay

An employee who substitutes shall continue to receive <u>their</u> regular rate of pay as a minimum applicable rate.

# (c) Higher Class Specification

When an employee agrees to substitute into a higher (rate) class specification <u>they</u> shall be paid at the same step <u>they hold</u> in <u>their</u> former position and the pay group for the appropriate classification.

## (d) Calculation of Higher Rate of Pay

- (i) The higher pay rate shall apply for the period worked, providing this period is at least fourteen (14) hours.
- (ii) Effective November 1, 1993, the higher pay rate shall apply for the period worked, providing this period is at least seven (7) hours.

# (e) **Duration**

Except as agreed by the Parties, substitution may not exceed a duration of four (4) months.

# (f) Return to Former Position

Employees return to their former position upon completion of substitution period.

#### 27.06 MILEAGE ALLOWANCE

Authorized employees using their personal vehicle for University business, at University request, will be paid at the prevailing Board rate in accordance with Employee Expense Policy E207, but not less than \$0.55 per kilometre.

#### 27.07 EMPLOYEE'S VEHICLE

## (a) Vehicle

It shall not be a condition of employment for an employee to supply a vehicle. An employee may refuse to use <u>their</u> private vehicle on University business.

#### (b) Driver's License

It shall not be a condition of employment for an employee to hold a valid B.C. Driver's license unless the job specifically requires that <u>they drive</u> a University vehicle.

Notwithstanding the above, the University and the Union agree that the following positions may include a requirement for a valid B.C. Driver's license:

Alumni Relations Officer Audience Services Assistant Business Development & Operations Assistant, Continuing Studies & **Executive Education** Career Development Officer Career Services Assistant Development Officer – Sponsorship Employer Engagement and Experiential Learning Facilitator **Events and Ceremonies Assistant** Facilities Inventory Assistant Film Centre Technical Supervisor First Nations Liaison Officer First Nations Student Recruitment Advisor Health and Safety Emergency Preparedness Assistant International Outreach and Events Officer International Student Admissions and Recruitment Assistant International Student Engagement Facilitator International Student Recruitment and Admissions Advisor Marketing and Communications Assistant, Continuing Studies & **Executive Education** Media Production Specialist Motion Picture/Film Crafts Technician Program Administrator (CSEE) **Programs Assistant** Production Technician

Senior Development Officer Student Recruiter <u>Supervisor, Cap U Lonsdale</u> <u>Supervisor, Recruitment and Transition</u> <u>Visual Media Specialist</u>

Other positions may be added to this list only by mutual agreement of the Union and the University.

## 27.08 TRANSPORTATION

In circumstances where an employee must work after 10:00 pm and before 6:30 am, the University shall, at <u>their</u> request, reimburse <u>them</u> for taxi transportation covering the trip to or from work.

## 27.09 PAY MATRIX (INCREMENTS)

Increments shall be granted to all employees in accordance with the accrual of 1,820 hours of seniority from the date of initial placement or previous increment.

# 27.10 REIMBURSEMENT FOR PARKING EXPENSES WHILE ON UNIVERSITY BUSINESS

Other than when parking at the employee's work place, the University shall reimburse an employee in full for any parking expenses incurred by the employee when using <u>their</u> personal or other vehicle for the business purposes of the University.

## ARTICLE 28 - HEALTH AND WELFARE

#### **28.01 BASIC MEDICAL PLAN**

All employees with an employment status of twenty (20) or more hours per week for four (4) or more consecutive months per year may participate in the Basic Medical Plan, premiums for which will be paid by the University during periods of employment.

# 28.02 LONG TERM DISABILITY

#### (a) **Participation**

All employees with an employment status of twenty (20) or more hours per week for four (4) or more consecutive months per year must participate in the Long Term Disability Plan. The employee shall pay 100% of the premiums for the coverage afforded by the Plan during periods of employment. The Plan shall provide 66.7% of the employee's regular monthly salary up to \$2,500 per month plus 50% of monthly salary above \$2,500 per month.

#### (b) Coverage and Eligibility

Coverage and eligibility shall be governed by the terms of the Plans. The Union is aware of the name of the Insurance Consultant to the University.

See Appendix D for information on the Appeal Process.

#### 28.03 FLEXIBLE BENEFITS PLAN

The University agrees to arrange and administer the Flexible Benefits Plan. Benefits covered include:

- Life Insurance
- Accidental Death and Disablement
- Extended Health
- Dental
- Orthodontic
- Health Funding Account
- Other benefits may be added by mutual consent

The University will provide funding for the default level of coverage (where applicable).

#### (a) Eligibility

(i) Regular Employees

All Regular (Full Time and Part Time) employees with an employment status of twenty (20) or more hours per week must participate in the Flexible Benefits Plan. For Dental Coverage, a new Regular employee

must complete the six (6) month period for eligibility. A Benefit Year is defined as that period commencing on April 1 and ending on the succeeding March 31.

(ii) Temporary Employees

All Temporary employees with an employment status of twenty (20) or more hours per week for four (4) or more consecutive months per year will receive only the Default level of coverage (paid by the University) for those plans for which they are eligible (default levels are described below). These plans include Extended Health, Dental, Life Insurance and Accidental Death and Disablement Insurance. For Dental Coverage for an employee with a definite duration of employment, four (4) clear months of employment must remain after completion of the six (6) month period for eligibility, all of which must be continuous. Temporary employees will not gain access to a spending account by waiving enrolment in any benefit plans.

## (b) Spending Account

(i) Each eligible employee shall have a Spending Account, the value of which shall be the premium for providing coverage as follows:

• Life Insurance	two (2) times annual salary
<ul> <li>Accidental Death and</li> </ul>	two (2) times annual salary
Disablement	
• Extended Health	default level coverage
• Dental	default level coverage
• Orthodontics	default level coverage if eligible

The dollar value of the coverage shall be determined against the employee's status (i.e. single, couple, family with dependents).

- (ii) Eligible employees may increase or decrease their Spending Accounts by choosing to participate in any of the plans at other than the default level of coverage.
- (iii) If a Spending Account has a negative balance, the University may recover the amount necessary to cover that balance by deduction from the employee's salary and/or other amounts owed by the University to the employee.

# (c) Selection

All eligible employees must maintain the mandatory coverage in Group Life Insurance and Accidental Death and Disablement as described in 28.02(g) and 28.02(h). Once every four years, eligible employees shall select their coverage level from amongst the available options. An eligible employee whose family status changes during the four year cycle (i.e. couple to single, single to couple, new dependants) may apply to change <u>their</u> dependant level based upon that change in family status.

# (d) Administration of the Plan

During each year, all enrolled employees shall receive an accounting of their Spending Account for the current Benefit Year.

Following receipt of the accounting, each enrolled employee shall select at least one designation for any unspent money in the Spending Account. Such designation could include:

- Directing it to be allocated to the Health Funding Account
- Depositing it in the University's group Registered Retirement Savings Plan
- Directing it to be paid as taxable income
- Directing it to other non-taxable purposes (such as professional development)

If the employee does not make a choice, the default allocation will be having it paid as taxable income.

Any employee may appeal to the Joint Benefits Review Subcommittee of the Joint Standing Committee about any aspect of the Flexible Benefits Plan, including eligibility for coverage, eligibility of particular expenses for reimbursement, and the amount of premium prepayment required. The committee may recommend any action that it feels is appropriate in the circumstances.

# (e) Dental Plan Coverage

Default level is:

- No annual deductible
- Plan A: 100% reimbursement for eligible basic and preventive work
- Plan B: 75% reimbursement for eligible restorative work
- Plan A & B combined: \$1,750 per calendar year maximum reimbursement per mouth (employee and dependants covered under the plan)

Options:

The deductible is increased from zero to \$100 per mouth per year (maximum of four mouths), or

The deductible is increased from zero to \$250 per mouth per year (maximum of four mouths).

# (f) Orthodontic Coverage

Default level provides per dependant under the age of 19, reimbursement of 50% to a lifetime maximum of \$2,500 per dependant.

Option is for no coverage.

## (g) Life Insurance Coverage

Default level is an amount equal to two (2) times annual salary.

Options:

Coverage may be increased (to a maximum of \$305,000) or decreased (to the minimum mandatory level of coverage of \$5,000) in units of \$50,000.

Coverage ceases at age 65.

#### (h) Accidental Death and Disablement Coverage

Default level is an amount equal to two (2) times annual salary.

Options:

Coverage may be increased (to a maximum of \$305,000) or decreased (to the minimum mandatory level of coverage of \$5,000) in units of \$50,000.

Coverage ceases at age 65.

# (i) Extended Health Coverage

Default level is:

- \$25 annual deductible for single, couple, or family
- 80% reimbursement of eligible expenses
- Prescription drugs; prescription drug reimbursements will only be issued for those prescription drugs that are included under the BC Provincial Pharmacare Formulary.
- Ambulance
- Hospital user fees
- International travel assistance
- Vision care (reimbursement to a maximum of two hundred dollars (\$200) per year or four hundred dollars (\$400) every two years per eligible employee or dependent)
- All other items that were covered in the Extended Health plan with Canada Life effective May 1st, 1992, with additions dated March1st, 1996.

#### Options:

Deductible raised to \$250 per year; or Deductible raised to \$500 per year and no prescription drug coverage.

Effective January 1, 2016, the extended health coverage will be as follows:

Default level is:

- \$50 annual deductible for single, couple, or family
- 80% reimbursement of eligible expenses
- Prescription drugs; prescription drug reimbursements will only be issued for those prescription drugs that are included under the BC Provincial Pharmacare Formulary.

- Ambulance
- Hospital user fees
- International travel assistance
- Vision care (reimbursement to a maximum of two hundred dollars (\$200) per year or four hundred dollars (\$400) every two years per eligible employee or dependant)
- All other items that were covered in the Extended Health plan with Canada

Life effective May 1st, 1992, with additions dated March1st, 1996.

Options: Deductible raised to \$275 per year; or Deductible raised to \$525 per year and no prescription drug coverage.

# (j) Health Funding Account

No default level exists, but eligible employees may once per year allocate amounts for the Health Funding Account. This account may be used for medical and dental claims and recognized alternate professional health care coverage not eligible for coverage under other aspects of the flexible benefits plan, that are considered eligible medical expenses by Revenue Canada. Any funds not used after two years are forfeited in order to maintain the insurance aspect of the plan.

# 28.04 BENEFITS NOT CHANGED EXCEPT THROUGH NEGOTIATIONS

The insurance and welfare benefits shall not be changed or modified during the life of this Agreement except by negotiation and the mutual agreement of both parties.

# 28.05 COVERAGE REQUIRED BY LAW

The University shall deduct such sums as are required by law and make such contributions on behalf of employees as legislation binding upon the University may require. Examples are: E.I., W.C.B., Municipal Pension Plan where applicable.

# 28.06 EMPLOYEE'S RESPONSIBILITY FOR BENEFIT COVERAGE

# (a) Benefit Plans

All benefit plan coverage, terms, conditions and specific eligibility requirements shall at all times be covered by the actual terms and conditions of the benefit plans as amended from time to time. Any benefit plan description contained in this agreement is provided only for the purpose of general information.

# (b) Employee Responsible for Details

It is understood and agreed that it is the responsibility of each employee to be familiar with the specific details of coverage and eligibility requirements of all benefit plans and that neither the Union nor the University has any direct responsibility for insuring that all requirements for eligibility or conditions of coverage or entitlement to benefits are made by the employee beyond the obligations specifically stipulated in this Agreement.

## (c) Requesting Information

The actual policy contracts arranged with insurers shall be deemed to be an integral part of this agreement. The University and the Union shall make every effort to provide promptly any information regarding the benefit plans, to any employee requesting it.

#### 28.07 LIMITATION OF LIABILITY OF UNIVERSITY

The Union recognizes and agrees that the University's obligation and liability with regard to providing the benefit and insurance coverages agreed to herein is in all events limited to arranging for the underwriting of coverages by insurers and internal procedural administration of the plans. The University cannot be held liable for refusal by insurers to underwrite any plan, for cancellation of coverage by insurers, or for the rejection of any claim or claims by insurers.

## 28.08 "BRIDGE PERIOD" BENEFIT COVERAGE

During <u>their</u> "Bridge Period", an employee shall maintain the coverage of those plans <u>they have</u> been participating in. Both the University and the employee shall continue paying their share of the premiums, and the employee's share must be paid in advance.

#### **28.09 BENEFIT COVERAGE DURING LEAVE OF ABSENCE**

# (a) Leave of Absence Without Pay

The parties to this Agreement recognize and agree that except where specifically provided in this Agreement, all benefits and entitlements provided by this Agreement are suspended for an employee who is absent on a leave of absence without pay.

# (b) Employee Must Maintain Coverage

Where an employee has been granted an approved leave of absence without pay, in excess of thirty (30) calendar days, and the employee has been participating in the following benefit coverages:

Group Life Insurance	28.03(g)
Accidental Death & Disablement Benefit	28.03(h)
Long Term Disability	28.02
Dental Plan	28.03(e)

the employee must, as a condition of being granted the leave of absence, maintain the coverages in those plans. The employee must pay 100% of the premiums for those coverages during such leave in advance.

# 28.10 ENROLMENT

Enrolment in any Benefit or Insurance Plan is not completed until the employee has completed an Application/Waiver form (Appendix "C"), and application cards have been signed by the employee, acknowledged by the University and acceptance by the Insurer has been confirmed.

# 28.11 DEDUCTIONS: PREMIUMS AND CONTRIBUTIONS

The administrative procedure applicable to a benefit or insurance plan shall determine if the deduction of an employee's contribution toward the premium payments precedes or follows the coverage period to which it is applicable.

## **ARTICLE 29 - GENERAL CONDITIONS**

## **29.01 POLITICAL ACTIVITY**

The University agrees not to apply restrictions on employees who wish to engage in political activities on their own time.

## 29.02 CONTRACTING OUT

#### (a) Work Normally Performed by Employees

The University agrees not to contract out any work normally performed by employees covered by this Agreement, as of the date of signing of this Agreement, which would result in the laying off or displacement of such employees, or which would delay the recall of employees who have been laid off or the rehire of employees on the Recall List.

#### (b) Contracts for Work

The University will consult with the Union if any contract for work which could be performed by members of the bargaining unit falls for renewal during the period when employees are laid off or displaced or on the Recall List.

#### (c) Discussions with Union

Any contract which the University contemplates and which is a break in the present pattern of work contracted out by the University will be discussed with the Union.

#### (d) Exceptions

Notwithstanding the aforementioned provisions of this Article, it is mutually agreed to exempt from application of 29.02(a) and 29.02(b) the following existing areas of contracted services: janitorial, security, payroll, accounting and food. This provision has application until and unless a ruling or direction of the Labour Relations Board of British Columbia nullifies its effectiveness.

#### 29.03 CONSULTATION

## (a) Change in Equipment

Where a requirement arises to change, replace or add to existing equipment, employees using such equipment must be consulted prior to the rental or purchase of such equipment for their working area.

#### (b) Renovations

Where renovations which may affect the working area of employees are planned for an existing building, employees from the areas concerned shall be consulted regarding such renovations, before they may begin.

## (c) Consultation

Where permanent change is considered in the location of work areas or in working procedures, employees concerned must be consulted before any changes may begin.

# (d) Implication of Consultation

The consultation as noted above does not in any way imply that the University's decision to act is subject to employee consent, only that <u>they are</u> consulted.

## 29.04 COMMUNITY FACILITIES

Employees shall be allowed the use of University recreational facilities such as tennis courts, exercise room, etc. during hours scheduled for such purposes.

## **29.05 ABSENCE WITHOUT LEAVE**

#### (a) Absence Without Notification

If an employee is absent without having notified the Human Resources office of the University, and the University has been unsuccessful in a reasonable attempt to contact the employee, such absence may be treated by the University as just cause for discipline.

#### (b) Repeated Absence

Repeated absence without leave may be just cause for termination.

# **29.06 RETIREMENT**

## (a) Age of Retirement

An employee may retire effective at the end of the pay period in which the employee attains the age of 65.

#### (b) **Rights and Privileges**

An employee working beyond the age of sixty-five (65) shall continue to receive all the rights and privileges provided in this Agreement except those specifically excluded by legislation or the provisions of a benefit plan.

# (c) Library Card and Access to Facilities

A retiree shall be provided with a library card at no charge as a community borrower and allowed access to the University's facilities as if <u>they</u> were a registered student.

## (d) Tuition Costs

The University shall waive all tuition costs of any courses taken at the University (subject to the restrictions in 29.10 (b) and (c)) by retirees provided the employee has completed a minimum of five (5) years service with the University and no additional costs result to the University by the employee's enrolment in the course.

## (e) Employee Assistance Plan

Retirees shall have access to the Employee Assistance Plan as if they were employees of the University for a period of two (2) years immediately following the date of retirement.

# 29.07 EARLY RETIREMENT PLAN

# (a) Eligibility

The University shall offer to any employee, who requests early retirement, the choice of one of the early retirement incentive alternatives described herein, provided the employee meets the following qualifications:

- is age 55 or over;
- has a minimum of ten (10) years of pensionable service under the Municipal <u>Pension</u> Plan;
- is a regular employee at the time of early retirement;
- is on the maximum step of the salary scale;
- elects to cease employment with the University for the purposes of retirement.

The University is not required to offer early retirement to an employee under this provision where the costs of such early retirement will not be entirely recovered through the lower costs of a replacement hire in the two years following the effective date of the retirement.

# (b) Agreement

An employee has the right to accept or decline an early retirement incentive offer made by the University within thirty (30) days of the offer being proposed. In the event of acceptance of an offer of early retirement, the employee's date of retirement or commencement date of leave under Option "B" hereof shall be effective on a date mutually agreed upon between the employee and the University.

- (i) Agreement shall be in writing and shall specify the early retirement date with the incentive option agreed upon.
- Acceptance must take place before the termination date of this Agreement. Retirement or commencement of leave under Option "B" hereof may take place after the termination date of this Agreement.

## (c) Benefit Options

(i) Lump Sum Payment (Option "A")

The retirement benefit will be paid in one sum on the date of retirement or on an agreed-upon deferred date or in predetermined instalments acceptable to the employee, and will be based on scale salary without allowances in the following amounts:

Full Years to Retirement Payout

1	Up to 15% of annual salary
2	Up to 30% of annual salary
3	Up to 45% of annual salary
4	Up to 60% of annual salary
5 or more	Up to 75% of annual salary

(ii) Purchase of Past Service Contributions for Early Retirement (Option "B")

The University will purchase, on behalf of the employee, past service contributions to the Municipal <u>Pension</u> Plan equivalent to the value of the lump sum payment to provide for immediate early retirement benefits.

#### (d) Financial Counselling

Each employee who accepts one of the foregoing incentives is entitled to attend a financial Planning Workshop and to receive three (3) subsequent personal financial consultations conducted by a firm of qualified financial consultants selected by the University. Fees for consultative sessions to a maximum of \$80 per session will be borne by the University.

## **29.08 EMPLOYEE FILE**

#### (a) Inspection

An employee may, at a mutually convenient time, inspect the contents of <u>their</u> Personnel File. Any document found therein may be copied. If the contents of any document on file are disputed, the eventual resolution, be it through the grievance procedure or otherwise, shall be entered in <u>their</u> file.

## (b) Incidental Information

The University and the Union agree that in the interest of good communication, employees shall receive copies of incidental written complaints and positive reports on their performance which, although they do not constitute Personnel file documentation, have been brought to the attention of an Administrative Head. Employees will be given the opportunity to reply to incidental complaints.

# (c) Union Access to Employee Personnel Files

A representative of the Union shall have the right to read and review an employee's personnel file on written authorization of the employee and upon reasonable notice and by written request to the Human Resources Department. On request, the Union Representative shall be provided with copies of all pertinent documents.

See also Article 10.06.

# **29.09 EVALUATION**

When an employee's performance is evaluated, the employee concerned shall be given the opportunity to read and review the evaluation. The procedure shall provide for the employee's acknowledgement through signature that <u>they have</u> read the evaluation and <u>have</u> been given a copy of it. The procedure shall also provide that an employee may comment on the evaluation and that <u>their</u> comment shall be considered as part of that evaluation.

# 29.10 CAPILANO UNIVERSITY COURSES

# (a) Credit Courses

Any employee with an employment status of twenty (20) or more hours per week may register for unlimited Capilano University credit courses without payment of tuition fees and registration fees.

# (b) Continuing Education Courses

Any employee and <u>their</u> current spouse and children may register for a total of three (3) Capilano University continuing education courses (credit free) in any calendar year without payment of tuition fees. In the case of continuing education courses (credit free) the employee shall be entitled to the waiver only after the minimum enrolment has been met by fee payers. There shall be no waiver for non-tuition costs such as travel costs for field trips, excursions, and tours.

#### (c) Cost Recoverable/Non-base Funded Courses

In the case of credit courses that are offered on a cost recoverable basis (i.e. non-base funded courses), only the amount of tuition equivalent to the fees

for a regular base funded credit course shall be waived; any additional tuition costs must be paid by the employee.

# (d) Time Spent Attending a Course

Such courses shall be taken outside the employee's normal working hours. If such courses are only offered during regular working hours, permission shall be obtained from the Administrative Head responsible to enable course attendance provided that arrangements satisfy the making up of time absent at no cost to the University. Where permission has been granted, the employee may request to use <u>their</u> accrued and unused vacation time, subject to the Administrator's approval, instead of making up the time.

## (e) Commencement and Completion

All courses taken must commence and end within the duration of an employee's employment status. If an employee ceases to be employed by the University before completion of a course, <u>they</u> may be required to pay the tuition fees for the course.

# **29.11 MULTIPLE APPOINTMENTS**

Where an employee holds more than one (1) appointment with the University within the Union's certification, the University shall review Health Benefit entitlements on an "individual case" basis and extend where possible benefit coverage based on multiple appointments.

#### 29.12 PERSONAL AND SEXUAL HARASSMENT

#### (a) Prohibition Against Personal and/or Sexual Harassment

The University recognizes the right of all employees to work in an environment which is free of personal and/or sexual harassment. Accordingly, the personal and/or sexual harassment of any employee is prohibited.

The Parties also recognize the right of students to learn in an environment free from personal and/or sexual harassment.

#### (b) Definition of Personal Harassment

Any discriminatory behaviour at or related to the workplace which denies an individual their dignity or respect or which adversely affects their terms or conditions of employment or their job security or career advancement prospects by creating an intimidating, coercive, restrictive, offensive, embarrassing or humiliating work environment is considered to be personal harassment.

Such prohibited discriminatory behaviour is defined as any discrimination on the basis of race, national or ethnic origin, colour, citizenship, place of residence, age, sex, sexual preference or orientation, marital status, family status, number of dependants, pregnancy or childbirth, physical or mental disability where the disability does not render the employee incapable of fulfilling <u>their</u> employment duties and obligations, conviction for which a pardon has been granted, political or religious affiliation or beliefs, or membership or activity in any trade union.

## (c) Definition of Sexual Harassment

Sexual harassment is comment or conduct of a sexual nature, including sexual advances, requests for sexual favours, suggestive comments or gestures, or physical contact, including assault, when any one of the following occurs:

- (i) the conduct is engaged in, or the comment is made by, a person who knows, or ought reasonably to know, that the conduct or comment is unwanted or unwelcome;
- (ii) the conduct or comment has the effect of creating an intimidating, hostile or offensive environment, and may include the expression of sexist attitudes, language and behaviour;
- (iii) the conduct or comment exploits the fiduciary relationship between students and employees of the University;
- (iv) the conduct or comment is accompanied by a reward, or the express or implied promise of a reward, for compliance;
- (v) the conduct or comment is accompanied by reprisal, or an express or implied threat of reprisal, for refusal to comply;
- (vi) the conduct or comment is accompanied by the actual denial of opportunity, or the express or implied threat of the denial of opportunity, for failure to comply.

#### (d) Complaints

An employee who believes that <u>they have</u> a complaint of sexual harassment or personal harassment is encouraged to make a direct request of the <u>respondent</u> that the offensive behaviour or actions cease. If the request is unsuccessful, or if it is considered inappropriate, or uncomfortable to make such a request, the complainant may seek the confidential advice of the Union or the Harassment Policy Advisor.

An employee may initiate a grievance under this clause at any step of the grievance procedure or the procedures set out in the Capilano University Harassment Policy and Procedures document, dated July 18, 2018. Grievances under this clause will be handled with all possible confidentiality and dispatch. An employee who commences a complaint under the Capilano University Harassment

Policy and Procedures document shall not forfeit their entitlement to pursue a complaint through the grievance procedure under the Collective Agreement.

- (ii) If the alleged offender is an employee covered by the Union's certification, they shall be entitled to notice, as soon as possible, of the substance of the personal and/or sexual harassment complaint made against them.
- (iii) During any investigation and/or grievance procedure, the University agrees to monitor the working environment of individuals involved to protect their rights pursuant to Article 29.12 (a).
- (iv) The university agrees to share all investigative findings from the grievance process subject to the approval of the complainant, and relevant redaction where applicable.

# (e) Personal Harassment

The Union will incorporate the definition of Personal Harassment developed by the Personal Harassment Committee at the University into the Collective Agreement, replacing the documents signed January 29, 1993, provided the Union membership at Capilano University ratify the replacement definition by a two-thirds majority, at a Union membership meeting.

# 29.13 ELECTRONIC MONITORING

The University agrees to provide the Union with notice of equipment and facilities which are to be installed for the purpose of monitoring and measuring individual employee performance.

# 29.14 PERSONAL DUTIES NOT REQUIRED

No employee shall be required to perform work or duties that are unrelated to the work of the University and are of a personal nature for any other employee of the University.

# 29.15 PROTECTION AGAINST LEGAL ACTION

The University shall:

#### (a) **Protection for Employees**

exempt and save harmless each employee from any liability action arising from the proper performance of <u>their</u> duties for the University, and

## (b) **Protection for Retired Employees**

exempt and save harmless each retired employee from any liability action arising from the proper performance of <u>their</u> duties for the University prior to <u>their</u> retirement, and

## (c) University to Assume Costs

assume all costs, legal fees and other expenses arising from any such action.

# 29.16 LEGISLATION

The Parties subscribe to the principles of the B.C. Human Rights <u>Code, B.C.</u> <u>Labour Relations Code, B.C. Employment Standards Act</u> and the Canadian Charter of Rights and Freedoms insofar as this legislation establishes minimum acceptable standards. It is agreed that more favourable provisions of this Agreement shall prevail.

# **29.17 LIE DETECTOR TESTS**

The University agrees that polygraph or similar lie detector tests will not be used. Without limiting the generality of the foregoing, the University agrees that this prohibition against use of polygraph or similar lie detector tests shall apply equally during pre-hiring of any person into the bargaining unit.

# **ARTICLE 30 - EMPLOYEE ASSISTANCE PLAN**

## 30.01 ADMINISTRATION OF EMPLOYEE ASSISTANCE PLAN

The Parties agree to an Employee Assistance Plan to be administered by a University/Union/CFA committee. Such plan will be 100% employer paid; will have no relation to discipline or career development decisions and will treat employee involvement as absolutely confidential.

# **ARTICLE 31 - DURATION OF CONTRACT**

## 31.01 DURATION

This agreement shall be binding from July 1, 2019 to midnight, June 30, 2022.

#### **31.02 NEW AGREEMENT**

Either party to this Agreement may, not more than four (4) months and not less than one (1) month prior to June 30, 2022, present to the other party, in writing, proposed terms of a new or further agreement, or amendments to this agreement.

#### 31.03 CONTINUATION OF PRESENT AGREEMENT

In the event that a new Collective Agreement is not entered into by the parties on or before the 30th day of <u>June, 2022</u>, this Agreement shall continue in force and effect until the parties ratify a new or renewed Collective Agreement.

#### 31.04 LABOUR RELATIONS ACT 50(2) EXCLUSION

The parties expressly agree that the operation of Section 50(2) of the Labour Relations Code of British Columbia is specifically excluded.

#### **31.05 EFFECTIVE DATES**

Where this Agreement contains terms or conditions which are changed from those contained in the prior Agreement, such shall be effective on the date of signing, except where the agreement change can only be available on the first of the month, in which case it shall be effective the first of the month next following the signing of the agreement.

#### 31.06 ERRORS & OMISSIONS

Whereas substantive changes to this Agreement shall not occur, any errors or omissions may be corrected by mutual agreement of both parties.

## **31.07 PAYMENT OF COLLECTIVE AGREEMENT**

The University will pay for and provide the Union with <u>30030</u> Collective Agreements at the conclusion of negotiations. A copy of the collective agreement will be posted on the website.

## **ARTICLE 32 - EXPLANATORY NOTES**

#### **32.01 INTENT AND MEANING**

Except where specific definitions or words, terms, or phrases are expressly provided within specific provisions of this Agreement, the following shall be used to establish the intent and meaning of the language of this Agreement:

"Agreement" - means the entirety of the current Collective Agreement entered into between the Union and the University, including the appendices and any alteration or amendment which may from time to time be mutually agreed to by the parties, and specifically incorporated into the Agreement.

"As required basis" - means an employee's work day and work week are not precisely scheduled at the time of hire, and reflect the specific need at any given time. (see 15.04)

**"Blue-circled"** or **"Protected Salary Treatment"** - shall mean that an employee's salary will be maintained above the maximum of the salary range for <u>their</u> job and that such salary will be increased by all subsequent negotiated and length-of-service salary increases until the incumbent leaves the position.

**"Calendar Days"** - where this phrase is used to designate a time span for purposes of giving a notice, or relative to specific action or a required response by the parties to the Agreement, it shall be taken to intend the inclusion of calendar weekends and holidays.

"Class Specification" - a statement of the job to be done with a brief list of duties and responsibilities in order of importance and the qualifications an individual must possess in order to perform the job.

"University" - means Capilano University, a party to the Agreement.

"Consultation" - means a discussion process intended as a vehicle through which the parties may make known their respective views and opinions.

"Discipline" - shall include reprimand, suspension or discharge.

"Discharge" - means employment is brought to an end for disciplinary reasons.

**"Hours of work per week"** - shall include an equivalent number of hours which result from an arithmetic average of the hours worked in a modified work week schedule cycle, in addition to those definitions contained in Article 13.

"Invited to attend" - means an employee has the choice whether to attend or not.

**"Layoff List"** - is a list maintained in a seniority order by classification/work area of all those laid off pursuant to Type 1 Layoff (see 25.03 for precedence of Layoff List for staffing purposes).

**"Leave of absence without pay or leave without pay" -** applies only to times when the employee would otherwise have worked.

"New position" - means creation of a wholly new job position (assigned a new position number) within the University, which position did not exist within the University at the date of execution of this Agreement. It does not include positions which have merely been given new titles, or positions which have been altered through changes in their duration or the number of hours of work required per week.

**"Officially requested"** - means a request originating from a person who has been designated by the University as one who has the authority to make such a request.

"Parties" - shall mean the Union and the University.

"**Previous Incumbent**" - means an employee who was appointed to a temporary position pursuant to the competition procedure. (See Article 12.13(ii)).

**"Red-circled"** - shall mean that an employee's salary will be maintained above the maximum of the salary range for <u>their</u> job until such maximum is raised to a level above <u>their</u> salary.

**"Required to attend"** - means an employee has been informed by a person with University authority that <u>their</u> attendance is mandatory.

"Seniority List" - is a list maintained in seniority order of all current and former employees who have seniority (see 11.07).

"Shift" - means an employee's approved work day.

"Supervisor" - means the individual to whom an employee normally reports or from whom an employee normally receives <u>their</u> direction.

"Suspension" - means the temporary removal of an employee from active employment for disciplinary reasons.

"Union" - means the <u>MoveUP (</u>Canadian Office and Professional Employees' Union, Local 378).

"Union Representative" - shall mean any employee authorized by the Union to act on its behalf.

# **32.02 OBJECTIVE INTERPRETATION**

Where no specific definition of a word, term or phrase is expressly provided in this Agreement, such word, term, or phrase shall be interpreted objectively, not subjectively; and according to common and normal grammatical usage.

# 32.03 NOTIFICATION REQUIREMENTS

Where this Agreement requires notice to be given between the parties to the Agreement, such notice shall be in writing, and mailing of such notice by ordinary mail shall be deemed to be effective notice, (as determined by the postmark), unless the Agreement specifically requires another mode of delivery or service.

## ARTICLE 33 - WORKLOAD

## **33.01 REASONABLE WORKLOAD**

Employees' workload shall be fair and reasonable when compared with the workload of other staff employees of the University. This provision is not intended to diminish the University's ability to address employee performance issues.

## **33.02 WORKLOAD DISPUTES**

When an employee believes that <u>their</u> workload is not fair or reasonable as cited in Article 33.01 <u>they</u> may refer the matter to the Joint Workload Review Committee. The committee will review the complaint and issue recommendations on it to the parties.

#### **33.03** JOINT WORKLOAD REVIEW COMMITTEE

The parties agree to establish a Joint Workload Review Committee comprised of an equal number of representatives of the University and of the Union, with each party selecting its own representatives.

The committee shall:

#### (a) Complaints

investigate complaints concerning workloads;

#### (b) Recommendations

make recommendations to the parties for corrective action if warranted.

The committee shall meet in response to specific complaints. Employees, who are committee members, will receive time off with pay for their work on the committee.

## **ARTICLE 34 - WORKERS' COMPENSATION SUPPLEMENT**

## 34.01 LEAVE OF ABSENCE

An employee shall be granted a leave of absence by the University while on Workers' Compensation, which time shall be deemed to be time worked. During such leave of absence the employee shall continue to accrue seniority as per Clause 11.03(a)(ii) and shall continue to be entitled to full benefits and all other rights and entitlements under this Agreement as if <u>they</u> had remained working, except that salary payment shall be subject to Clause 34.02 below.

#### 34.02 WORKERS' COMPENSATION SUPPLEMENT

The University shall pay semi-monthly to an employee who is in receipt of Workers' Compensation Benefits a supplement to provide the employee, including the payment from Workers' Compensation, an amount equal to the employee's normal semi-monthly net earnings.

IN WITNESS WHEREOF, the Parties have executed this Agreement on the <u>18</u> day of

June , 2020. Nathan Beausoleil

Union Representative

Christy Slusarenko Union Vice President

Brenda Rovner Bargaining Committee

abito

Eric Roberts Bargaining Committee

Darin Feist Bargaining Committee

Wendy Dansie

Wendy Dansie Bargaining Committee

Mike Knudson Associate VP, Human Resources

Sara Colliss Manager, Human Resources

Ryan Blades Director, Facilities Services & Campus Planning

Hayley Elliot Associate, Human Resources

**POST SECONDARY EMPLOYERS' ASSOCIATION** by:

Michael Marchbank, PSEA Chair, Board of Directors

**APPENDIX A** 



NAME LAST NAME, FIRST NAME, INITIAL HOME PHONE CELL PHONE ADDRESS OFFICE PHONE OFFICE FAX CITY POSTAL CODE SOCIAL INSURANCE BIRTHDATE (Month/Day/Year) NUMBER HOME EMAIL: WORK EMAIL: Previous OPEIU, Local 378 member? \_Yes No\_ Previous CLC/AFL-CIO Union Membership? \_\_Yes No\_\_\_\_ (If yes, attach withdrawal card for initiation fee Waiver)

I am applying for membership in MoveUP and either:

In applying for membership I understand that I am part of the already certified bargaining unit at <u>Capilano University</u> or;
 In applying for a membership I understand that the Union intends to apply to be certified as my exclusive bargaining agent and to represent me in collective bargaining.

I hereby authorize <u>Capilano University</u> to deduct the appropriate dues from my salary or wages and remit to the Movement of United Professionals.

I further authorize <u>Capilano University</u> to deduct and remit to the Movement of United Professionals any initiation fees, assessments, or arrears as requested by the Union.

I further authorize <u>Capilano University</u> to release any information regarding my employment to the Movement of United Professionals for collective bargaining, representation, grievance or administrative purposes in accordance with Section 33, subsection (L) of the Freedom of Information and Protection of Privacy Act or the Personal Information Protection Act.

DATE	(Month/Day/Year)
------	------------------

MEMBER'S SIGNATURE

#### <u>EMPLOYER – PLEASE COMPLETE THIS SECTION:</u>

DATE OF HIRE WORK LOCATION			TITLE LOYEE I.	D.#	
FULL TIME REGULAR	FULL TIME TEMPORARY	PART TIME REGULAR		Salary \$ Hourly/Daily/Bi-weekly/Mo	nthly
PART TIME TEMPORARY	SEASONAL	CASUAL		Salary Group	
***	ALL SECTIONS OF TH PLEASE RETURN		_		

301 - 4501 Kingsway, Burnaby, B.C. V5H 0E5

#### APPENDIX B CLASS SPECIFICATION INFORMATION AND CRITERIA

(Reference Article 26 of the Agreement)

An integral part of the Classification Plan is the Class Specification. It acts as a guide in determining the nature and scope of work assigned to an employee. It assists in determining the type of employee needed to perform certain functions within a department or section. The Class Specification provides an employee with the requirements and qualifications of a position so that <u>they</u> may prepare <u>themselves</u> accordingly.

<u>The Class Specification</u> includes a Class Title, a statement of the distinguishing features of work of the class, a list of examples of work performed, a section in which are listed those knowledges, abilities and skills which are needed by the applicant or incumbent for successful performance of the work, and a statement of required training and experience for a new appointee, which would ordinarily provide a person with the necessary qualifications. The class specification may also include areas relating to: decisions required; supervision received; effect of errors, contacts; supervision of other employees; physical demands; and working conditions. Each specification must be considered as a whole for the purpose of describing and defining a class, or for allocating a position to a class. Although each section has its special uses, an adequate understanding of the class can be gained only by considering all parts together.

<u>The Distinguishing Features of Work</u> section provides a description of the salient features which distinguish the class from other classes of work. This is an evaluation of the class in terms of the elements of difficulty and responsibility involved in the work. As such, it serves to establish the level of the class and is one of the sections most heavily relied upon in allocating positions to the class.

The section setting forth the <u>Examples of Work Performed</u> is intended to enable the reader to obtain a more complete picture of the actual work performed in positions in the class. The list is descriptive and is not intended to limit administrative authorities in the assignment of duties and responsibilities to the position. It does not describe all the work performed. It merely serves to illustrate the typical portions of work.

<u>The Requirements of Work</u> section lists those knowledges, skills and abilities which are necessary for successful performance of the work of the class. These are written in terms that are requirements of an applicant before appointment. The levels of knowledge used in this section are expressed in five broad degrees; they include: some knowledge, working knowledge, considerable knowledge, thorough knowledge and extensive knowledge. As a guide for interpreting the Specifications, definitions for these terms are listed:

Extensive knowledge implies the most advanced degree of knowledge necessary for complete mastery and understanding of the subject.

<u>Thorough knowledge</u> implies almost complete coverage of the subject matter area. For successful performance of the work sufficient comprehension of the subject matter is necessary to solve unusual as well as commonplace problems.

<u>Considerable knowledge</u> implies sufficient acquaintance with the subject to perform the work with little direct supervision and a journeyperson comprehension of normal work situations.

Working knowledge implies an awareness of the subject sufficient to work effectively in a limited range of work situations.

Some knowledge implies familiarity with the elementary principles and terminology of the subject matter.

Skills refer to and are limited to manual skills.

<u>Abilities</u> refer to the capacities making possible the application of knowledge and skills to work situations.

<u>Requirements</u> such as honesty, integrity, initiative and sobriety are implied for all positions.

In some specifications a statement of necessary special requirements is included. This is done only when statutory or other legal requirements establish a definite minimum qualification such as possession of a license or registration certificate before appointment.

#### Conception of the Class Specification

It must be emphasized that each specification must be considered as a whole for the purpose of describing and defining a class. Although each section has its special usages, an adequate conception can be gained only by consideration of all the sections in relation to each other.

#### Flexibility of the Classification Plan

In order to serve its purpose as an administrative tool in developing the Human Resources Program, the Classification Plan is not intended to be rigid or binding to future changes in organization, individual positions or functions. The Classification Plan is an analysis and description of the work presently performed in the University. Because organization, positions and functions are constantly changing, the Classification Plan must also be changed to keep abreast of the requirements of the University. Unless it is made a perpetual inventory by creating, abolishing and changing classes as required, it will be out of date and will be a hindrance rather than an aid to effective Human Resources administration. In addition to day-to-day adjustments, periodic audits of the whole plan are essential for maintaining good relationships and preventing accumulation of discrepancies.

#### APPENDIX C(a) APPLICATION AND WAIVER FORM RE: REGULAR EMPLOYEES BENEFITS (Reference – Article 28.10)

In accordance with the provisions of Article 28 of the Collective Agreement between the University and the Union, I \_\_\_\_\_\_, acknowledge

and accept that:

- a) Disability Income Plan coverage is mandatory for all Regular employees;
- b) the information below represents my election with regard to the benefit plans available;
- c) as a condition of enrolment, I must maintain benefit coverage for the four-year period of the benefit selection and that coverage cannot be changed during this period unless there is a change in my employee status or in my status as single/couple/family;
- d) my application must be accepted by the Insurer(s) and the acceptance confirmed to me before the coverage becomes effective.

#### Dependent status (check one box below)

- $\Box$  Single
- $\Box$  Couple (one dependent)
- □ Family (more than one dependent)

I herein make application for enrolment in those benefits indicated below, or I have on this date voluntarily elected to waive my rights to enrol in those benefits indicated below:

Benefit	Enrolment Requested	Enrolment Waived
Basic Medical (M.S.P.)		
Extended Health		
Dental *		
Orthodontics *		
Additional Coverage Possible		

\* An employee must complete the six (6) month period of eligibility.

#### CAPILANO UNIVERSITY

Signature of Employee

Received per

Date

Date

#### APPENDIX C(b) APPLICATION AND WAIVER FORM RE: TEMPORARY EMPLOYEES BENEFITS (Reference – Article 28.10)

To: Capilano University

In accordance with the provisions of the Article 28 of the Collective Agreement between the University and the Union,

I,	_(ple	ase print	t name)
----	-------	-----------	---------

(a) hereby acknowledge that the following coverages are mandatory for eligible employees as a condition of employment:

Group Life Insurance Accidental Death & Dismemberment Insurance Short Term Disability Long Term Disability

- (b) hereby acknowledge that whereas I am eligible and entitled to participate in the following coverages, that
  - (i) I herein make application for enrolment in those benefit plans as indicated, or
  - (ii) I have on this date voluntarily elected to waive my rights to participate in those benefit plans as indicated.

	Enrolment Requested	Requested Waived **
Basic Medical		
Extended Health		
Dental *		

- \* An employee must complete a six (6) month eligibility period and still have four (4) clear months of employment.
- \*\* (Note: Future participation, if permitted, is at the employee's expense with respect to satisfying the Insurer's requirements for late enrolment.)

Employee's signature:	Date:
Employee Number:	
Received by Capilano University per:	Date:

#### APPENDIX D CARRIER'S APPEAL PROCESS

#### TERMINATION OR DENIAL OF A DISABILITY CLAIM

An employee wishing to appeal either the termination or denial of a disability claim must do so in compliance with the requirements of the Carrier. The employee can contact Human Resources to request information about the Carrier's appeal process.

#### PAYMENT OF MEDICAL REPORTING

Any specialist or physician's charges incurred in providing medical reporting, either written narrative or completion of claims forms, are paid by the employee. This includes the expense of having a physician provide medical information for the purpose of appealing the termination or denial of a disability claim. The insurer will only pay for those reports which are written in response to questions that the Carrier has asked directly of the physician or for reports resulting from an employee's attendance at an independent medical examination required and scheduled by the Carrier.

Pay Group	Step 1	Step 2	Step 3	Step 4	Step 5
	_		_		_
6	2,822.00	2,911.00	2,993.00	3,088.00	3,178.00
	18.57	19.15	19.69	20.32	20.91
8	2,993.00	3,088.00	3,178.00	3,273.00	3,373.00
	19.69	20.32	20.91	21.53	22.19
10	3,178.00	3,273.00	3,373.00	3,473.00	3,578.00
	20.91	21.53	22.19	22.85	23.54
12	3,373.00	3,473.00	3,578.00	3,684.00	3,795.00
	22.19	22.85	23.54	24.24	24.97
14	3,578.00	3,684.00	3,795.00	3,909.00	4,024.00
	23.54	24.24	24.97	25.72	26.47
1.6		2 000 00	4		
16	3,795.00	3,909.00	4,024.00	4,147.00	4,271.00
	24.97	25.72	26.47	27.28	28.10
10	4 024 00	4 1 4 7 00	4 271 00	4 206 00	4 520 00
18	4,024.00	4,147.00	4,271.00	4,396.00	4,529.00
	26.47	27.28	28.10	28.92	29.80
20	4,271.00	4,396.00	4,529.00	4,664.00	4,804.00
20	-		-	-	-
	28.10	28.92	29.80	30.68	31.61
22	4,529.00	4,664.00	4,804.00	4,950.00	5,096.00
22	29.80	30.68	31.61	32.57	33.53
	27.00	50.00	51.01	52.57	55.55
24	4,804.00	4,950.00	5,096.00	5,252.00	5,407.00
	31.61	32.57	33.53	34.55	35.57
26	5,096.00	5,252.00	5,407.00	5,572.00	5,735.00
	33.53	34.55	35.57	36.66	37.73
28	5,407.00	5,572.00	5,735.00	5,908.00	6,083.00
	35.57	36.66	37.73	38.87	40.02

<u>APPENDIX E – Salary Scale Effective July 1, 2019 – June 30, 2020</u>

For the purpose of calculating hourly rates from the basic monthly salaries, the monthly salary as shown above is divided by 152.

Pay Group	Step 1	Step 2	Step 3	Step 4	Step 5
6	2,878.00	2,969.00	3,053.00	3,150.00	3,242.00
	18.93	19.53	20.09	20.72	21.33
8	3,053.00	3,150.00	3,242.00	3,338.00	3,440.00
	20.09	20.72	21.33	21.96	22.63
10	3,242.00	3,338.00	3,440.00	3,542.00	3,650.00
10	21.33	21.96	22.63	23.30	24.01
	21.55	21.90	22.03	23.30	24.01
12	3,440.00	3,542.00	3,650.00	3,758.00	3,871.00
	22.63	23.30	24.01	24.72	25.47
14	3,650.00	3,758.00	3,871.00	3,987.00	4,104.00
	24.01	24.72	25.47	26.23	27.00
1.6	2 071 00	2 007 00	4 1 0 4 0 0	4 220 00	1256.00
16	3,871.00	3,987.00	4,104.00	4,230.00	4,356.00
	25.47	26.23	27.00	27.83	28.66
18	4,104.00	4,230.00	4,356.00	4,484.00	4,620.00
10	27.00	27.83	28.66	29.50	30.39
	27.00	27.00	20.00	27.00	00.09
20	4,356.00	4,484.00	4,620.00	4,757.00	4,900.00
	28.66	29.50	30.39	31.30	32.24
22	4,620.00	4,757.00	4,900.00	5,049.00	5,198.00
	30.39	31.30	32.24	33.22	34.20
24	4 000 00	5 0 4 0 0 0	5 109 00	5 257 00	5 5 1 5 00
24	4,900.00	5,049.00	5,198.00	5,357.00	5,515.00
	32.24	33.22	34.20	35.24	36.28
26	5,198.00	5,357.00	5,515.00	5,683.00	5,850.00
	34.20	35.24	36.28	37.39	38.49
	•				
28	5,515.00	5,683.00	5,850.00	6,026.00	6,205.00
	36.28	37.39	38.49	39.64	40.82

<u>APPENDIX F – Salary Scale Effective July 1, 2020 – June 30, 2021</u><sup>1</sup>

For the purpose of calculating hourly rates from the basic monthly salaries, the monthly salary as shown above is divided by 152.

Pay Group	Step 1	Step 2	Step 3	Step 4	Step 5
6	2,936.00	3,028.00	3,114.00	3,213.00	3,307.00
	19.32	19.92	20.49	21.14	21.76
8	3,114.00	3,213.00	3,307.00	3,405.00	3,509.00
	20.49	21.14	21.76	22.40	23.09
10	3,307.00	3,405.00	3,509.00	3,613.00	3,723.00
	21.76	22.40	23.09	23.77	24.49
12	3,509.00	3,613.00	3,723.00	3,833.00	3,948.00
	23.09	23.77	24.49	25.22	25.97
14	3,723.00	3,833.00	3,948.00	4,067.00	4,186.00
	24.49	25.22	25.97	26.76	27.54
16	3,948.00	4,067.00	4,186.00	4,315.00	4,443.00
	25.97	26.76	27.54	28.39	29.23
18	4,186.00	4,315.00	4,443.00	4,574.00	4,712.00
	27.54	28.39	29.23	30.09	31.00
20	4,443.00	4,574.00	4,712.00	4,852.00	4,998.00
	29.23	30.09	31.00	31.92	32.88
22	4,712.00	4,852.00	4,998.00	5,150.00	5,302.00
	31.00	31.92	32.88	33.88	34.88
24	4,998.00	5,150.00	5,302.00	5,464.00	5,625.00
	32.88	33.88	34.88	35.95	37.01
26	5,302.00	5,464.00	5,625.00	5,797.00	5,967.00
	34.88	35.95	37.01	38.14	39.26
28	5,625.00	5,797.00	5,967.00	6,147.00	6,329.00
	37.01	38.14	39.26	40.44	41.64

<u>APPENDIX G – Salary Scale Effective July 1, 2021 – June 30, 2022<sup>1</sup></u>

For the purpose of calculating hourly rates from the basic monthly salaries, the monthly salary as shown above is divided by 152.

#### **APPENDIX N - EXCLUSIONS**

It is agreed that the following positions are excluded from the bargaining unit. The University may add positions to, or delete positions from, this list. If the Union does not agree with changes to the list it may ask the Labour Relations Board to review the changes and make a final determination of the bargaining unit status of the positions in question.

\*\*Assistant Registrar \*\*Associate Registrars \*\*Associate Vice President, Human Resources \*\*Associate Vice President, Student Success Budget Officer \*\*Chief Information Officer Conflict Resolution Advisor \*\*Deans \*\*Director, Academic Initiatives and Planning \*\*Director, Alumni Relations and Development \*\*Director, Centre for International Experience **\*\***Director, Centre for Teaching Excellence \*\*Director, Continuing Studies & Executive Education \*\*Director, Creative Activity, Research and Scholarship \*\*Director, Communications and Marketing \*\*Director, Enterprise Risk Management \*\*Director, Facilities \*\*Director, Film Centre **\*\***Director of Finance \*\*Director, Indigenous Education and Affairs \*\*Director, IT Architecture \*\*Director, IT Strategic Projects \*\*Director, Student Affairs and Services \*\*Director, Systems **Employee Benefits Specialist** Executive Assistant, AVP Student Success Executive Assistant, Finance and Administration Executive Assistant, President Executive Assistant to Director Human Resources Executive Assistant to the Board Executive Assistant, Vice President Academic and Provost Executive Assistant, VP International, External Relations and New Initiatives Executive Assistant, VP Strategic Planning, Assessment and International Effectiveness Executive Assistant, VP University Relations **Executive Receptionist** External Relations Officer \*\*General Manager, Performing Arts Theatre Human Resources Analyst Human Resources Advisor Human Resources Assistant

Human Resources Associate \*\*Human Resources Manager \*\*Manager, Accounting Services \*\*Manager, Alumni Relations \*\*Manager, Athletics & Recreation \*\*Manager, Bookstore \*\*Manager, Budget & Analysis \*\*Manager, Campus Planning \*\*Manager, Campus Security \*\*Manager, Career Services \*\*Manager, Centre for International Experience \*\*Manager, Children's Centre \*\*Manager, Client Services & Deployment \*\*Manager, Creative Services \*\*Manager, Development \*\*Manager, Events & Ceremonies \*\*Manager, Facilities Building Systems \*\*Manager, Facilities **\*\***Manager, Facilities Project Management \*\*Manager, Financial Planning & Analysis \*\*Manager, Institutional Research \*\*Manager, International Student Recruitment \*\*Manager, IT Infrastructure \*\*Manager, IT Enterprise Systems \*\*Manager, OHS and Emergency Preparedness \*\*Manager, Pavroll & Benefits \*\*Manager, Policy, Privacy and Governance \*\*Manager, Project Management Office \*\*Manager, Purchasing \*\*Manager, Student Financial Accounts \*\*Manager, Student Housing Services \*\*Manager, Student Recruitment \*\*Manager, Sustainability \*\*Manager, IT Architecture and Security \*\*Manager, University Relations \*\*Manager, Web and Instructional Technology Payroll Team Lead Policy Analyst \*\*Program Director, Blueshore Theatre Centre for Performing Arts Project Manager, Facilities \*\*President \*\*Regional Director, kálax-ay/Sunshine Coast Residence Life Facilitator – Student Housing \*\*Registrar Senior Communications Advisor Senior Human Resources Advisor Student Right and Responsibilities Officer **\*\***University Librarian \*\*Vice President, Academic and Provost \*\*Vice President, Finance & Administration

\*\*Vice President, Strategic Planning, Assessment and Institutional Effectiveness \*\*Vice President, University Relations

For the purposes of this Agreement, positions indicated by "\*\*" are Administrative Heads.

The University will provide the Union with the names of the individuals currently in excluded positions within thirty (30) calendar days of the ratification of the Collective Agreement. In addition, the University will advise the Union of any changes, in writing, within thirty (30) calendar days.

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

#### AND:

# MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEE'S UNION, LOCAL 378) (the "Union")

#### Re: Testing

With respect to the above cited subject matter, the University and the Union do hereby expressly and mutually agree that if any employee is to be given any test for any purpose under the Collective Agreement, the following provisions must be complied with, otherwise the test shall be deemed to be invalid.

#### (1) Tests to be Designed and Administered Fairly

Any test given any employee for any purpose under the Collective Agreement must be designed and administered fairly.

#### (2) Other Criteria for Determining Propriety of Tests

Without limiting the generality of paragraph (1) above, any test given any employee for any purpose under this Agreement must be:

- (a) relevant to the job, if used for selection purposes;
- (b) based on material, the general nature of which will be provided to the employee in advance;
- (c) preceded by advance notice of the test procedures, the standards to be applied and the time and place where it is to be conducted;
- (d) administered on a uniform basis to all employees to be tested and in such manner as to ensure the secrecy of test papers before the test is given and where possible, to preserve the anonymity of each employee being tested.

#### (3) Adequate Preparation Time

Employees must be given adequate time to prepare for any test given for any purpose under the Collective Agreement.

#### (4) Employee Right to Review Test and Results

Any employee who is given any test for any purpose under the Collective Agreement shall have the right to review <u>their</u> test paper and results.

SIGNED this 18 day of \_\_\_\_\_, 2020

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### BETWEEN: CAPILANO UNIVERSITY (the "University")

#### AND: MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEE'S UNION, LOCAL 378) (the"Union")

#### **RE:** University Closure Policy

It is agreed that the staff shall receive equal treatment with faculty under the University Closure Policy. <u>Should the University</u>, or an area of the University, be officially closed temporarily due to environmental conditions, utility disruptions, road conditions, acts of God, or other reasons beyond the control of the employees covered by this Agreement, employees shall receive salary maintenance during the closure.

Those employees required to remain on duty to provide essential services shall be given the equivalent time off with such time off to be taken within the following calendar month. Should these employees be required to remain beyond their normal hours of work, the usual provisions for overtime would apply.

Employees who are requested to report for work for an afternoon or night shift which commences during a closure shall receive equivalent time off for all hours worked on the shift, with such time off to be taken within the following calendar month. Should these employees be required to remain beyond their normal hours of work, the usual provisions for overtime would apply.

SIGNED this 18 day of June , 2020

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

AND:

# MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "Union")

#### **RE: ARTICLE 19 CAREER DEVELOPMENT**

The parties agree to the following in dealing with career development issues:

- 1. The parties are committed to the intended purpose of Article 19: "to encourage and assist employees in maintaining and improving skills or to assist in preparing them for foreseeable jobs within the University."
- 2. The parties acknowledge that the Staff Career Development Committee will make recommendations to the University on the dispersal of staff career development funds, made available by the University in the amount of not less than ten thousand (\$10,000) dollars annually.
- 3. To ensure a good working relationship, the University will make available a resource person to work with the Staff Career Development Committee when requested.
- 4. Prior to committing to any activity, the Staff Career Development committee will submit a plan of activities with anticipated learning outcomes and budget for as part of its recommendation to the University.

SIGNED this <u>18</u> day of <u>June</u>, **2020** 

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

#### AND:

# MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEE'S UNION, LOCAL 378) (the "Union")

#### Re: Exceptional Work Day – Centre for Sport & Wellness

With respect to the above subject matter, the University and the Union ("Parties") do hereby expressly and mutually agree that an employee hired to work in the <u>Centre for Sport &</u> <u>Wellness</u> for a special event may agree to work an exceptional work day of longer than seven (7) hours per day. The employee shall be paid at the overtime rate for all hours in excess of ten (10) hours per day or thirty-five (35) hours per week.

This letter of understanding will expire June 30, <u>2022</u> with the expiry of the current Collective Agreement.

SIGNED this 18 day of June ,2020:

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

H

Nathan Beausoleil, Union Representative

M

**BETWEEN:** 

#### CAPILANO UNIVERSITY (the "UNIVERSITY")

AND:

# MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "UNION")

#### **Re: Labour Market Adjustment for Selected IT Systems Positions**

<u>All employees currently red-circled due to the labour market adjustment stipend in IT shall</u> now be considered blue circled for the remainder of their employment at Capilano University, or until the parties negotiate otherwise.

SIGNED this 18 day of June ,2020

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

AND:

#### **MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "Union")**

#### **RE: EARLY CARE AND LEARNING RECRUITMENT AND STRATEGY FUNDING**

All eligible Early Childhood Educators (ECEs) at Capilano University shall be granted funding as per the B.C. Government's "Early Care and Learning Recruitment and Retention Strategy."

All eligible ECEs shall receive a \$1 per hour wage redress retroactive to September 1, 2018.

All eligible ECEs shall receive any additional wage redress as determined by the "Early Care and Learning Recruitment and Retention Strategy."

To be eligible, ECEs must be in good standing with the BC ECE registry, work 20 hours or more per week in direct care of children or spend 50 percent or more of their working time in direct contact with children.

Should the University no longer be in receipt of the funding through this program, the program ceases, or should the employee no longer hold an ECE in good standing with the BC ECE Registry, then the ECE Wage Enhancement will also cease.

SIGNED this <u>18</u> day of <u>June</u>, **202** 

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

#### AND:

#### **MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "Union")**

#### **RE: TARGETED PRESCRIPTION SPENDING ACCOUNT (TPSA)**

#### **Targeted Prescription Spending Account (TPSA)**

The University and the Union agree that where an individual who is covered by the extended health care insurance plan and has prescription drug coverage is:

- a) Prescribed a drug that is not part of the Pharmacare Formulary ("Non-Pharmacare Formulary Drug"); or
- b) Prescribed a drug that is on the Pharmacare Formulary but which requires "Special Authority" ("Special Authority Drug") and the individual either:
  - i) Does not want to go through the approval process to obtain the Special Authority Drug; or
  - ii) Does go through the approval process and is unsuccessful in being granted the Special Authority for coverage of a particular drug,

the individual shall be eligible to receive coverage for the Non-Pharmacare Formulary Drug of the Special Authority Drug provided that the individual pays fifty percent (50%) of the cost of the Non-Pharmacare Formulary Drug or Special Authority Drug. The insurance plan will pay the remaining fifty percent (50%) of the cost of the Non-Pharmacare Formulary Drug or Special Authority Drug.

Where an individual is covered by the extended health care insurance plan and has prescription drug coverage is covered for a Formulary drug and that drug was recently delisted from the Formulary (the "Delisted Drug") and the individual requires a period of time to transition from the Delisted Drug to a Formulary Drug, the University will instruct the insurance carrier to cover the Delisted Drug for that individual for a period of up to three (3) months following the date the Delisted Drug was denied by the insurance carrier.

In addition, a Targeted Prescription Spending Account (TPSA) is available for the purpose of providing up to an additional 25% reimbursement for certain drug prescriptions.

To avail of this benefit, the following requirements apply:

- The prescription drug is not on the Pharmacare Formulary (including those available through special authority), and
- The drug was previously available under the Staff drug plan in place prior to the University and Union adopting the BC Pharmacare Formulary.

The TPSA shall be available for the purpose of reimbursing eligible claims for Staff employees and their dependents combined, up to a maximum of \$400 per year. Unused TPSA funds from any given calendar year will not carry forward to the next year.

SIGNED this <u>18</u> day of <u>June</u>, **202** 

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### **CAPILANO UNIVERSITY (the "University")**

AND:

#### **MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "Union")**

#### **RE: TEMPORARY VACATION CARRY-OVER INCREASE**

Recognizing that more senior employees may have carried larger banks when the parties agreed last round of bargaining to cap the vacation banks at a maximum of twenty (20) days, this group may have had challenges transitioning to the new balance cap. In recognition of these senior employees, a longer transition period will be provided.

Irrespective of the current ten (10) day carry over, for employees who are entitled to thirty (30) days' vacation, they may carry over up to fifteen (15) days per year and hold a maximum bank of thirty days (30) days per year, until December 31, 2021. As of January 1, 2022, all employees will be expected to adhere to the maximums outlined in Article 17.08.

SIGNED this <u>18</u> day of <u>June</u>, **2020** 

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "University")

AND:

#### **MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "Union")**

#### **RE: STRATEGIC FUNDING FOR FRONT LINE SUPPORT**

The parties recognize that front line student engagement occurs at all levels of the university. For those employees in the following positions, positive service delivery plays a significant role in the University's ability to achieve its strategic goal and the parties wish to acknowledge this commitment by providing a \$1 per hour premium.

- Child Care Facility Attendant
- <u>Clerk 1 (Bookstore)</u>
- Substitute, Child Care Assistant
- <u>Facilities Worker</u>
- Facilities Worker (As & When)
- Centre for Sport & Wellness Attendant
- <u>On-call Department Support Assistant (Basic)</u>

SIGNED this 18 day of June , 2020

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

#### **BETWEEN:**

#### CAPILANO UNIVERSITY (the "UNIVERSITY")

#### AND:

#### MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES' UNION, LOCAL 378) (the "UNION")

#### **Re: Support Staff Innovation Fund**

<u>Capilano University agrees to establish a Support Staff Innovation Fund to encourage specific transportation initiatives and address recruitment and retention issues. The amount of the fund will be:</u>

Year 1	\$40,000
Year 2	\$85,000
Year 3 and ongoing	\$130,000

Funds not allocated within a calendar year will be carried over for one year only. Due to the timing of collective bargaining for the renewal of the 2014-2019 collective agreement, it is likely that funds from Year 1 may not be spent in Year 1. The intent of the parties is to carry any unspent amount to Year 2 and to ensure those funds are allocated in Year 2.

#### SPECIFIC INITIATIVES

#### A. <u>GREEN TRANSIT INITIATIVES</u>

#### (i) TRANSPORTATION SUBSIDY

As part of its efforts to reduce its carbon footprint, the University shall provide to regular employees who commute to and from work by public transit, a transportation incentive of fifty (\$50.00) dollars per month toward a monthly transit pass.

Starting January 1, 2020, to receive the transportation subsidy employees shall, on a quarterly basis, submit a copy of their transit payment receipts (ie. Compass Pass payment receipt) through the normal expense reporting process. Payment will be issued quarterly.

#### (ii) <u>CARPOOLING PILOT PROJECT</u>

To encourage carpooling as a transportation option to employees, the parties have agreed to participate in a carpooling pilot project.

The University will provide 10 spaces for MoveUp members who carpool. Employees who carpool do not have to pay for parking. As the pilot progresses, this number will be reviewed to determine the necessity of increasing spaces to a maximum of 20. Carpools must consist of at least 2 MoveUp members who arrive on campus together, in one vehicle, park in the designated carpool area, and exit the vehicle together. As per current regulations around carpooling, if the registered drivers listed for the carpool do not exit the car together upon parking, the driver will lose their carpooling privileges for the remainder of the pilot project. The application form and further terms and conditions can be found on the University's Frontlines web site under "Carpool Terms & Conditions".

This 1 year pilot project will run from January 1, 2020 to December 31, 2020. The pilot project will be evaluated in November 2020, and if 80% utilization is achieved between January 1, 2020 and October 31, 2020, the project will continue for another year. It will be evaluated with the union at 6 month intervals thereafter.

#### B. LABOUR MARKET ADJUSTMENT COMMITTEE

The parties recognize the value of developing positive recruitment and retention practices within the University. As such, the Parties agree to establish a Joint Committee consisting of up to two (2) representatives from the Union and up to two (2) representatives from the Employer to review positions with difficulties in recruitment and/or retention and recommend strategies to eliminate these difficulties.

Recommendations shall be made to the Associate VP Human Resources and the MoveUP Business Representative for final review and agreement before implementation.

Positions may be eligible for additional labour market adjustments by mutual agreement if:

- <u>There is a demonstrated recruitment or retention issue that can be objectively</u> <u>determined, and;</u>
- <u>There is an objective concern for the retention of incumbents who are being</u> <u>underpaid in contrast to appropriate market comparators, and;</u>
- Any other eligibility requirements the parties may mutually agree to as necessary.

Labour market adjustments may include, but will not be limited to, the following:

- <u>Annual Stipend (paid bi-weekly)</u>
- <u>Salary adjustments to match market reference points for similar positions</u> within the sector

Either party may initiate a meeting of the committee.

SIGNED this <u>18</u> day of <u>June</u>, **2020** 

MoveUP (CANADIAN OFFICE AND PROFESSIONAL EMPLOYEES UNION, LOCAL 378)

d

Nathan Beausoleil, Union Representative

Mike Knudson, Associate VP Human Resources

## **INDEX**

# Α

Absence without Leave	
Absence, Leave of	See (Leaves of Absence)
Access to University Premises	
Accural, Seniority	
Acquainting New Employees	
Adjudication	
Administrative Heads	45 <i>,</i> 157
Adoption Leave	
Appeals, Classification	
Application for Union Membership	
Appointment, Procedures	
Appointments	
Greater than 3 Months	
Job Duties	
Notification to Union	
One Month Hires	
One Month to Three Months	
Seniority	
To New Position	
Apprenticeship	
Arbitration	
Arbitrators	
Expedited	
Pre-hearing Conference	
Attendance at Meetings	
Automobile, Employee's	

#### В

Bargaining Unit	
Benchmarks	
Benefit Maintenance	
Bridge Period	
Layoff	59
Leaves	
Benefits	
Accidental Death & Disablement	
Application & Waiver Form	128, 149
Basic Medical Plan	
Coverage	
Dental Plan	
Eligibility	
Extended Health	
Flexible Benefits Plan	
Group Life Insurance	
Health & Welfare	
Long Term Disability	
Spending Account	
Bereavement Leave	91
Definition of Immediate Family	
During Vacation	
Timing	
Bridge Period	
-	

No Call Back	
No Change	
Bumping	
Burden of Proof	

# С

Call Back	
Call Back from Vacation	
Career Development	
Career Development Committee	
Carriers Appeal Process	
Cessation of Operations	
Check-off of Fees & Dues	
Class Specification	
Appeal	See (Appeals, Classification)
Classification	
Review	
Temporary Reclassification	
Closure of University	
Community Facilities	
Competitions	
Consultation	
Contracting Out	
Correspondence	
Courses	
Continuing Education	
Credit Courses	
Non-base Funded	
Time Spent Attending	
Critical Care Leave	See (Bereavement Leave)

# D

Deferred Salary Leave	
Dental Plan	See (Benefits)
Discharges	
Discipline	
Burden of Proof	
Notice of	
Disclosure of Information	
Donating Blood	
Driver's License	
Duration of Contract	

# Ε

Early Intervention Program	See (Joint Early Intervention Program)
Early Retirement Plan	
Benefit Options	
Eligibility	
Effect of Settlements	
Effective Dates	
Elections, Leave For	
Federal, Provincial or Municipal	
Electronic Monitoring	
Employee	
Changes in Status	
Evaluation	
New	

Personnel File	
Employee Appointments	
Personnel Hiring Form	
Employee Assistance Plan	
Employees, Categories of	
Exceptions	
Regular Full Time (RFT)	
Regular Part Time (RPT)	
Temporary Full Time (TFT)	
Temporary Part Time (TPT)	
Equal Pay	
Errors & Omissions	
Evaluations	<i>See</i> (Employee)
Examination Leave	
Exceptional Work Day	
Exclusions, Bargaining Unit	
Expedited Arbitration	
Explanatory Notes	
External Hires	

## F

Facilities	
Community	See (Community Facilities)
Community Use of	
Family	
Definition of Immediate Family	91
Responsibility Leave Fees & Dues	
Fees & Dues	
Deduction of	
Record of	
Tuition Fee Waiver	
File, Employee's	
Materials On	
Union Access to	
Financial Assistance	See (Sick Leave)
First Aid	
Flexible Benefits Plan	
Funeral Attendance (Mourning)	See (Special Leave)

# G

Graduated Return to Work	
Grievance Procedure	
Competitions	
Complaint - Step 1	
Grievance - Step 2	
Step 3	
Time Limits	
Group Life Insurance	
	220

## Н

135
. 135
19
efits)
75
75

Coinciding with Vacation	
Falling on Saturday or Sunday	
Paid (Statutory) & Other	
Payment for	
Required to Work on	
Hourly Premiums	70
Hours of Work	
Approval of	
Definition	
Exceptional	
Modified	
Standard Work Week, Work Day	
Human Rights Act	

Immediate Family, Definition of Incidental, Non-File Written Data	See (Bereavement Leave)
Re - Employee	
Increments	
Pay Matrix	
Information	
Collective Bargaining Purposes	
Provided to Union	
Provided to Union (Layoff)	
Interim Appointments	

I

# J

Job Evaluation	
Specialist	
Job Opening	
Job Posting	
Advertisements	
Contents of Posting	
Union Copy of Posting	
Joint Early Intervention Program	
Joint Standing Committee	

# L

Labour Relations Code	
Layoff	
Defintion of	
Elections	51
Notice	
Severance Pay	
Time Lines	
Layoff List	
Layoff, Type 1 (Involuntary Cessation)	
Employer's Reduction of Hours	
Permanent	
Temporary	
Temporary Becomes Permanent	
Layoff, Type 2 (Bridge Period)	
Layoff, Type 3 (Normal Expiry of Temporary Appointment)	
Leaves of	
Union Activity	23
Leaves of Absence	
Benefits	

Bereavement Leave	
Community Emergency	
Court Appearance	
Deferred Salary	
Donating Blood	
During Vacation	
Elections	
Examination	
Family Responsibility	
No Call Back	
Parental, Maternity, Adoption	
Personal, With Pay	
Public Duties	
Sick Leave	
Special Leave	
Temporary Employees	
Termination of Leave	
Training (Employee Requested)	
Training (Employer Required)	
Without Pay	
Legal Action, Protection Against	
Legislation	
Existing	
Future	
Minimum Acceptable Standards	
Letter of Understanding	
Re - Article 19 Career Development	
Re - Exceptional Work Day	
Re - Labour Market Adjustment for Selected IT Systems Positions	
Re - Testing	
Re - University Closure Policy	
Lie Detector Tests	
Lieu Day	77
Long Term Disability	
Long Term Illness	
Long Term Replacements	

#### Μ

67

# New Positions20Notification of Union20Salary21No Other Agreement19

Ν

## 

Oral Warning	
Overtime	
Application to Overtime Premium (Article 4)	
Right to Refuse	
Ovetime	
Compensation	71

Ρ

Parental Leave	
Supplemental Employment Benefit	
Parking Expenses (Reimbursement)	
Part Time Employees	
Additional Hours of Work	
Definition of	
Pay	
, Days	
Group & Matrix	
Minimum Daily Pay	
New Positions	
Rates	
Reclassification	
Substitution	
Performance Evaluation	
Personal Duties	
Personal Harassment	
Personnel Hiring Form	
Picket Lines	
Political Activity	
Post-Maternity Rights	
Practicum	
Pre-hearing Conference	See (Arbitration)
Previous Incumbent	
Definition	
List	
Probation	
Proof of Illness	

# R

Recall	53
List	
Reduced Hours, Phased-In Return to Work	
Reduction of Hours	
Employee's Request	
Employer's Request	
Regularization	
Rest Periods	
Retirement	
Early	See (Early Retirement)
Return to Work	

# S

Safety & Health	
Committee	
Refusal to Work	
Safety Footwear	
Training	-
	200

Salary Scale.131, 152, 153, 154Search Committee, Administration.30Seniority	Unsafe Area	
Seniority       47         Accrual       47         Carry Over       49         Determination Of       49         List       141         Loss       48         Maintenance       48, 93         Severance Pay       49, 51, 56, 58, 59, 107         Sexual Harassment       See (Harassment)         Shift Premium       70         Short Change-Over       70         Short Change-Over       70         Shudowns       103         Sick Leave       83, 84, 85         During Vacation       80         Long Term Illness       83         Pregnant Employee       83         Special Leave       92         Citizenship       92         Mourning       92         Statutory Holiday       See (Holidays)         Step Increases       See (Increments)	Salary Scale	
Acrual47Carry Over49Determination Of49List141Loss48Maintenance48, 93Severance Pay49, 51, 56, 58, 59, 107Sexual HarassmentSee (Harassment)Shift Premium5ee (Harassment)Shift Premium70Short Change-Over70Short Change-Over70Short Term Disability150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Special Leave92Citizenship92Mouming92Standby Duty68Statutory HolidaySee (Increments)Step IncreasesSee (Increments)	Search Committee, Administration	
Carry Over49Determination Of49List141Loss48Maintenance48, 93Severance Pay.49, 51, 56, 58, 59, 107Sexual HarassmentSee (Harassment)Shift Premium.5ee (Harassment)Short Change-Over.70Short Change-Over.70Short Term Disability.150Shutdowns.103Sick Leave.83, 84, 85During Vacation.80Long Term Illness.83Pregnant Employee.83Special Leave.92Citizenship.92Mouming.92Statutory Holiday.5ee (Increments)Step Increases.5ee (Increments)	Seniority	
Determination Of49List141Loss48Maintenance48, 93Severance Pay49, 51, 56, 58, 59, 107Sexual HarassmentSee (Harassment)Shift Premium70Short Change-Over70Short Crame Over70Short Term Disability150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Special Leave92Citizenship92Mourning92Mourning92Stadby Duty.68Statutory HolidaySee (Increments)	Accrual	
List141Loss48Maintenance48,93Severance Pay49,51,56,58,59,107Sexual HarassmentSee (Harassment)Shift Premium70Short Change-Over70Short Change-Over70Short Term Disability150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Special Leave92Citizenship92Mourning92Standby Duty68Statutory HolidaySee (Increments)Step IncreasesSee (Increments)	Carry Over	
Loss48Maintenance48,93Severance Pay49, 51, 56, 58, 59, 107Sexual HarassmentSee (Harassment)Shift Premium70Short Change-Over.70Short Term Disability.150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave92Citizenship92Mourning92Standby Duty68Statutory HolidaySee (Increments)Step IncreasesSee (Increments)	Determination Of	
Maintenance48, 93Severance Pay49, 51, 56, 58, 59, 107Sexual HarassmentSee (Harassment)Shift Premium70Short Change-Over70Short Term Disability150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Special Leave92Citizenship92Standby Duty68Statutory HolidaySee (Increments)Step IncreasesSee (Increments)	List	
Severance Pay.49, 51, 56, 58, 59, 107Sexual Harassment	Loss	
Sexual Harassment.See (Harassment)Shift Premium70Short Change-Over.70Short Term Disability.150Shutdowns103Sick Leave.83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Special Leave.92Citizenship92Mourning92Standby Duty.68Statutory HolidaySee (Increments)	Maintenance	
Shift Premium70Short Change-Over.70Short Term Disability.150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave92Citizenship92Mourning92Standby Duty.68Statutory HolidaySee (Holidays)Step IncreasesSee (Increments)	Severance Pay	
Short Change-Over.70Short Term Disability.150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave92Citizenship.92Mourning.92Standby Duty.68Statutory HolidaySee (Holidays)Step IncreasesSee (Increments)	Sexual Harassment	<i>See</i> (Harassment)
Short Term Disability.150Shutdowns103Sick Leave83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave92Citizenship92Mourning92Standby Duty.68Statutory HolidaySee (Holidays)Step IncreasesSee (Increments)	Shift Premium	
Shutdowns       103         Sick Leave       83, 84, 85         During Vacation       80         Long Term Illness       83         Pregnant Employee       83         Proof of Illness       83         Special Leave       92         Citizenship       92         Mourning       92         Standby Duty.       68         Statutory Holiday       See (Holidays)         Step Increases       See (Increments)	Short Change-Over	
Sick Leave.83, 84, 85During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave.92Citizenship.92Mourning.92Standby Duty.68Statutory HolidaySee (Holidays)Step IncreasesSee (Increments)	Short Term Disability	
During Vacation80Long Term Illness83Pregnant Employee83Proof of Illness83Special Leave92Citizenship92Mourning92Standby Duty68Statutory HolidaySee (Holidays)Step IncreasesSee (Increments)	Shutdowns	
Long Term Illness	Sick Leave	
Pregnant Employee       83         Proof of Illness       83         Special Leave       92         Citizenship       92         Mourning       92         Standby Duty.       68         Statutory Holiday       See (Holidays)         Step Increases       See (Increments)	During Vacation	
Proof of Illness	Long Term Illness	
Special Leave       92         Citizenship       92         Mourning       92         Standby Duty       68         Statutory Holiday       See (Holidays)         Step Increases       See (Increments)	Pregnant Employee	
Citizenship	Proof of Illness	
Mourning	Special Leave	
Standby Duty	Citizenship	
Statutory Holiday	Mourning	
Step Increases	Standby Duty	
	Statutory Holiday	See (Holidays)
Substitution	Step Increases	<i>See</i> (Increments)
	Substitution	

# Т

Technological Change	
Dispute Resolution	
Technological Change Dispute Resolution Retraining	
Temporary Promotion	
Term of Agreement	
Time Report	67
Training	
Health & Safety	
Health & Safety Leave, Employee Requested	
Leave Employer Required	87
Transportation	
Transportation	
Travel	
Domestic	
International	
Tuition Fee Waivers	

# U

Unfair Labour Practices	
Union Insignia	
Union Membership	
Union Rights & Activity	
Business Leave	
Contacting at Work	
General Meeting	
Notice of Absence from Workplace	
Office Space	
Record of Deductions	

Use of Facilities	
Voting	
Union Security	

# V

Vacation	
Banking	
Bridge Period	
Impact of Temporary Promotion	
No Call Back	
No Change	
Overlap with Leaves of Absence	
Vehicle, Use of	
Voting	

## W

Work Clothing	
Work Experience Placements	
Work Schedule	
Workers' Compensation	, , , , , , , , , , , , , , , , , , ,
Act	
Payment from	
Supplement	
Workload	